### **United States** SECURITIES AND EXCHANGE COMMISSION

Washington DC 20549

	FORM 10-Q	
(Mark ⊠	One) QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECUI 1934	RITIES EXCHANGE ACT OF
	For the quarterly period ended March 31, 2005	
	-or-	
	TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECUL 1934	RITIES EXCHANGE ACT OF
	For the transition period from to	
	Commission File Number 1-12298	
	REGENCY CENTERS CORPOR	ATION
		59-3191743 (IRS Employer dentification No.)
	121 West Forsyth Street, Suite 200  Jacksonville, Florida 32202  (Address of principal executive offices) (Zip Code)	
	(904) 598-7000 (Registrant's telephone number, including area code)	
	Unchanged (Former name, former address and former fiscal year, if changed since last report)	

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes  $\boxtimes$  No  $\square$ 

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act). Yes ⊠ No □

(Applicable only to Corporate Registrants)

As of May 9, 2005, there were 63,123,053 shares outstanding of the Registrant's common stock.

### TABLE OF CONTENTS

		Form 10-Q Report Page
PART I	- FINANCIAL INFORMATION	
Item 1.	<u>Financial Statements</u>	
	Consolidated Balance Sheets as of March 31, 2005 and December 31, 2004	1
	Consolidated Statements of Operations for the three months ended March 31, 2005 and 2004	2
	Consolidated Statement of Stockholders' Equity and Comprehensive Income (Loss) as of March 31, 2005	3
	Consolidated Statements of Cash Flows for the three months ended March 31, 2005 and 2004	4
	Notes to Consolidated Financial Statements	5
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	24
Item 3.	Quantitative and Qualitative Disclosures about Market Risk	39
Item 4.	Controls and Procedures	39
PART II	- OTHER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	40
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	40
Item 3.	<u>Defaults Upon Senior Securities</u>	41
Item 4.	Submission of Matters to a Vote of Security Holders	41
Item 5.	Other Information	41
Item 6.	<u>Exhibits</u>	41
SIGNAT	<u>ture</u>	42

### PART I – FINANCIAL INFORMATION

### **Item 1. Financial Statements**

### REGENCY CENTERS CORPORATION Consolidated Balance Sheets

March 31, 2005 and December 31, 2004 (in thousands, except share data)

	2005	2004
	(unaudited)	
<u>Assets</u>	` ′	
Real estate investments at cost:		
Land	\$ 826,266	806,207
Buildings and improvements	1,944,493	1,915,655
	2,770,759	2,721,862
Less: accumulated depreciation	352,818	338,609
	2,417,941	2,383,253
Properties in development	379,313	426,216
Operating properties held for sale	15,910	4,916
Investments in real estate partnerships	180,478	179,677
Net real estate investments	2,993,642	2,994,062
Cash and cash equivalents	53,591	95,320
Notes receivable	23,252	25,646
Tenant receivables, net of allowance for uncollectible accounts of \$3,683 and \$3,393 at March 31, 2005 and December 31, 2004, respectively	50,051	60,911
Deferred costs, less accumulated amortization of \$27,751 and \$25,735 at March 31, 2005 and December 31, 2004, respectively	40,502	41,002
Acquired lease intangible assets, net	13,280	14,172
Other assets	21,248	12,711
	\$3,195,566	3,243,824
<u>Liabilities and Stockholders' Equity</u>		
Liabilities:		
Notes payable	\$1,291,039	1,293,090
Unsecured line of credit	175,000	200,000
Accounts payable and other liabilities	79,919	102,443
Acquired lease intangible liabilities, net	4,923	5,161
Tenants' security and escrow deposits	9,959	10,049
Total liabilities	1,560,840	1,610,743
Preferred units	101,762	101,762
Exchangeable operating partnership units	29,324	30,775
Limited partners' interest in consolidated partnerships	1,932	1,827
Enfined paralets interest in consolidated paraletsinps		
Total minority interest	133,018	134,364
Stackhaldow' aguiten		
Stockholders' equity:  Preferred stock, \$.01 par value per share, 30,000,000 shares authorized; 800,000 shares issued and outstanding at March 31,	200 000	200,000
2005 and December 31, 2004, liquidation preference \$250 per share  Common stock \$.01 par value per share, 150,000,000 shares authorized; 68,390,396 and 67,970,538 shares issued at March	200,000	200,000
31, 2005 and December 31, 2004, respectively	684	680
Treasury stock at cost, 5,302,804 and 5,161,559 shares held at March 31, 2005 and December 31, 2004, respectively	(111,414)	(111,414)
Additional paid in capital	1,497,124	1,494,312
Accumulated other comprehensive (loss) income	(5,148)	(5,291)
Distributions in excess of net income	(79,538)	(79,570)
Total stockholders' equity	1,501,708	1,498,717
Commitments and contingencies	\$3,195,566	3,243,824

See accompanying notes to consolidated financial statements.

### REGENCY CENTERS CORPORATION

# Consolidated Statements of Operations For the three months ended March 31, 2005 and 2004 (in thousands, except per share data) (unaudited)

(unautiteu)	2005	2004
	2005	2004
Revenues:		
Minimum rent	\$ 73,682	2 68,185
Percentage rent	55	1 452
Recoveries from tenants	21,740	
Management fees and commissions	3,318	
Equity in income of investments in real estate partnerships	2,39	1 2,745
Total revenues	101,688	92,453
Operating expenses:		
Depreciation and amortization	21,004	4 19,561
Operating and maintenance	13,592	2 12,778
General and administrative	8,652	2 5,883
Real estate taxes	10,488	8 10,100
Other expenses	1,428	8 488
Total operating expenses	55,16	4 48,810
Other expense (income)		
Interest expense, net of interest income of \$505 and \$837 in 2005 and 2004, respectively	21,070	6 21,051
Gain on sale of operating properties and properties in development	(6,54)	
	(0,5 1.	
Total other expense	14,53	4 17,068
Income before minority interests	31,990	0 26,575
Minority interest of preferred units	(2,112	2) (5,081)
Minority interest of exchangeable operating partnership units	(650	6) (355)
Minority interest of limited partners	(70	6) (79)
Income from continuing operations	29,140	6 21,060
Discontinued operations, net:		
Operating income from discontinued operations	400	6 1,745
Gain on sale of operating properties and properties in development	8,79	
Income from discontinued operations	9,202	2 1,757
·		
Net income	38,348	8 22,817
Preferred stock dividends	(3,662	2) (1,397)
Net income for common stockholders	\$ 34,680	6 21,420
Income per common share - basic:	\$ 0.40	0 0.33
Continuing operations Discontinued operations	\$ 0.40	
Discontinued operations	0.13	0.03
Net income for common stockholders per share	\$ 0.55	5 0.36
Income per common share - diluted:		
Continuing operations	\$ 0.40	0 0.32
Discontinued operations	0.1	
Net income for common stockholders per share	\$ 0.55	5 0.35

See accompanying notes to consolidated financial statements.

### REGENCY CENTERS CORPORATION

# Consolidated Statement of Stockholders' Equity and Comprehensive Income (Loss) For the three months ended March 31, 2005 (in thousands, except per share data) (unaudited)

	Preferred Stock	Common Stock	Treasury Stock	Additional Paid In Capital	Accumulated Other Comprehensive Income (Loss)	Distributions in Excess of Net Income	Total Stockholders' Equity
Balance at December 31, 2004	\$200,000	680	(111,414)	1,494,312	(5,291)	(79,570)	1,498,717
Comprehensive Income:							
Net income	_	_	_	_		38,348	38,348
Amortization of loss on derivative instruments	_	_	_	_	143	_	143
Total comprehensive income	_	_	_	_	_	_	38,491
Common stock issued as compensation and stock options							
exercised, net	_	3	_	1,257	_	_	1,260
Common stock issued for partnership units exchanged	_	1	_	1,769	_	_	1,770
Reallocation of minority interest	_	_	_	(214)	_	_	(214)
Cash dividends declared:							
Preferred stock	_	_	_	_		(3,662)	(3,662)
Common stock (\$.55 per share)	_	_	_	_	<del>_</del>	(34,654)	(34,654)
Balance at March 31, 2005	\$200,000	684	(111,414)	1,497,124	(5,148)	(79,538)	1,501,708

See accompanying notes to consolidated financial statements.

### REGENCY CENTERS CORPORATION

# Consolidated Statements of Cash Flows For the three months ended March 31, 2005 and 2004 (in thousands) (unaudited)

Nemicones   Say also   2281   Adjustments to reconcile net income to net cash provided by operating activities   Depreciation and amoritzation   22.60   20.34   Deferred base cost and debt premium amoritzation   4,343   3.54   Minority interest of preferred units   4,343   3.54   Minority interest of preferred units   6,75   7.75   Figuity in income of investments in real estate partnership units   6,75   7.75   Figuity in income of investments in real estate partnership   2,741   2,744   Net gain on sale of properties   2,741   2,744   Net gain provided by operating activities   2,741   2,744   Net cash provided by operating activities   2,741   2,744   Net cash provided by operating activities   2,741   2,744   Net cash provided by (used in investing activities   2,741   2,744   Net cash provided by (used in investing activities   2,741   2,744   Net cash provided by (used in investing activities   2,744   2,744   Net cash provided by (used in investing activities   2,744   2,744   Net cash provided by (used in investing activities   2,744   2,744   Net cash provided by (used in investing activities   2,744   2,744   Net cash provided by (used in investing activities   2,744   2,744   Net cash provided by (used in investing activities   2,744   2,744   Net cash provided by (used in inve		2005	2004
Adjustments to reconcile set income to net cash provided by operating activities:   Deperted loan cost and debt premium amoritzation   348   616     Stock based compensation   4,343   3,544     Minority interest of perferred units   75   77     Equity in interest of perferred units   76   77     Equity in income of investing pairnership units   76   77     Equity in income of investing pairnership   77   77     Equity in income of investing activities   77   77     Equity in income of investing set in each part part part part part part part part	Cash flows from operating activities:		
Dependance and annotization   2,266   20,34     Defende class costs and debt premium amoritation   4,34   3,54     Stock based compensation   4,34   3,54     Minority interest of preferred units   603   303     Minority interest of exchangeable operating partnership units   603   303     Minority interest of interest partnership arrises   76   772     Equity in income of investments in real estate partnerships   2,230   2,744     Net gain on sale of tripropriate partnerships   3,76   4,54     Heldge settlement   7,724   3,795     Distributions from operations of investments in real estate partnerships   3,76   4,54     Heldge settlement   10,860   17,02     Tennat receivables   10,860   17,02     Tennat receivables   10,860   17,02     Tennat receivables   10,860   17,02     Tennat receivables   10,860   17,02     Tennative security and secrow deposits   10,860   17,02     Accounts payable and other liabilities   19,970   33,91     Tennative security and secrow deposits   19,970   33,91     Tennative security		\$ 38,348	22,817
Deferred loan can ad debt premium amoritation			
Stock based compensation         4,343         3,54           Ministry innerest of preferred units         863         38           Ministry innerest of exchangeable operating partnership units         863         38           Ministry innerest of preferred units         675         75           Feguity in income of investments in real estate partnerships         3,761         4,724           Net gation use of properties         3,761         4,64           Heldes estatlement         10,809         17,022           Changes in assets and liabilities         10,809         17,022           Tenant receivables         10,809         17,022           Accounts possible and other liabilities         635,399         18,152           Tenants Security and escrive diposits         10,909         33,912           Net cash provided by operating activities         9,970         33,912           Cash flows from investing activities         55,600         63,333           Development of von estate         56,600         63,333           Development of von estate         55,600         63,333           Development of von estate partnership         11,166         61,166           Distributions received from investments in real estate partnership activities         3,220         26,22			20,344
Minority increat of perferred units			616
Minority incress of cichangeable operating partnership units   1683   385			3,548
Minority innerso of limited partners			5,081
Equity in income of investments in real estate partnerships         (7,74)         (3,98)           Ne gain on sale of properties         (7,76)         (4,84)           Distributions from operations of investments in real estate partnerships         — (5,70)         (5,70)           Changes in assets and liabilities:         — (1,06)         1,060		863	385
1,7724   3,98     Distributions from operations of investments in real estate partnerships   3,76   4,64     Hedge settlement   1,086   1,08			79
Distributions from operations of investments in real estate partnerships   3,761   4,64   Hedge settlement   1,066   1,066   1,066   Tennant receivables   1,066   1,066   1,066   Deferred leasing costs   1,066   1,066   1,066   Other assets		(2,391)	(2,745)
Hedge settlement			(3,995
Changes in assets and labilities	Distributions from operations of investments in real estate partnerships	3,761	4,644
Tenant receivables	Hedge settlement	_	(5,720)
Deferred leasing costs	Changes in assets and liabilities:		
1.888	Tenant receivables	10,860	17,025
Accounts payable and other liabilities (3.3.90 (28,155 (200 15) (2	Deferred leasing costs	(1,668)	(2,058
Tenants security and escrow deposits   19,97   33,912	Other assets	(5,971)	1,889
Tenants security and escrow deposits   19,97   33,912	Accounts payable and other liabilities	(35,399)	(28,155
Net cash provided by operating activities   Section			158
Cash flows from investing activities:   Development of real estate   (56,690)   (63,332)     Proceeds from sale of real estate investments   (50,690)   (63,332)     Proceeds from sale of real estate investments   (50,690)   (63,332)     Repayment of notes receivable, net   (2,394)   (60,700)     Distributions received from investments in real estate partnerships   (11,166)   (51,500)     Net cash provided by (used in) investing activities   (75,600)     Net cash provided by (used in) investing activities   (74,432)     Net proceeds from common stock issuance   (74,432)     Redemption of exchangeable operating partnership units   (757)   (757)     Redemption of exchangeable operating partnership units   (757)   (757)     Distributions to exchangeable operating partnership unit holders   (757)   (757)     Distributions to exchangeable operating partnership unit holders   (2,112)   (5,08)     Dividends paid to common stockholders   (3,662)   (1,37)     Dividends paid to preferred unit holders   (3,662)   (1,37)     Dividends paid to preferred stockholders   (3,662)   (1,37)     Repayment of notes payable, net   (3,500)   (3,500)     Repayment of notes payable, net   (3,500)   (3,500)     Repayment of notes payable, net   (3,500)   (3,500)   (3,500)     Observed loan costs   (4,172)   (3,562)   (3,500)     Net (decrease) increase in cash and cash equivalents   (4,172)   (3,562)   (3,500)     Observed loan costs   (4,172)   (3,562)   (3,500)   (	· ·		
Cash flows from investing activities:   Development of real estate   (56,690)   (63,332)     Proceeds from sale of real estate investments   (50,690)   (63,332)     Proceeds from sale of real estate investments   (50,690)   (63,332)     Repayment of notes receivable, net   (2,394)   (60,700)     Distributions received from investments in real estate partnerships   (11,166)   (51,500)     Net cash provided by (used in) investing activities   (75,600)     Net cash provided by (used in) investing activities   (74,432)     Net proceeds from common stock issuance   (74,432)     Redemption of exchangeable operating partnership units   (757)   (757)     Redemption of exchangeable operating partnership units   (757)   (757)     Distributions to exchangeable operating partnership unit holders   (757)   (757)     Distributions to exchangeable operating partnership unit holders   (2,112)   (5,08)     Dividends paid to common stockholders   (3,662)   (1,37)     Dividends paid to preferred unit holders   (3,662)   (1,37)     Dividends paid to preferred stockholders   (3,662)   (1,37)     Repayment of notes payable, net   (3,500)   (3,500)     Repayment of notes payable, net   (3,500)   (3,500)     Repayment of notes payable, net   (3,500)   (3,500)   (3,500)     Observed loan costs   (4,172)   (3,562)   (3,500)     Net (decrease) increase in cash and cash equivalents   (4,172)   (3,562)   (3,500)     Observed loan costs   (4,172)   (3,562)   (3,500)   (	Net cash provided by operating activities	19.970	33,913
Development of real estate   \$56,990   \$63,38     Proceeds from sale of real estate investments   \$9,240   \$0,85     Repayment of notes receivable, net   \$2,394   \$60     Investments in real estate partnerships   \$9,814   \$17,96     Distributions received from investments in real estate partnerships   \$9,814   \$17,96     Net cash provided by (used in) investing activities   \$3,592   \$14,43     Cash flows from financing activities:   Net proceeds from common stock issuance   \$3,128   \$10,76     Redemption of exchangeable operating partnership units   \$- (7,78     Contributions (abstributions) from limited partners in consolidated partnerships   \$2 9   \$0     Contributions to exchangeable operating partnership units   \$- (757   \$675     Distributions to exchangeable operating partnership unit holders   \$0,757   \$675     Distributions to preferred unit holders   \$0,757   \$675     Distributions to preferred unit holders   \$0,757   \$675     Dividends paid to common stockholders   \$0,454   \$0,177     Dividends paid to partnership units   \$0,854   \$0,177     Dividends paid to preferred stockholders   \$0,852   \$0,950     Repayments of notes payable, net   \$0,852   \$0,950     Repayment of notes payable, net   \$0,852   \$0,950     Repayment of notes payable, net   \$0,852   \$0,950     Scheduled principal payments   \$0,852   \$0,950     Repayment of notes payable, net   \$0,852   \$0,950     Scash and cash equivalents at beginning of the period   \$0,852   \$0,950     Scash and cash equivalents at end of the period   \$0,852   \$0,950     Cash and cash equivalents at end of the period   \$0,852   \$0,950     Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)   \$0,950   \$0,950     Supplemental disclosure of non-cash transactions:   \$0,950   \$0,950   \$0,950     Supplemental disclosure of non-cash transactions:   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0	to the Charles of the control of the		
Development of real estate   \$56,990   \$63,38     Proceeds from sale of real estate investments   \$9,240   \$0,85     Repayment of notes receivable, net   \$2,394   \$60     Investments in real estate partnerships   \$9,814   \$17,96     Distributions received from investments in real estate partnerships   \$9,814   \$17,96     Net cash provided by (used in) investing activities   \$3,592   \$14,43     Cash flows from financing activities:   Net proceeds from common stock issuance   \$3,128   \$10,76     Redemption of exchangeable operating partnership units   \$- (7,78     Contributions (abstributions) from limited partners in consolidated partnerships   \$2 9   \$0     Contributions to exchangeable operating partnership units   \$- (757   \$675     Distributions to exchangeable operating partnership unit holders   \$0,757   \$675     Distributions to preferred unit holders   \$0,757   \$675     Distributions to preferred unit holders   \$0,757   \$675     Dividends paid to common stockholders   \$0,454   \$0,177     Dividends paid to partnership units   \$0,854   \$0,177     Dividends paid to preferred stockholders   \$0,852   \$0,950     Repayments of notes payable, net   \$0,852   \$0,950     Repayment of notes payable, net   \$0,852   \$0,950     Repayment of notes payable, net   \$0,852   \$0,950     Scheduled principal payments   \$0,852   \$0,950     Repayment of notes payable, net   \$0,852   \$0,950     Scash and cash equivalents at beginning of the period   \$0,852   \$0,950     Scash and cash equivalents at end of the period   \$0,852   \$0,950     Cash and cash equivalents at end of the period   \$0,852   \$0,950     Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)   \$0,950   \$0,950     Supplemental disclosure of non-cash transactions:   \$0,950   \$0,950   \$0,950     Supplemental disclosure of non-cash transactions:   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0,950   \$0	Cash flows from investing activities:		
Proceeds from sale of real estate investments		(56,690)	(63 335)
Repayment of notes receivable, net         2,394         60.           Investments in real estate partnerships         (11,166)         61.           Distributions received from investments in real estate partnerships         3,592         (14,43)           Net cash provided by (used in) investing activities         3,592         (14,43)           Cash flows from financing activities:         8         10,76           Redemption of exchangeable operating partnership units         9         (7,78           Contributions (distributions) from limited partners in consolidated partnerships         9         (7,78           Contributions to preferred unit holders         (7,79         (67           Distributions to preferred unit holders         (3,66)         (3,73)           Dividends paid to common stockholders         (3,66)         (3,73)           Obvidends paid to preferred stockholders         (3,66)         (3,73)           (Repayments) proceeds from unsecured line of credit, net         (2,500)         35,000           (Repayment of notes payable, net         (5,21)         (5,81)           Deferred loan costs         (41,72)         13,66           Cash and cash equivalents at seginning of the period         \$ 3,32         29,86           Cash and cash equivalents at end of the period         \$ 3,32			
Investments in real estate partnerships   0,512   0,			
Distributions received from investments in real estate partnerships         9,814         17,966           Net cash provided by (used in) investing activities           Cash flows from financing activities:           Net proceeds from common stock issuance         3,128         10,766           Redeemption of exchangeable operating partnership units         29         67           Contributions (distributions) from limited partners in consolidated partnerships         29         67           Distributions to exchangeable operating partnership unit holders         (57)         (67           Distributions to preferred unit holders         (31,77)         (67           Dividends paid to common stockholders         (36,62)         (1,39)           Dividends paid to preferred stockholders         (3,662)         (1,39)           Repayment of notes payable, net         (35,2)         —           Scheduled principal payments         (65,201)         (5,81)           Pet cash used in financing activities         (65,201)         (5,81)           Post (decrease) increase in cash and cash equivalents         (41,72)         (3,662)           Cash and cash equivalents at beginning of the period         5,35,61         3,334           Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005			
Net cash provided by (used in) investing activities			
Cash flows from financing activities:         3,128         10,76           Net proceeds from common stock issuance         3,128         10,76           Redemption of exchangeable operating partnership units         29         7.78           Contributions (distributions) from limited partnership unit holders         (757)         (675)           Distributions to exchangeable operating partnership unit holders         (2,112)         (5,08           Distributions to preferred unit holders         (2,112)         (5,08           Dividends paid to preferred stockholders         (3,662)         (1,39           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (65,291)         (5,813           Net cash used in financing activities         (65,291)         (5,813 </td <td>Distributions received from investments in real estate partnersmps</td> <td>9,014</td> <td>17,900</td>	Distributions received from investments in real estate partnersmps	9,014	17,900
Cash flows from financing activities:         3,128         10,76           Net proceeds from common stock issuance         3,128         10,76           Redemption of exchangeable operating partnership units         29         7.78           Contributions (distributions) from limited partnership unit holders         (757)         (675)           Distributions to exchangeable operating partnership unit holders         (2,112)         (5,08           Distributions to preferred unit holders         (2,112)         (5,08           Dividends paid to preferred stockholders         (3,662)         (1,39           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayments) proceeds from unsecured line of credit, net         (65,291)         (5,813           Net cash used in financing activities         (65,291)         (5,813 </td <td>Not each provided by (used in) investing activities</td> <td>3 502</td> <td>(14.431)</td>	Not each provided by (used in) investing activities	3 502	(14.431)
Net proceeds from common stock issuance         3,128         10,76           Redemption of exchangeable operating partnership units         — (778           Contributions (distributions) from limited partners in consolidated partnerships         29         (6           Distributions to exchangeable operating partnership unit holders         (675)         (675)           Distributions to preferred unit holders         (3,127)         (3,127)           Dividends paid to common stockholders         (3,662)         (1,393)           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           Repayment of notes payable, net         (352)         —           Scheduled principal payments         (41,724)         (3,854)           Deferred loan costs         (427)         (3,354)           Net (decrease) increase in cash and cash equivalents         (65,291)         (5,811)           Cash and cash equivalents at beginning of the period         \$ 5,352         29,863           Cash and cash equivalents at end of the period         \$ 3,324         30,011           Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2,323         29,863           Common stock issued for partnership units exchanged         \$ 1,770         2,863           Real estate	ivet cash provided by (used iii) investing activities		(14,431)
Redemption of exchangeable operating partnership units         — (7.78-Contributions) from limited partners in consolidated partnerships         29 (2.75-Contributions) from limited partnership unit holders         (757- (675-Contributions) for exchangeable operating partnership unit holders         (757- (675-Contributions) for exchangeable operating partnership unit holders         (2,112)         (5,08)           Distributions to preferred unit holders         (3,662)         (1,39)           Dividends paid to common stockholders         (3,662)         (1,39)           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayment of notes payable, net         (352)         —           Scheduled principal payments         (427)         (3,350)           Scheduled principal payments         (65,291)         (5,813)           Net cash used in financing activities         (65,291)         (5,813)           Net (decrease) increase in cash and cash equivalents         (41,729)         13,669           Cash and cash equivalents at beginning of the period         5,350         29,869           Cash and cash equivalents at end of the period         5,351         43,534           Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)         31,246         30,013           Supplemental disclosur	Cash flows from financing activities:		
Redemption of exchangeable operating partnership units         — (7.78-Contributions) from limited partners in consolidated partnerships         29 (2.75-Contributions) from limited partnership unit holders         (757- (675-Contributions) for exchangeable operating partnership unit holders         (757- (675-Contributions) for exchangeable operating partnership unit holders         (2,112)         (5,08)           Distributions to preferred unit holders         (3,662)         (1,39)           Dividends paid to common stockholders         (3,662)         (1,39)           (Repayments) proceeds from unsecured line of credit, net         (25,000)         35,000           (Repayment of notes payable, net         (352)         —           Scheduled principal payments         (427)         (3,350)           Scheduled principal payments         (65,291)         (5,813)           Net cash used in financing activities         (65,291)         (5,813)           Net (decrease) increase in cash and cash equivalents         (41,729)         13,669           Cash and cash equivalents at beginning of the period         5,350         29,869           Cash and cash equivalents at end of the period         5,351         43,534           Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)         31,246         30,013           Supplemental disclosur		3,128	10,761
Contributions (distributions) from limited partners in consolidated partnerships		_	(7,784)
Distributions to exchangeable operating partnership unit holders		29	(2
Distributions to preferred unit holders   (2,112)   (5,08]     Dividends paid to common stockholders   (34,654)   (31,75)     Dividends paid to preferred stockholders   (36,62)   (1,73)     (Repayments) proceeds from unsecured line of credit, net   (25,000)   35,000     Repayment of notes payable, net   (352)   — (352)     Scheduled principal payments   (1,444)   (1,49)     Deferred loan costs   (427)   (3,354)     Net cash used in financing activities   (65,291)   (5,813)     Net (decrease) increase in cash and cash equivalents   (41,729)   13,663     Cash and cash equivalents at beginning of the period   (95,320)   29,863     Cash and cash equivalents at end of the period   (95,320)   29,863     Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)   (3,354   30,017     Supplemental disclosure of non-cash transactions:    Common stock issued for partnership units exchanged   (3,179)   (2,865   4,1770)   (3,865   4,1770)			
Dividends paid to common stockholders   (34,654   (31,776   Dividends paid to preferred stockholders   (3,662   (1,393   (1,393   (1,394	Distributions to preferred unit holders		
Dividends paid to preferred stockholders (Repayments) proceeds from unsecured line of credit, net (25,000) 35,000 Repayment of notes payable, net (352) — Scheduled principal payments (1,484) (1,495) Deferred loan costs (65,291) (5,813)  Net cash used in financing activities (65,291) (5,813)  Net (decrease) increase in cash and cash equivalents (41,729) 13,663  Cash and cash equivalents at beginning of the period (53,591) 43,538  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  Real estate contributed as investments in real estate partnerships  \$ 5,264 —			
(Repayments) proceeds from unsecured line of credit, net(25,000)35,000Repayment of notes payable, net(352)—Scheduled principal payments(1,484)(1,497)Deferred loan costs(427)(3,355)Net cash used in financing activities(65,291)(5,813)Net (decrease) increase in cash and cash equivalents(41,729)13,665Cash and cash equivalents at beginning of the period95,32029,865Cash and cash equivalents at end of the period\$ 53,59143,538Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)\$ 31,24630,017Supplemental disclosure of non-cash transactions:Common stock issued for partnership units exchanged\$ 1,7702,862Real estate contributed as investments in real estate partnerships\$ 5,264—	•		
Repayment of notes payable, net (352) — Scheduled principal payments (1,484) (1,491) Deferred loan costs (427) (3,351)  Net cash used in financing activities (65,291) (5,813)  Net (decrease) increase in cash and cash equivalents (41,729) 13,669  Cash and cash equivalents at beginning of the period 95,320 29,869  Cash and cash equivalents at end of the period \$53,591 43,530  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively) \$31,246 30,017  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged \$1,770 2,862  Real estate contributed as investments in real estate partnerships \$5,264 —			
Scheduled principal payments Deferred loan costs  (1,484) (1,497) (3,356)  Net cash used in financing activities (65,291) (5,813)  Net (decrease) increase in cash and cash equivalents (41,729) 13,669  Cash and cash equivalents at beginning of the period (29,860)  Cash and cash equivalents at end of the period (30,350)  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  \$ 1,770 2,860  Real estate contributed as investments in real estate partnerships  \$ 5,264  —			55,000
Net cash used in financing activities  Net (decrease) increase in cash and cash equivalents  Cash and cash equivalents at beginning of the period  Cash and cash equivalents at end of the period  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  \$ 1,770		` /	(1.407
Net cash used in financing activities (65,291) (5,813)  Net (decrease) increase in cash and cash equivalents (41,729) 13,669  Cash and cash equivalents at beginning of the period 95,320 29,869  Cash and cash equivalents at end of the period \$53,591 43,530  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively) \$31,246 30,013  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged \$1,770 2,863  Real estate contributed as investments in real estate partnerships \$5,264 —		* * * *	
Net (decrease) increase in cash and cash equivalents  Cash and cash equivalents at beginning of the period  95,320  29,865  Cash and cash equivalents at end of the period  \$53,591  43,536  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  \$1,770  2,865  Real estate contributed as investments in real estate partnerships  \$5,264  —	Deferred loan costs	(427)	(3,358)
Cash and cash equivalents at beginning of the period 95,320 29,866  Cash and cash equivalents at end of the period \$53,591 43,536  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively) \$31,246 30,017  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged \$1,770 2,867  Real estate contributed as investments in real estate partnerships \$5,264 —	Net cash used in financing activities	(65,291)	(5,813)
Cash and cash equivalents at beginning of the period 95,320 29,866  Cash and cash equivalents at end of the period \$53,591 43,536  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively) \$31,246 30,017  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged \$1,770 2,867  Real estate contributed as investments in real estate partnerships \$5,264 —			
Cash and cash equivalents at end of the period \$53,591 43,538  Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively) \$31,246 30,017  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged \$1,770 2,862  Real estate contributed as investments in real estate partnerships \$5,264 —	Net (decrease) increase in cash and cash equivalents	(41,729)	13,669
Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  \$ 1,770	Cash and cash equivalents at beginning of the period	95,320	29,869
Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and 2004, respectively)  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  \$ 1,770	Cash and cash equivalents at end of the period	\$ 53 591	43 538
2004, respectively)  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  Real estate contributed as investments in real estate partnerships  \$ 5,264 —	Cash and cash equivalents at end of the period	\$ 55,551	45,550
2004, respectively)  Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  Real estate contributed as investments in real estate partnerships  \$ 5,264 —	Supplemental disclosure of cash flow information - cash paid for interest (net of capitalized interest of \$2,721 and \$3,323 in 2005 and		
Supplemental disclosure of non-cash transactions:  Common stock issued for partnership units exchanged  \$ 1,770		\$ 31,246	30.017
Common stock issued for partnership units exchanged  Real estate contributed as investments in real estate partnerships  \$ 1,770			
Real estate contributed as investments in real estate partnerships \$ 5,264 —	Supplemental disclosure of non-cash transactions:		
	Common stock issued for partnership units exchanged	\$ 1,770	2,862
		ф. <b>-</b>	
Change in fair value of derivative instrument \$ — 175	Real estate contributed as investments in real estate partnerships	\$ 5,264	
Omninge in turn value of derivative instrument	Change in fair value of derivative instrument	\$	175
	Change in an value of activative instrument	Ψ	

# Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005 (unaudited)

#### 1. Summary of Significant Accounting Policies

#### (a) Organization and Principles of Consolidation

The accompanying consolidated financial statements include the accounts of Regency Centers Corporation and partnerships in which it has a majority ownership or controlling interest (the "Company" or "Regency"). All significant intercompany balances and transactions have been eliminated in the consolidated financial statements.

The Company owns approximately 98% of the outstanding common units ("Units") of Regency Centers, L.P. ("RCLP"). Regency invests in real estate through its partnership interest in RCLP. Generally all of the acquisition, development, operating and financing activities of Regency, including the issuance of Units and preferred units, are executed by RCLP. The equity interests of third parties held in RCLP or its majority owned partnerships are included in the consolidated financial statements as preferred units, exchangeable operating partnership units or limited partners' interest in consolidated partnerships. The Company is a qualified real estate investment trust ("REIT"), which began operations in 1993.

The consolidated financial statements reflect all adjustments that are of a normal recurring nature, and in the opinion of management, are necessary to properly state the Company's results of operations and financial position. Certain information and footnote disclosures normally included in annual financial statements prepared in accordance with U.S. generally accepted accounting principles have been condensed or omitted although management believes that the disclosures are adequate to make the information presented not misleading. The financial statements should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's December 31, 2004 Form 10-K filed with the Securities and Exchange Commission.

#### (b) Revenues

The Company leases space to tenants under agreements with varying terms. Leases are accounted for as operating leases with minimum rent recognized on a straight-line basis over the term of the lease regardless of when payments are due. Accrued rents are included in tenant receivables.

Substantially all of the lease agreements contain provisions that grant additional rents based on tenants' sales volume (contingent or percentage rent) and reimbursement of the tenants' share of real estate taxes, insurance and common area maintenance ("CAM") costs. Percentage rents are recognized when the tenants achieve the specified targets as defined in their lease agreements. Recovery of real estate taxes, insurance and CAM costs are recognized as the respective costs are incurred in accordance with the lease agreements.

The Company accounts for profit recognition on sales of real estate in accordance with Statement of Financial Accounting Standards ("SFAS") Statement No. 66, "Accounting for Sales of Real Estate." In summary, profits from sales will not be recognized by the Company unless a sale has been consummated; the buyer's initial and continuing investment is adequate to demonstrate a commitment to pay for the property; the Company has transferred to the buyer the usual risks and rewards of ownership; and the Company does not have substantial continuing involvement with the property.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (b) Revenues (continued)

The Company has been engaged by joint ventures to provide asset and property management services for such ventures' shopping centers. The fees are market based and generally calculated as a percentage of either revenues earned or the estimated values of the properties and are recognized as services are provided.

#### (c) Real Estate Investments

Land, buildings and improvements are recorded at cost. Tenant allowances are treated as tenant improvements. All specifically identifiable costs related to development activities are capitalized into properties in development on the consolidated balance sheet. The capitalized costs include predevelopment costs essential to the development of the property, development costs, construction costs, interest costs, real estate taxes, direct employee costs, and other costs incurred during the period of development.

The Company incurs costs prior to land acquisition including acquisition contract deposits, as well as legal, engineering and other external professional fees related to evaluating the feasibility of developing a shopping center. These pre-acquisition development costs are included in properties in development. If the Company determines that the development of a shopping center is no longer probable, any pre-development costs previously incurred are immediately expensed. At March 31, 2005 and December 31, 2004, the Company had capitalized pre-development costs of \$9.3 million and \$10.5 million, respectively.

The Company's method of capitalizing interest is based upon applying its weighted average borrowing rate to that portion of the actual development costs expended. The Company ceases cost capitalization when the property is available for occupancy upon substantial completion of tenant improvements. In no event would the Company capitalize interest on the project beyond 12 months after substantial completion of the building shell.

Maintenance and repairs that do not improve or extend the useful lives of the respective assets are reflected in operating and maintenance expense.

Depreciation is computed using the straight-line method over estimated useful lives of up to 40 years for buildings and improvements, term of lease for tenant improvements, and three to seven years for furniture and equipment.

The Company allocates the purchase price of assets acquired (net tangible and identifiable intangible assets) and liabilities assumed based on their relative fair values at the date of acquisition pursuant to the provisions of SFAS No. 141, "Business Combinations" ("Statement 141"). Statement 141 provides guidance on allocating a portion of the purchase price of a property to intangible assets. The Company's methodology for this allocation includes estimating an "as-if vacant" fair value of the physical property, which is allocated to land, building and improvements. The difference between the purchase price and the "as-if vacant" fair value is allocated to intangible assets. There are three categories of intangible assets to be considered: (i) value of in-place leases, (ii) above and below-market value of in-place leases and (iii) customer relationship value.

The value of in-place leases is estimated based on the value associated with the costs avoided in originating leases comparable to the acquired in-place leases as well as the value

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (c) Real Estate Investments (continued)

associated with lost rental and recovery revenue during the assumed lease-up period. The value of in-place leases is amortized to expense over the estimated weighted-average remaining lease lives.

Above-market and below-market in-place lease values for acquired properties are recorded based on the present value of the difference between (i) the contractual amounts to be paid pursuant to the in-place leases and (ii) management's estimates of fair market lease rates for the comparable in-place leases, measured over a period equal to the remaining non-cancelable term of the lease. The value of above-market leases is amortized as a reduction of base rental revenue over the remaining terms of the respective leases. The value of below-market leases is accreted as an increase to base rental revenue over the remaining terms of the respective leases, including renewal options.

The Company allocates no value to customer relationship intangibles if it has pre-existing business relationships with the major retailers in the acquired property because the customer relationships associated with the acquired property provide no incremental value over the Company's existing relationships.

The Company follows the provisions of SFAS No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets" ("Statement 144"). In accordance with Statement 144, the Company classifies an operating property as held for sale when it determines that the property is available for immediate sale in its present condition and management is reasonably certain that a sale will be consummated. Operating properties held for sale are carried at the lower of cost or fair value less costs to sell. Depreciation and amortization are suspended during the held-for-sale period. The operations of properties held for sale are reclassified into discontinued operations for all periods presented.

In accordance with Statement 144, when the Company sells a property and will not have continuing involvement after disposition, its operations and gain on sale are reported in discontinued operations when the operations and cash flows are clearly distinguished. Once classified as discontinued operations, these properties are eliminated from ongoing operations. Prior periods are also restated to reflect the operations of these properties as discontinued operations. When the Company sells operating properties to its joint ventures or to third parties, and it will have continuing involvement, the operations and gains on sales are included in income from continuing operations.

The Company reviews its real estate portfolio for impairment whenever events or changes in circumstances indicate that the carrying amount may not be recoverable based upon expected undiscounted cash flows from the property. The Company determines impairment by comparing the property's carrying value to an estimate of fair value based upon varying methods such as i) estimating future cash flows, ii) determining resale values by market, or iii) applying a capitalization rate to net operating income using prevailing rates in a given market. These methods of determining fair value can fluctuate significantly as a result of a number of factors, including changes in the general economy of those markets in which the Company operates, tenant credit quality and demand for new retail stores. In the event that the carrying amount of a property is not recoverable and exceeds its fair value, the Company will write down the asset to fair value for "held-and-used" assets and to fair value less costs to sell for "held-for-sale" assets.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (d) Deferred Costs

Deferred costs include deferred leasing costs and deferred loan costs, net of accumulated amortization. Such costs are amortized over the periods through lease expiration or loan maturity. Deferred leasing costs consist of internal and external commissions associated with leasing the Company's shopping centers. Net deferred leasing costs were \$30.5 million and \$30.8 million at March 31, 2005 and December 31, 2004, respectively. Deferred loan costs consist of initial direct and incremental costs associated with financing activities. Net deferred loan costs were \$10.0 million and \$10.2 million at March 31, 2005 and December 31, 2004, respectively.

#### (e) Earnings per Share and Treasury Stock

Basic net income per share of common stock is computed based upon the weighted average number of common shares outstanding during the period. Diluted net income per share also includes common share equivalents for stock options, restricted stock and exchangeable operating partnership units, if dilutive. See note 8 for the calculation of earnings per share ("EPS").

Repurchases of the Company's common stock (net of shares retired) are recorded at cost and are reflected as Treasury stock in the consolidated statement of stockholders' equity. Outstanding shares do not include treasury shares.

#### (f) Cash and Cash Equivalents

Any instruments which have an original maturity of 90 days or less when purchased are considered cash equivalents. Cash distributions of normal operating earnings from investments in real estate partnerships are included in cash flows from operations in the consolidated statements of cash flows.

#### (g) Estimates

The preparation of financial statements in conformity with U.S. generally accepted accounting principles requires the Company's management to make estimates and assumptions that affect the reported amounts of assets and liabilities, and disclosure of contingent assets and liabilities, at the date of the financial statements and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

#### (h) Stock-Based Compensation

Prior to January 1, 2005, the Company followed the provisions of SFAS No. 148, "Accounting for Stock-Based Compensation - Transition and Disclosure" ("Statement 148"). Statement 148 provided alternative methods of transition for a voluntary change to the fair value based method of accounting for stock-based employee compensation. In addition, Statement 148 amended the disclosure requirements of SFAS No. 123, "Accounting for Stock-Based Compensation" ("Statement 123"), to require more prominent and frequent disclosures in financial statements about the effects of stock-based compensation. As permitted under Statement 123 and Statement 148, the Company previously followed the accounting guidelines pursuant to Accounting Principles Board Opinion No. 25, "Accounting

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (h) Stock-Based Compensation (continued)

for Stock Issued to Employees" ("Opinion 25"), for stock-based compensation and furnished the pro-forma disclosures as required under Statement 148.

On December 16, 2004, the Financial Accounting Standards Board ("FASB") issued SFAS No. 123 (revised 2004), "Share-Based Payment" ("Statement 123(R)"), which is a revision of Statement 123. Statement 123(R) supersedes Opinion 25. Generally, the approach in Statement 123(R) is similar to the approach described in Statement 123. However, Statement 123(R) requires all share-based payments to employees, including grants of employee stock options, to be recognized in the statement of operations based on their fair values. Pro-forma disclosure is no longer an alternative under Statement 123(R).

Statement 123(R) is effective for fiscal years beginning after December 31, 2005. The Company elected early adoption of Statement 123(R) effective January 1, 2005. As permitted by Statement 123(R) the Company has applied the "modified prospective" method in which compensation cost is recognized beginning with the effective date (a) based on the requirements of Statement 123(R) for all share-based payments granted after the effective date and (b) based on the requirements of Statement 123 for all awards granted to employees prior to the effective date of Statement 123(R) that remain unvested on the effective date.

During the three months ended March 31, 2005, the Company recorded compensation expense of \$4.3 million which represents amortization of deferred compensation related to share based payments and a portion of which is capitalized to development projects. The expense includes \$274,301 related to stock options and \$4.0 for restricted stock and dividend equivalents. Deferred compensation is recorded as a reduction to additional paid in capital in the statement of stockholders' equity. Prior to 2005, as permitted by Statement 123, the Company accounted for share-based payments to employees using Opinion 25's intrinsic value method and recognized no compensation cost for employee stock options in prior years. Had the Company adopted Statement 123(R) in the prior period, the impact of that standard would have approximated the impact of Statement 123 in the disclosure of pro-forma net income and earnings per share described as follows (in thousands except per share data):

	March 31, 2004
Net income for common stockholders as reported:	\$21,420
Add: stock-based employee compensation expense included in reported net income	3,548
Deduct: total stock-based employee compensation expense determined under fair value based methods for all	
awards	4,729
Pro-forma net income	\$20,239
Earnings per share:	
Basic – as reported	\$ 0.36
Basic – pro-forma	\$ 0.34
Diluted – as reported	\$ 0.35
Diluted – pro-forma	\$ 0.33
-	

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (h) Stock-Based Compensation (continued)

The Company has a Long-Term Omnibus Plan (the "Plan") under which the Board of Directors may grant stock options and other stock-based awards to officers, directors and other key employees. The Plan allows the Company to issue up to 5.0 million shares in the form of common stock or stock options, but limits the issuance of common stock excluding stock options to no more than 2.75 million shares. At March 31, 2005, there were approximately 3.5 million shares available for grant under the Plan either through options or restricted stock of which 2.0 million shares are limited to common stock awards other than stock options. The Plan also limits outstanding awards to no more than 12% of outstanding common stock.

Stock options are granted under the Plan with an exercise price equal to the stock's fair market value at the date of grant. All stock options granted have ten-year lives, contain vesting terms of one to five years from the date of grant and some have dividend equivalent rights. Stock options granted prior to 2005 also contained "reload" rights, which allowed for an option holder to receive new options each time existing options were exercised if the existing options were exercised under specific criteria provided for in the Plan. Upon exercise of options under the Plan, the Company will issue new shares. In January 2005, the Company offered to acquire the "reload" rights of existing stock options from the option holders by issuing them additional stock options or restricted stock that will vest 25% per year and be expensed over a four-year period beginning in 2005 in accordance with Statement 123(R). As a result of the offer, on January 18, 2005, the Company granted 771,645 options to 37 employees with an exercise price of \$51.36, the fair value on the date of grant, and granted 7,906 restricted shares to 11 employees representing value of \$363,664, substantially canceling all of the "reload" rights on existing stock options. One employee chose to retain their reload rights. The stock option reload right buy-out program was not offered to the non-employee directors.

The fair value of each option award is estimated on the date of grant using the Black-Scholes-Merton closed-form model ("Black-Scholes") that uses the assumptions noted in the following table. Expected volatilities are based on historical volatility of the Company's stock. The Company uses historical data to estimate option exercises and employee terminations within the valuation model. The expected term of options granted is derived from the output of the option valuation model and represents the period of time that options granted are expected to be outstanding. The risk-free rate for periods within the contractual life of the option is based on the U.S. Treasury yield curve in effect at the time of grant.

The Company believes that the use of the Black-Scholes model meets the fair value measurement objectives of Statement 123(R) and reflects all substantive characteristics of the instruments being valued. The following table represents the assumptions used for the Black-Scholes option-pricing model to determine the per share weighted average fair value of \$5.90 for options granted during the period ended March 31, 2005:

Expected dividend yield	4.3%
Risk-free interest rate	3.7%
Expected volatility	18.0%
Expected life in years	4.4

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (h) Stock-Based Compensation (continued)

The following table reports stock option activity during the three month period ended March 31, 2005:

	Number of Shares	Weighted Average Exercise Price	Remaining Contractual Term	 ntrinsic Value housands)
Outstanding at January 1, 2005	1,675,163	\$ 44.32		
Granted	771,645	51.36		
Exercised	(90,557)	34.54		\$ 1,430
Outstanding at March 31, 2005	2,356,251	\$ 47.00	5.94	\$ 1,487
Exercisable at March 31, 2005	1,525,504	\$ 45.57	4.18	\$ 3,147

The following table presents information regarding unvested share activity during the three month period ended March 31, 2005:

	Non-vested Number of Shares	Av Gra	eighted verage ant-Date ir Value
Non-vested at January 1, 2005	59,102	\$	2.22
Granted	771,645	\$	5.90
Non-vested at March 31, 2005	830,747	\$	5.64

As of March 31, 2005, there was \$4.1 million of total unrecognized compensation cost related to non-vested share-based compensation arrangements related to stock options granted under the Plan. That cost is expected to be recognized over a weighted-average period of 3.8 years.

Restricted stock granted under the Plan generally vests over a period of four years, although certain grants cliff-vest after eight years, but contain provisions that allow for accelerated vesting over a shorter term if certain performance criteria are met. Compensation expense is measured at the grant date and recognized ratably over the vesting period. The Company considers the likelihood of meeting the performance criteria in determining the amount to expense on a periodic basis. In general, such criteria have historically been met, thus expense is recognized at a rate commensurate with the actual vesting period. Restricted stock grants also have certain dividend rights under the Plan, which are expensed in a manner similar to the underlying stock.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (h) Stock-Based Compensation (continued)

The following table reports restricted stock activity during the three month period ended March 31, 2005:

	Number of Shares	Intrinsic Value (in thousands
Unvested Shares at January 1, 2005	827,024	
Shares Granted	255,158	
Shares Vested and Distributed	(335,993)	\$ 16,501
Unvested Shares at March 31, 2005	746,189	\$ 35,541

As of March 31, 2005, there was \$27.1 million of total unrecognized compensation cost related to non-vested share-based compensation arrangements related to restricted stock granted under the Plan. That cost is expected to be recognized over a weighted-average period of 3.8 years.

The following table represents restricted stock granted in January 2005 and 2004 for individual and company level performance during 2004 and 2003, respectively. Shares granted in 2005 include 7,906 shares related to the stock option reload buy-out program discussed above.

	2005	2004
	<del></del>	
Fair value of stock at date of grant	\$ 51.19	39.97
4-year vesting grant	255,158	219,787
8-year vesting grant	<del></del>	64,649
Total stock grants	255,158	284,436
-		

The 4-year stock grants vest at the rate of 25% per year and the 8-year stock grants cliff- vest after eight years, but have the ability for accelerated vesting under the terms described above. Based upon restricted stock vesting for the three months ended March 31, 2005 and 2004, the Company recorded compensation expense of \$3.6 million and \$2.7 million, respectively, including the dividends vesting on restricted stock. During the three months ended March 31, 2005 and 2004, the Company recorded compensation expense for dividend equivalents related to stock options of \$404,840 and \$793,404 respectively, related to unexercised stock options. The Company also incurs stock based compensation related to fees paid to its Board of Directors, and for non-exempt employee anniversaries.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (i) Consolidation of Variable Interest Entities

In December 2003, the FASB issued Interpretation No. 46 ("FIN 46") (revised December 2003 ("FIN 46R")), "Consolidation of Variable Interest Entities", which addresses how a business enterprise should evaluate whether it has a controlling financial interest in an entity through means other than voting rights and accordingly should consolidate the entity. FIN 46R replaced FIN 46, which was issued in January 2003. FIN 46R was applicable immediately to a variable interest entity created after January 31, 2003 and as of the first interim period ending after March 15, 2004 to those variable interest entities created before February 1, 2003 and not already consolidated under FIN 46 in previously issued financial statements. The Company did not create any variable interest entities after January 31, 2003. The Company has adopted FIN 46R, analyzed the applicability of this interpretation to its structures and determined that they are not party to any significant variable interest entities.

#### (j) Segment Reporting

The Company's business is investing in retail shopping centers through direct ownership or through joint ventures. The Company actively manages its portfolio of retail shopping centers and may from time to time make decisions to sell lower performing properties, or developments not meeting its long-term investment objectives. The proceeds of sales are reinvested into higher quality retail shopping centers through acquisitions or new developments, which management believes will meet its planned rate of return. It is management's intent that all retail shopping centers will be owned or developed for investment purposes. The Company's revenue and net income are generated from the operation of its investment portfolio. The Company also earns incidental fees from third parties for services provided to manage and lease retail shopping centers owned through joint ventures.

The Company's portfolio is located throughout the United States; however, management does not distinguish or group its operations on a geographical basis for purposes of allocating resources or measuring performance. The Company reviews operating and financial data for each property on an individual basis, therefore, the Company defines an operating segment as its individual properties. No individual property constitutes more than 10% of the Company's combined revenue, net income or assets, and thus the individual properties have been aggregated into one reportable segment based upon their similarities with regard to both the nature of the centers, tenants and operational processes, as well as long-term average financial performance. In addition, no single tenant accounts for 10% or more of revenue and none of the shopping centers are located outside the United States.

#### (k) Derivative Financial Instruments

The Company adopted SFAS No. 133 "Accounting for Derivative Instruments and Hedging Activities" as amended by SFAS No. 149 ("Statement 133"). Statement 133 requires that all derivative instruments be recorded on the balance sheet at their fair value. Gains or losses resulting from changes in the values of those derivatives are accounted for depending on the use of the derivative and whether it qualifies for hedge accounting. The Company's use of derivative financial instruments is normally to mitigate its interest rate risk on a related financial instrument or forecasted transaction through the use of interest rate swaps.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (k) Derivative Financial Instruments (continued)

Statement 133 requires that changes in fair value of derivatives that qualify as cash flow hedges be recognized in other comprehensive income ("OCI") while the ineffective portion of the derivative's change in fair value be recognized immediately in earnings. Upon the settlement of a hedge, gains and losses associated with the transaction are recorded in OCI and amortized over the underlying term of the hedge transaction. Historically all of the Company's derivative instruments have qualified for hedge accounting.

To determine the fair value of derivative instruments, the Company uses standard market conventions and techniques such as discounted cash flow analysis, option pricing models and termination costs at each balance sheet date. All methods of assessing fair value result in a general approximation of value, and such value may never actually be realized.

#### (l) Financial Instruments with Characteristics of Both Liabilities and Equity

In May 2003, the FASB issued SFAS No. 150, "Accounting for Certain Financial Instruments with Characteristics of both Liabilities and Equity" ("Statement 150"). Statement 150 affects the accounting for certain financial instruments, which requires companies having consolidated entities with specified termination dates to treat minority owners' interests in such entities as liabilities in an amount based on the fair value of the entities. Although Statement 150 was originally effective July 1, 2003, the FASB has indefinitely deferred certain provisions related to classification and measurement requirements for mandatorily redeemable financial instruments that become subject to Statement 150 solely as a result of consolidation, including minority interests of entities with specified termination dates. As a result, Statement 150 has no impact on the Company's consolidated statements of operations for the period ended March 31, 2005.

At March 31, 2005, the Company held a majority interest in two consolidated entities with specified termination dates of 2017 and 2049. The minority owners' interests in these entities are to be settled upon termination by distribution or transfer of either cash or specific assets of the underlying entities. The estimated fair value of minority interests in entities with specified termination dates was approximately \$5.2 million at March 31, 2005 as compared to their carrying value of \$927,075. The Company has no other financial instruments that are affected by Statement 150.

#### (m) Recent Accounting Pronouncements

In December 2004, the FASB issued Statement No. 153, Exchange of Non-monetary Assets - an amendment of APB Opinion No 29 ("Statement 153"). The guidance in APB Opinion No. 29, Accounting for Non-monetary Transactions, is based on the principle that exchanges of non-monetary assets should be measured based on the fair value of the assets exchanged. The guidance in that Opinion, however, included certain exceptions to that principle. Statement 153 amends Opinion No. 29 to eliminate the exception for non-monetary assets that do not have commercial substance. A non-monetary exchange has commercial substance if the future cash flows of the entity are expected to change significantly as a result of the exchange. Statement 153 is effective for non-monetary asset exchanges occurring in fiscal periods beginning after June 15, 2005. The impact of adopting Statement 153 is not expected to have a material adverse impact on the Company's financial position or results of operations.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### (n) Reclassifications

Certain reclassifications have been made to the 2004 amounts to conform to classifications adopted in 2005.

#### Discontinued Operations

During the three months ended March 31, 2005, the Company sold 100% of its interest in two properties for net proceeds of \$34.7 million. The combined operating income and gains from these properties and properties classified as held for sale are included in discontinued operations. The revenues from properties included in discontinued operations were \$1.1 million and \$4.1 million for the three months ended March 31, 2005 and 2004, respectively. The operating income and gains from properties included in discontinued operations are reported net of minority interest of exchangeable operating partnership units and income taxes as follows:

	March	ı 31, 2005	March 31, 2004		
	Operating Income	Gain on sale of properties	Operating Income	Gain on sale of properties	
Operations and gain	\$ 468	11,182	1,774	12	
Less: Minority interest	9	198	29	_	
Less: Income taxes	53	2,188	_	_	
Discontinued operations, net	\$ 406	8,796	1,745	12	

#### Investments in Real Estate Partnerships

The Company accounts for all investments in which it owns 50% or less and does not have a controlling financial interest using the equity method. The Company's combined investment in these partnerships was \$180.5 million and \$179.7 million at March 31, 2005 and December 31, 2004, respectively. Any difference between the carrying amount of these investments and the underlying equity in net assets is amortized to equity in income of investments in real estate partnerships over the expected useful lives of the properties and other intangible assets which range in lives from 10 to 40 years. Net income from these partnerships, which includes all operating results, as well as gains and losses on sales of properties within the joint ventures, is allocated to the Company in accordance with the respective partnership agreements. Such allocations of net income are recorded in equity in income of investments in real estate partnerships in the accompanying consolidated statements of operations. Investments in real estate partnerships are primarily comprised of joint ventures where Regency invests with three co-investment partners, as further described below. In addition to earning its pro-rata share of net income in each of the partnerships, these co-investment partners pay the Company fees for asset management, property management, and acquisition and disposition services. During the three months ended March 31, 2005 and 2004, the Company received fees from these joint ventures of \$3.2 million and \$1.4 million, respectively.

The Company co-invests with the Oregon Public Employees Retirement Fund in three joint ventures (collectively "Columbia") in which it has ownership interests of 20% or 30%. As of March 31, 2005, Columbia owned 17 shopping centers, had total assets of \$483.2 million, and net

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### 3. Investments in Real Estate Partnerships (continued)

income of \$3.5 million. The Company's share of Columbia's total assets and net income was \$108.8 million and \$585,666, respectively. As of March 31, 2005, Columbia sold one shopping center to an unrelated party for \$19.7 million with a gain of \$326,443.

The Company co-invests with Macquarie CountryWide Trust of Australia ("MCW"), in two joint ventures (collectively, "MCWR") in which it has an ownership interest of 25%. As of March 31, 2005, MCWR owned 53 shopping centers, had total assets of \$780.7 million, and net income of \$4.1 million. Regency's share of MCWR's total assets and net income was \$195.2 million and \$1.2 million, respectively. As of March 31, 2005, MCWR acquired one shopping center from an unrelated party for a purchase price of \$24.4 million. The Company contributed \$4.5 million for its proportionate share of the purchase price, which was net of loan financing placed on the shopping center by MCWR. In addition, MCWR acquired one property from the Company valued at \$22.1 million, for which the Company received cash of \$17.1 million.

The Company co-invests with the California State Teachers' Retirement System ("CalSTRS") in a joint venture called ("RegCal") in which it has an ownership interest of 25%. As of March 31, 2005, RegCal owned four shopping centers, had total assets of \$126.6 million, and net income of \$725,611. The Company's share of RegCal's total assets and net income was \$31.7 million and \$210,567, respectively.

On February 14, 2005, Regency and MCW entered into a contract with CalPERS/First Washington to acquire 101 shopping centers operating in 17 states, located primarily in the Washington D.C./Baltimore metro area as well as northern and southern California ("FW Portfolio"). The contract purchase price is \$2.74 billion. The portfolio of shopping centers will be owned in a new joint venture between Regency and MCW ("MCWR II") in which the Company will have an ownership interest of 35%. The acquisition is expected to close during the second quarter of 2005. The Company expects to account for its investment in the venture as an unconsolidated investment in real estate partnerships, which it expects to approximate \$385 million.

The Company has executed a bank commitment to provide the financing for its share of the purchase price discussed further in note 5.

Recognition of gains from sales to joint ventures is recorded on only that portion of the sales not attributable to the Company's ownership interest. The gains and operations are not recorded as discontinued operations because of Regency's continuing involvement in these shopping centers. Columbia, MCWR and RegCal intend to continue to acquire retail shopping centers, some of which they may acquire directly from the Company. For those properties acquired from third parties, the Company is required to contribute its pro-rata share of the purchase price to the partnerships.

With the exception of Columbia, MCWR, and RegCal, all of which intend to continue expanding their investments in shopping centers, the investments in real estate partnerships represent single asset entities formed for the purpose of developing and owning retail shopping centers.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### 3. Investments in Real Estate Partnerships (continued)

The Company's investments in real estate partnerships as of March 31, 2005 and December 31, 2004 consist of the following (in thousands):

	Ownership	2005	2004
Macquarie CountryWide-Regency (MCWR)	25%	\$ 68,134	65,134
Macquarie CountryWide Direct (MCWR)	25%	7,944	8,001
Columbia Regency Retail Partners (Columbia)	20%	40,240	41,380
Cameron Village LLC (Columbia)	30%	21,348	21,612
Columbia Regency Partners II (Columbia)	20%	2,018	3,107
RegCal, LLC (RegCal)	25%	13,256	13,232
Other investments in real estate partnerships	50%	27,538	27,211
Investments in Real Estate Partnerships		\$180,478	179,677

Summarized financial information for the unconsolidated investments on a combined basis, is as follows (in thousands):

	March 31, 2005	December 31, 2004
Balance Sheet:		
Investment in real estate, net	\$1,352,587	1,320,871
Acquired lease intangibles, net	76,966	79,240
Other assets	43,306	39,506
Total assets	\$1,472,859	1,439,617
Notes payable	\$ 692,658	665,517
Other liabilities	28,616	24,471
Partners' equity	751,585	749,629
Total liabilities and equity	\$1,472,859	1,439,617

Unconsolidated investments in real estate partnerships had notes payable of \$692.7 million as of March 31, 2005 and the Company's proportionate share of these loans was \$175.2 million. The Company does not guarantee any debt of these partnerships and is responsible for only its pro-rata share based upon its ownership percentage.

The revenues and expenses for the unconsolidated investments on a combined basis are summarized as follows for the three months ended March 31, 2005 and 2004 (in thousands):

	2005	2004
Statements of Operations:		
Total revenues	\$40,636	22,463
Total expenses	31,748	16,007
Gain on sale of real estate	326	8,211
Net income	\$ 9,214	14,667

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### 4. Acquired Lease Intangible

Acquired lease intangible assets are net of accumulated amortization of \$3.5 million and \$2.6 million at March 31, 2005 and December 31, 2004, respectively. These assets have a remaining weighted average amortization period of approximately six years. The aggregate amortization expense from acquired leases was \$892,174 and \$490,820 for the three months ended March 31, 2005 and 2004, respectively. Acquired lease intangible liabilities are net of previously accreted minimum rent of \$2.1 million and \$1.9 million at March 31, 2005 and December 31, 2004, respectively and have a remaining weighted average amortization period of approximately six years.

#### 5. Notes Payable and Unsecured Line of Credit

The Company's outstanding debt at March 31, 2005 and December 31, 2004 consists of the following (in thousands):

	2005	2004
Notes Payable:		
Fixed rate mortgage loans	\$ 273,821	275,726
Variable rate mortgage loans	68,227	68,418
Fixed rate unsecured loans	948,991	948,946
Total notes payable	1,291,039	1,293,090
Unsecured line of credit	175,000	200,000
Total	\$1,466,039	1,493,090

The Company has an unsecured revolving line of credit (the "Line") with a commitment of \$500 million with the right to expand the Line by an additional \$150 million subject to additional lender syndication. The Line has a three-year term with a one-year extension option at an interest rate of LIBOR plus .75%. At March 31, 2005, the balance on the Line was \$175 million. Contractual interest rates on the Line, which are based on LIBOR plus .75%, were 3.5625% and 3.1875% at March 31, 2005 and December 31, 2004, respectively. The spread paid on the Line is dependent upon the Company maintaining specific investment-grade ratings. The Company is also required to comply, and is in compliance, with certain financial covenants such as Minimum Net Worth, Total Liabilities to Gross Asset Value ("GAV") and Secured Indebtedness to GAV and other covenants customary with this type of unsecured financing. The Line is used primarily to finance the development of real estate, but is also available for general working-capital purposes.

On February 15, 2005, the Company executed a commitment letter related to the Line which will temporarily modify certain Line covenants related to borrowing capacity and leverage, and will also add a temporary bridge loan for \$275 million ("Bridge Commitment"). The temporary modifications will expire and the Bridge Commitment will mature nine months after the closing of the FW Portfolio into MCWR II. The Bridge Commitment combined with existing borrowing capacity under the Line will provide sufficient cash for Regency's equity investment into MCWR II. These borrowings will raise the Company's debt to assets leverage ratio above current levels, which could exceed the current allowable Line covenants. The temporary modification to the leverage covenant is intended to keep Regency from defaulting on the Line during the term that the Bridge Commitment is outstanding. The Company intends to pay off the Bridge Commitment within the nine month term through a combination of issuing equity and selling shopping centers under our capital recycling program.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### 5. Notes Payable and Unsecured Line of Credit (continued)

Mortgage loans are secured by certain real estate properties and may be prepaid, but could be subject to a yield-maintenance premium. Mortgage loans are generally due in monthly installments of interest and principal and mature over various terms through 2017. Variable interest rates on mortgage loans are currently based on LIBOR plus a spread in a range of 125 to 150 basis points. Fixed interest rates on mortgage loans range from 5.01% to 9.50%.

The fair value of the Company's notes payable and Line are estimated based on the current rates available to the Company for debt of the same remaining maturities. Notes payable with variable interest rates and the Line are considered to be at fair value, since the interest rates on such instruments reprice based on current market conditions. Fixed rate loans assumed in connection with real estate acquisitions are recorded in the accompanying financial statements at fair value.

Based on the borrowing rates currently available to the Company for loans with similar terms and average maturities, the fair value of notes payable and the Line is \$1.5 billion.

As of March 31, 2005, scheduled principal repayments on notes payable and the Line were as follows (in thousands):

Scheduled Payments by Year	Scheduled Principal Payments	Term Loan Maturities	Total Payments
Current year	\$ 4,042	175,428	179,470
2006	3,775	20,950	24,725
2007 (includes the Line)	3,542	237,076	240,618
2008	3,388	19,535	22,923
2009	3,458	53,047	56,505
2010	3,919	177,092	181,011
Beyond 5 Years	13,271	744,242	757,513
Unamortized debt premiums		3,274	3,274
Total	\$ 35,395	1,430,644	1,466,039

#### 6. Derivative Financial Instruments

The Company is exposed to capital market risk, such as changes in interest rates. In order to manage the volatility relating to interest rate risk, the Company may enter into interest rate hedging arrangements from time to time. The Company does not utilize derivative financial instruments for trading or speculative purposes.

On April 1, 2005, the Company entered into three forward-starting interest rate swaps of approximately \$65.6 million each with fixed rates of 5.029%, 5.05% and 5.05%. The Company designated the \$196.8 million swaps as cash flow hedges to fix the rate on \$200 million of unsecured notes expected to be issued during July 2005, the proceeds of which will be used to repay \$100 million of fixed rate unsecured notes maturing on July 15, 2005, and to reduce the Line.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### 6. Derivative Financial Instruments (continued)

During 2003, the Company entered into two forward-starting interest rate swaps for a total of \$144.2 million to fix the rate on a refinancing in April 2004. On March 31, 2004, the Company settled the swaps with a payment to the counter-party for \$5.7 million. The swaps qualify for hedge accounting under Statement 133, therefore the losses associated with the swaps have been included in OCI and the unamortized balance is amortized as additional interest expense over the ten year term of the hedged loan.

#### 7. Stockholders' Equity and Minority Interest

#### (a) Preferred Stock

Terms and conditions of the preferred stock outstanding as of March 31, 2005 are summarized as follows:

Series	Shares Outstanding	Depositary Shares	Par Value	Distribution Rate	Callable by Company
Series 3	300,000	3,000,000	\$ 75,000,000	7.450%	04/03/08
Series 4	500,000	5,000,000	125,000,000	7.250%	08/31/09
	800,000	8,000,000	\$200,000,000		

These depositary shares are perpetual preferred stock, are not convertible into common stock of the Company, are redeemable at par upon our election five years after the issuance date, and have a liquidation value of \$25 per depositary share. The terms of the Series 3 and Series 4 Preferred Stock do not contain any unconditional obligations that would require us to redeem the securities at any time or for any purpose.

#### (b) Common Stock

On March 30, 2005, the Company entered into an agreement to sell 4,312,500 shares of its common stock to an affiliate of Citigroup Global Markets Inc. ("Citigroup"), in connection with a forward sale agreement (the "Forward Sale Agreement"). The Forward Sale Agreement, which closed on April 5, 2005, is expected to settle on or before August 1, 2005 at which time the Company will deliver the shares and receive approximately \$210 million in proceeds from Citigroup's sale of shares. The net proceeds are intended to be used to repay a portion of the balance of the \$275 million Bridge Commitment discussed in note 5.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### 7. Stockholders' Equity and Minority Interest (continued)

#### (c) Preferred Units

Terms and conditions of the Preferred Units outstanding as of March 31, 2005 are summarized as follows:

Series	Units Outstanding	Issue Price	Amount Outstanding	Distribution Rate	Callable by Company	Exchangeable by Unit holder
Series D	500,000	\$100.00	\$ 50,000,000	7.450%	09/29/09	01/01/14
Series E	300,000	\$100.00	30,000,000	8.750%	05/25/05	05/25/10
Series F	240,000	\$100.00	24,000,000	8.750%	09/08/05	09/08/10
	•					
	1,040,000		\$104,000,000			

The Preferred Units, which may be called by RCLP at par after certain dates, have no stated maturity or mandatory redemption, and pay a cumulative, quarterly dividend at fixed rates. The Preferred Units may be exchanged by the holder for Cumulative Redeemable Preferred Stock ("Preferred Stock") at an exchange rate of one share for one unit. The Preferred Units and the related Preferred Stock are not convertible into common stock of the Company. At March 31, 2005 and December 31, 2004, the face value of total Preferred Units issued was \$104 million with an average fixed distribution rate of 8.13%.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### 8. Earnings per Share

The following summarizes the calculation of basic and diluted earnings per share for the three months ended March 31, 2005 and 2004, respectively (in thousands except per share data):

	2005	2004
Numerator:		
Income from continuing operations	\$ 29,146	21,060
Income from continuing operations		•
Discontinued operations	9,202	1,757
Net income	38,348	22,817
Less: Preferred stock dividends	3,662	1,397
Less: Common dividends paid on unvested restricted stock	411	431
Net income for common stockholders – basic	34,275	20,989
Add: Common dividends paid on unvested restricted stock using the Treasury method	85	99
Add: Minority interest of exchangeable operating partnership units – continuing operations	_	355
Add: Minority interest of exchangeable operating partnership units – discontinued operations	_	30
Net income for common stockholders – diluted	\$ 34,360	21,473
Denominator:		
Weighted average common shares outstanding for basic EPS	62,328	59,550
Exchangeable operating partnership units	<u> </u>	1,152
Incremental shares to be issued under unvested restricted stock using the Treasury method	154	185
Incremental shares to be issued under common stock options using the Treasury method	80	347
Weighted average common shares outstanding for diluted EPS	62,562	61,234
The grade a verage common on the continuing for antice 27 o		01,20
Income per common share – basic		
Income from continuing operations	\$ 0.40	0.33
Discontinued operations	0.15	0.03
Net income for common stockholders per share	\$ 0.55	0.36
Income per common share – diluted		
Income from continuing operations	\$ 0.40	0.32
Discontinued operations	0.15	0.03
Net income for common stockholders per share	\$ 0.55	0.35

In the first quarter of 2005, the exchangeable operating partnership units were anti-dilutive to EPS, therefore, the units and the related minority interest of exchangeable operating partnership units are excluded from the calculation of EPS.

## Regency Centers Corporation Notes to Consolidated Financial Statements March 31, 2005

#### 9. Contingencies

The Company is involved in litigation on a number of matters and is subject to certain claims which arise in the normal course of business, none of which, in the opinion of management, is expected to have a material adverse effect on the Company's consolidated financial position, results of operations or liquidity. The Company is also subject to numerous environmental laws and regulations and is primarily concerned with dry cleaning plants that currently operate or have operated at its shopping centers. The Company believes that the tenants who currently operate plants do so in accordance with current laws and regulations, and that the ultimate disposition of currently known environmental matters will not have a material effect on its financial position. However, the Company has no assurance that existing environmental studies with respect to its shopping centers have revealed all potential environmental liabilities; that any previous owner, occupant or tenant did not create a material unknown environmental condition; that the current environmental condition of its properties will not be impacted by current occupants, nearby properties, or unrelated third parties; or that changes in or interpretations of environmental laws and regulations will not result in additional environmental liability.

#### Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

#### Forward Looking Statements

In addition to historical information, the following information contains forward-looking statements as defined under federal securities laws. These statements are based on current expectations, estimates and projections about the industry and markets in which Regency operates, and management's beliefs and assumptions. Forward-looking statements are not guarantees of future performance and involve certain known and unknown risks and uncertainties that could cause actual results to differ materially from those expressed or implied by such statements. Such risks and uncertainties include, but are not limited to, changes in national and local economic conditions; financial difficulties of merchants and retailers; competitive market conditions, including pricing of acquisitions and sales of properties and out-parcels; changes in expected leasing activity and market rents; timing of acquisitions, development starts and sales of properties and out-parcels; weather; the ability to obtain governmental approvals; and meeting development schedules. The following discussion should be read in conjunction with the accompanying Consolidated Financial Statements and Notes thereto of Regency Centers Corporation appearing elsewhere within.

#### Overview and Operating Philosophy

Regency is a qualified real estate investment trust ("REIT"), which began operations in 1993. Our primary operating and investment goal is long-term growth in earnings per share and total shareholder return, which we hope to achieve by focusing on a strategy of owning, operating and developing neighborhood and community shopping centers that are anchored by market-leading supermarkets and retailers located in areas with attractive demographics. We own and operate our shopping centers through our operating partnership, Regency Centers, L.P. ("RCLP"), in which we currently own approximately 98% of the operating partnership units. Regency's operating, investing and financing activities are generally performed by RCLP, its wholly owned subsidiaries and its joint ventures with third parties.

Currently, our real estate investment portfolio before depreciation totals \$4.6 billion with 288 shopping centers in 23 states, including approximately \$1.4 billion in real estate assets composed of 79 shopping centers owned by unconsolidated joint ventures in 17 states. [Portfolio information is presented (a) on a combined basis, including unconsolidated joint ventures ("Combined Basis"), (b) on a basis that excludes the unconsolidated joint ventures ("Consolidated Properties").] We believe that providing our shopping center portfolio information under these methods provides a more complete understanding of the properties that we own, including those that we partially own and for which we provide property and asset management services. At March 31, 2005, our gross leasable area ("GLA") on a Combined Basis totaled 33.6 million square feet and was 93.0% leased. The GLA for the Consolidated Properties totaled 24.2 million square feet and was 92.0% leased. The GLA for the Unconsolidated Properties totaled 9.4 million square feet and was 95.7% leased.

We earn revenues and generate operating cash flow by leasing space to grocers and retail side-shop tenants in our shopping centers. We experience growth in revenues by increasing occupancy and rental rates at currently owned shopping centers, and by acquiring and developing new shopping centers. A neighborhood center is a convenient, cost-effective distribution platform for food retailers. Grocery-anchored centers generate substantial daily traffic and offer sustainable competitive advantages to their tenants. This high traffic generates increased sales, thereby driving higher occupancy, rental rates and rental-rate growth for Regency, which we expect to sustain our growth in earnings per share and increase the value of our portfolio over the long term.

We seek a range of strong national, regional and local specialty tenants, for the same reason that we choose to anchor our centers with leading grocers. We have created a formal partnering process — the Premier Customer Initiative ("PCI") — to promote mutually beneficial relationships with our non-grocer specialty retailers. The objective of PCI is for Regency to build a base of specialty tenants who represent the "best-in-class" operators in their respective merchandising categories. Such tenants reinforce the consumer appeal and other strengths of a center's grocery anchor, help to stabilize a center's occupancy, reduce re-leasing downtime, reduce tenant turnover and yield higher sustainable rents.

We grow our shopping center portfolio through acquisitions and new shopping center development, where we acquire the land and construct the building. Development is customer driven, meaning we generally have an executed lease from the anchor before we start construction. Developments serve the growth needs of our market-leading grocers and anchors, and our specialty retail customers, resulting in modern shopping centers with long-term leases from the grocery anchors and produce attractive returns on our invested capital. This development process can require up to 36 months from initial land or redevelopment acquisition through construction, lease-up and stabilization of rental income, depending upon the size of the project. Generally, anchor tenants begin operating their stores prior to the completion of construction of the entire center, resulting in rental income during the development phase.

We intend to maintain a conservative capital structure to fund our growth programs without compromising our investment-grade ratings. Our approach is founded on our self-funding business model. This model utilizes center "recycling" as a key component. Our recycling strategy calls for us to re-deploy the proceeds from the sales of properties into new, higher-quality developments that we expect to generate sustainable revenue growth and more attractive returns. Our commitment to maintaining a high-quality shopping center portfolio dictates that we continually assess the value of all of our properties and sell those that no longer meet our long-term investment criteria.

Joint venturing of shopping centers also provides us with a capital source for new development, as well as the opportunity to earn fees for asset and property management services. As asset manager, we are engaged by our partners to apply similar operating, investment, and capital strategies to the portfolios owned by the joint ventures. Joint ventures grow their shopping center investments through acquisitions from third parties or direct purchases from Regency. Although selling properties to joint ventures reduces our ownership interest, we continue to share in the risks and rewards of centers that meet our long-term investment strategy. Regency is not subject to liability and has no obligations or guarantees of the joint ventures beyond its ownership percentage.

We have identified certain significant risks and challenges affecting our industry, and we are addressing them accordingly. An economic downturn could result in declines in occupancy levels at our shopping centers, which would reduce our rental revenues; however, we believe that our investment focus on grocery anchored shopping centers that provide daily necessities will minimize the impact of a downturn in the economy. Increased competition from super-centers such as Wal-Mart could result in grocery-anchor closings or consolidations in the grocery store industry. We closely monitor the operating performance and tenants' sales in our shopping centers that operate near super-centers as well as those tenants operating retail formats that are incurring significant changes in competition or business practice such as the video rental format. A slowdown in our shopping center development program could reduce operating revenues and gains from sales. We believe that developing shopping centers in markets with strong demographics with leading grocery stores will enable us to continue to maintain our development program at historical averages.

### Shopping Center Portfolio

The following tables summarize general operating statistics related to our shopping center portfolio, which we use to evaluate and monitor our performance. The portfolio information below is presented (a) on a Combined Basis, (b) for Consolidated Properties and (c) for Unconsolidated Properties:

	March 31, 2005	December 31, 2004
Number of Properties (a)	288	291
Number of Properties (b)	209	213
Number of Properties (c)	79	78
Properties in Development (a)	29	34
Properties in Development (b)	25	32
Properties in Development (c)	4	2
Gross Leaseable Area (a)	33,634,563	33,815,970
Gross Leaseable Area (b)	24,220,864	24,532,952
Gross Leaseable Area (c)	9,413,699	9,283,018
Percent Leased (a)	93.0%	92.7%
Percent Leased (b)	92.0%	91.2%
Percent Leased (c)	95.7%	96.7%

The Company seeks to reduce its operating and leasing risks through diversification which it achieves by geographically diversifying its shopping centers; avoiding dependence on any single property, market, or tenant, and owning a portion of its shopping centers through joint ventures.

The following table is a list of the shopping centers summarized by state and in order of largest holdings presented on a Combined Basis:

	March 31, 2005			December 31, 2004				
Location	# Properties	GLA	% of Total GLA	% Leased	# Properties	GLA	% of Total GLA	% Leased
California	50	6,461,701	19.2%	92.0%	51	6,527,802	19.3%	91.9%
Florida	49	5,865,451	17.4%	94.9%	50	5,970,898	17.7%	94.9%
Texas	32	3,949,235	11.7%	89.8%	32	3,968,940	11.7%	89.3%
Georgia	36	3,406,392	10.1%	95.0%	36	3,383,495	10.0%	97.4%
North Carolina	13	1,896,368	5.6%	92.5%	13	1,890,444	5.6%	94.2%
Ohio	14	1,876,129	5.6%	86.9%	14	1,876,013	5.5%	87.7%
Colorado	15	1,639,056	4.9%	97.2%	15	1,639,055	4.8%	98.0%
Virginia	12	1,473,949	4.4%	92.8%	12	1,488,324	4.4%	91.1%
Illinois	9	1,221,223	3.6%	97.9%	9	1,191,424	3.5%	98.0%
Washington	10	1,058,880	3.1%	98.3%	11	1,098,752	3.2%	97.6%
Oregon	8	838,056	2.5%	96.0%	8	838,056	2.5%	95.5%
Tennessee	7	700,534	2.1%	95.2%	7	697,034	2.1%	70.4%
Arizona	5	588,486	1.7%	92.9%	5	588,486	1.7%	93.1%
South Carolina	8	522,027	1.6%	95.5%	8	522,109	1.5%	95.7%
Michigan	4	368,348	1.1%	93.4%	4	368,348	1.1%	93.4%
Maryland	2	326,376	1.0%	93.9%	2	326,638	1.0%	93.9%
Alabama	4	324,044	1.0%	85.9%	4	324,044	1.0%	86.7%
Kentucky	2	302,670	0.9%	97.5%	2	302,670	0.9%	97.5%
Delaware	2	240,418	0.7%	99.9%	2	240,418	0.7%	99.9%
Pennsylvania	2	225,697	0.7%	100.0%	2	225,697	0.7%	100.0%
New Hampshire	2	140,488	0.4%	49.3%	2	138,488	0.4%	50.0%
Nevada	1	118,488	0.4%	58.7%	1	118,495	0.4%	45.5%
Indiana	1	90,547	0.3%	69.3%	1	90,340	0.3%	69.2%
Total	288	33,634,563	100.0%	93.0%	291	33,815,970	100.0%	92.7%

The following table is a list of the shopping centers summarized by state and in order of largest holdings presented for Consolidated Properties:

	March 31, 2005				December 3	31, 2004		
Location	# Properties	GLA	% of Total GLA	% Leased	# Properties	GLA	% of Total GLA	% Leased
California	42	5,338,149	22.0%	90.5%	44	5,479,470	22.3%	90.5%
Florida	37	4,578,852	18.9%	94.7%	38	4,684,299	19.1%	94.6%
Texas	29	3,632,633	15.0%	89.5%	29	3,652,338	14.9%	88.8%
Ohio	13	1,767,226	7.3%	86.2%	13	1,767,110	7.2%	87.1%
Georgia	17	1,659,194	6.9%	95.9%	17	1,656,297	6.8%	96.1%
Colorado	11	1,093,404	4.5%	96.7%	11	1,093,403	4.4%	97.6%
North Carolina	9	970,508	4.0%	96.1%	9	970,508	3.9%	97.5%
Virginia	8	911,116	3.8%	88.7%	8	925,491	3.8%	86.4%
Washington	8	707,568	2.9%	98.7%	9	747,440	3.0%	97.3%
Tennessee	6	636,534	2.6%	94.7%	6	633,034	2.6%	67.4%
Oregon	6	574,458	2.4%	96.7%	6	574,458	2.3%	96.1%
Arizona	4	480,839	2.0%	91.3%	4	480,839	2.0%	91.6%
Illinois	3	415,011	1.7%	96.9%	3	415,011	1.7%	97.4%
Michigan	4	368,348	1.5%	93.4%	4	368,348	1.5%	93.4%
Delaware	2	240,418	1.0%	99.9%	2	240,418	1.0%	99.9%
Pennsylvania	2	225,697	0.9%	100.0%	2	225,697	0.9%	100.0%
South Carolina	2	140,900	0.6%	85.0%	2	140,982	0.6%	85.7%
New Hampshire	2	140,488	0.6%	49.3%	2	138,488	0.6%	50.0%
Alabama	2	130,486	0.5%	94.4%	2	130,486	0.5%	97.3%
Nevada	1	118,488	0.5%	58.7%	1	118,495	0.5%	45.5%
Indiana	1	90,547	0.4%	69.3%	1	90,340	0.4%	69.2%
Total	209	24,220,864	100.0%	92.0%	213	24,532,952	100.0%	91.2%

The following table is a list of the shopping centers summarized by state and in order of largest holdings presented for Unconsolidated Properties owned in joint ventures:

	March 31, 2005			December 31, 2004				
Location	# Properties	GLA	% of Total GLA	% Leased	# Properties	GLA	% of Total GLA	% Leased
Georgia	19	1,747,198	18.6%	94.3%	19	1,727,198	18.6%	98.6%
Florida	12	1,286,599	13.7%	95.8%	12	1,286,599	13.8%	96.1%
California	8	1,123,552	11.9%	98.9%	7	1,048,332	11.3%	99.1%
North Carolina	4	925,860	9.8%	88.7%	4	919,936	9.9%	90.7%
Illinois	6	806,212	8.6%	98.4%	6	776,413	8.4%	98.3%
Virginia	4	562,833	6.0%	99.4%	4	562,833	6.1%	98.9%
Colorado	4	545,652	5.8%	98.3%	4	545,652	5.9%	98.7%
South Carolina	6	381,127	4.0%	99.3%	6	381,127	4.1%	99.3%
Washington	2	351,312	3.7%	97.7%	2	351,312	3.8%	98.1%
Maryland	2	326,376	3.5%	93.9%	2	326,638	3.5%	93.9%
Texas	3	316,602	3.4%	94.0%	3	316,602	3.4%	94.6%
Kentucky	2	302,670	3.2%	97.5%	2	302,670	3.3%	97.5%
Oregon	2	263,598	2.8%	94.3%	2	263,598	2.8%	94.3%
Alabama	2	193,558	2.1%	80.2%	2	193,558	2.1%	79.6%
Ohio	1	108,903	1.2%	97.6%	1	108,903	1.2%	96.1%
Arizona	1	107,647	1.1%	100.0%	1	107,647	1.1%	100.0%
Tennessee	1	64,000	0.7%	100.0%	1	64,000	0.7%	100.0%
Total	79	9,413,699	100.0%	95.7%	78	9,283,018	100.0%	96.7%

The following summarizes the four largest grocery tenants occupying our shopping centers at March 31, 2005:

Grocery Anchor	Number of Stores (a)	Percentage of Company- owned GLA (b)	Percentage of Annualized Base Rent (c)
Kroger	69	10.8%	7.9%
Publix	61	8.5%	4.9%
Safeway	54	6.5%	4.0%
Albertsons	27	3.1%	2.4%

- (a) For the Combined Properties including stores owned by grocery anchors that are attached to our centers.
- (b) For the Combined Properties.
- (c) Annualized base rent includes the Consolidated Properties plus Regency's pro-rata share of the Unconsolidated Properties which reflects our effective risk related to those tenants.

#### **Liquidity and Capital Resources**

#### General

We expect that cash generated from revenues, including gains from the sale of real estate, will provide the necessary funds on a short-term basis to pay our operating expenses, interest expense, scheduled principal payments on outstanding indebtedness, recurring capital expenditures necessary to maintain our shopping centers properly, and distributions to stock and unit holders. Net cash provided by operating activities was \$20.0 million and \$33.9 million for the three months ended March 31, 2005 and 2004, respectively. For the three months ended March 31, 2005 and 2004, our gains from the sale of real estate were \$17.7 million and \$4.0 million, we incurred capital expenditures of \$2.5 million and \$1.4 million to maintain our shopping centers, paid scheduled principal payments of \$1.5 million and \$1.5 million to our lenders, and paid dividends and distributions of \$41.2 million and \$38.9 million to our share and unit holders, respectively.

Although base rent is supported by long-term lease contracts, tenants who file bankruptcy are able to cancel their leases and close the related stores. In the event that a tenant with a significant number of leases in our shopping centers files bankruptcy and cancels its leases, we could experience a significant reduction in our revenues. On February 21, 2005, Winn-Dixie Stores, Inc. filed a voluntary petition for reorganization under Chapter 11 of the U.S. Bankruptcy Code. We currently lease six stores to Winn-Dixie, three of which are owned directly by us and three are owned in joint ventures. Our annualized base rent from Winn-Dixie including our share of the joint ventures is \$1.4 million or less than one-half of 1% of our annual base rents. Winn-Dixie currently owes Regency \$131,426 in prepetition rent related to common area expense reimbursements, and is current on all rent post-petition. Winn-Dixie continues to review their portfolio of stores, and has given us notice of rejection on one store owned in a joint venture. It is possible that Winn-Dixie may reject additional stores owned by us. We are not aware at this time of the current or pending bankruptcy of any of our other tenants that would cause a significant reduction in our revenues, and no tenant represents more than 8% of our annual base rental revenues.

We expect to meet long-term capital requirements for redeemable preferred units, maturing debt, the acquisition of real estate, investments in joint ventures, and the renovation or development of shopping centers from: (i) residual cash generated from operating activities after the payments described above, (ii) proceeds from the sale of real estate, (iii) joint venturing of real estate, (iv) refinancing of debt, and (v) equity raised in the private or public markets. We currently have \$198.6 million available for equity securities under our shelf registration and RCLP has \$180 million available for debt under its shelf registration.

We intend to continue to grow our portfolio through new developments and acquisitions, either directly or through our joint venture relationships. Because development and acquisition activities are

discretionary in nature, they are not expected to burden the capital resources we have currently available for liquidity requirements. Capital necessary to complete developments-in-process are funded from our line of credit. We expect that cash provided by operating activities, unused amounts available under our line of credit and cash reserves are adequate to meet short-term and committed long-term liquidity requirements.

#### Shopping Center Developments, Acquisitions and Sales

On a Combined Basis, we had 29 projects under construction or undergoing major renovations at March 31, 2005, which, when completed, will represent an investment of \$665.4 million before the estimated reimbursement of certain tenant-related costs and projected sales proceeds from adjacent land and outparcels of \$113.3 million. Costs necessary to complete these developments are estimated to be \$292.5 million. These costs are usually already committed as part of existing construction contracts, and will be expended through 2008. These developments are approximately 56% complete and 75% pre-leased. The costs necessary to complete these developments will be funded from the Company's unsecured line of credit, which had \$325 million of available funding at March 31, 2005.

During the three months ended March 31, 2005, the Company sold 100% of its interest in two properties for net proceeds of \$34.7 million. The combined operating income and gains from these properties and properties classified as held for sale are included in discontinued operations. The revenues from properties included in discontinued operations were \$1.1 million and \$4.1 million for the three months ended March 31, 2005 and 2004, respectively.

#### Off Balance Sheet Arrangements

Investments in Unconsolidated Real Estate Partnerships

At March 31, 2005, we had investments in real estate partnerships of \$180.5 million. The following is a summary of unconsolidated combined assets and liabilities of these partnerships, and our pro-rata share at March 31, 2005 and December 31, 2004 (in thousands):

	2005	2004
Number of Joint Ventures	11	11
Regency's Ownership	20%-50	20%-50
Number of Properties	79	78
Combined Assets	\$1,472,859	\$1,439,617
Combined Liabilities	721,274	689,988
Combined Equity	751,585	749,629
Regency's Share of:		
Assets	\$ 384,867	\$ 374,430
Liabilities	188,473	179,459

We account for all investments in which we own 50% or less and do not have a controlling financial interest using the equity method. Investments in real estate partnerships are primarily composed of joint ventures where we invest with three co-investment partners, as further described below. In addition to earning our pro-rata share of net income in each of these partnerships, these co-investment partners pay us fees for asset management, property management, and acquisition and disposition services. During the three months ended March 31, 2005 and 2004, we received fees from these joint ventures of \$3.2 million and \$1.4 million, respectively. Our investments in real estate partnerships as of March 31, 2005 and December 31, 2004 consist of the following (in thousands):

	Ownership	2005	2004
Macquarie CountryWide-Regency (MCWR)	25%	\$ 68,134	65,134
Macquarie CountryWide Direct (MCWR)	25%	7,944	8,001
Columbia Regency Retail Partners (Columbia)	20%	40,240	41,380
Cameron Village LLC (Columbia)	30%	21,348	21,612
Columbia Regency Partners II (Columbia)	20%	2,018	3,107
RegCal, LLC (RegCal)	25%	13,256	13,232
Other investments in real estate partnerships	50%	27,538	27,211
Total		\$180,478	179,677

We co-invest with the Oregon Public Employees Retirement Fund in three joint ventures (collectively "Columbia"), in which we have ownership interests of 20% or 30%. As of March 31, 2005, Columbia owned 17 shopping centers, had total assets of \$483.2 million, and net income of \$ 3.5 million. Our share of Columbia's total assets and net income was \$108.8 million and \$585,666, respectively. Columbia sold one shopping centers during the three months ended March 31, 2005 for \$19.7 million to an unrelated party with a gain of \$326,443.

We co-invest with Macquarie CountryWide Trust of Australia ("MCW") in two joint ventures (collectively, "MCWR") in which we have an ownership interest of 25%. As of March 31, 2005, MCWR owned 53 shopping centers, had total assets of \$780.7 million, and net income of \$4.1 million. Our share of MCWR's total assets and net income was \$195.2 million and \$1.2 million, respectively. During the three months ended March 31, 2005, MCWR acquired one shopping center from an unrelated party for a purchase price of \$24.4 million. We contributed \$4.5 million for our proportionate share of the purchase price, which was net of loan financing placed on the shopping center by MCWR. In addition, MCWR acquired one shopping center from us valued at \$22.1 million, for which we received cash of \$17.1 million.

We co-invest with the California State Teachers' Retirement System ("CalSTRS") in a joint venture called ("RegCal") in which we have a 25% ownership interest. As of March 31, 2005, RegCal owned four shopping centers, had total assets of \$126.6 million, and had net income of \$725,611. Our share of RegCal's total assets and net income was \$31.7 million and \$210,567, respectively.

On February 14, 2005, we entered into a contract with CalPERS/First Washington to acquire 101 shopping centers operating in 17 states, but primarily in the Washington D.C./Baltimore metro area, as well as, northern and southern California ("FW Portfolio"). The contract purchase price is \$2.74 billion. The portfolio of shopping centers will be owned in a new joint venture between Regency and MCW ("MCWR II") in which we will have an ownership interest of 35%. The acquisition is expected to close during the second quarter of 2005. Upon closing of the acquisition into the joint venture, MCWR II will pay us an acquisition fee and a due diligence fee based upon 1% and .125% of the net purchase price, respectively. We are currently in process of arranging long term financing for the joint venture, and upon loan closing, we will also be paid a debt placement fee equal to .50% of the debt amount arranged. We will only recognize fee income on that portion of the joint venture not owned by us. Approximately 52.5% of the acquisition fee will be subject to achieving certain targeted income levels in 2006 and 2007; and therefore, that portion of the acquisition fee will be deferred and only recognized in 2006 and 2007 if earned. We will also earn recurring fees for asset and property management on a quarterly and monthly basis, respectively. To assist in the transition of the portfolio to us, the seller, First Washington, has agreed to provide property management services for up to 2 years on approximately 50% of the portfolio which will result in lower property management fee income to us during the transition period. We expect to account for our equity investment in the venture as an unconsolidated investment in real estate partnerships, which we expect to approximate \$385 million. We have executed a bank commitment to provide the financing for our share of the purchase price further discussed below as a part of Notes Payable.

Recognition of gains from sales to joint ventures is recorded on only that portion of the sales not attributable to our ownership interest. The gains and operations are not recorded as discontinued operations because of our continuing involvement in these shopping centers. Columbia, MCWR and RegCal intend to continue to acquire retail shopping centers, some of which they may acquire directly from us. For those properties acquired from unrelated parties, we are required to contribute our pro-rata share of the purchase price to the partnerships.

Shopping center acquisitions, sales and the net acquisitions or sales activities within our investments in real estate partnerships are included in investing activities in the accompanying consolidated statements of cash flows. Net cash provided by investing activities was \$3.6 million for the three months ended March 31, 2005 and net cash used in investing activities was \$14.4 million for the three months ended March 31, 2004.

#### Notes Payable

Outstanding debt at March 31, 2005 and December 31, 2004 consists of the following (in thousands):

	2005	2004
Notes Payable:		
Fixed rate mortgage loans	\$ 273,821	275,726
Variable rate mortgage loans	68,227	68,418
Fixed rate unsecured loans	948,991	948,946
Total notes payable	1,291,039	1,293,090
Unsecured line of credit	175,000	200,000
Total	\$1,466,039	1,493,090

Mortgage loans are secured and may be prepaid, but could be subject to yield maintenance premiums. Mortgage loans are generally due in monthly installments of interest and principal, and mature over various terms through 2017. Variable interest rates on mortgage loans are currently based on LIBOR, plus a spread in a range of 125 to 150 basis points. Fixed interest rates on mortgage loans range from 5.01% to 9.50% and average 7.03%.

We have an unsecured revolving line of credit (the "Line") with a commitment of \$500 million, and we have the right to expand the Line by an additional \$150 million subject to additional lender syndication. The balance of the Line on March 31, 2005 was \$175 million. Contractual interest rates on the Line, which are based on LIBOR plus .75%, were 3.5625% at March 31, 2005 and 3.1875% at December 31, 2004. The spread that we pay on the Line is dependent upon maintaining specific investment-grade ratings. We are also required to comply, and are in compliance, with certain financial covenants such as Minimum Net Worth, Total Liabilities to Gross Asset Value ("GAV"), Secured Indebtedness to GAV and other covenants customary with this type of unsecured financing. The Line is used primarily to finance the development and acquisition of real estate, but is also available for general working-capital purposes.

On February 15, 2005, we executed a commitment letter related to the Line which will temporarily modify certain Line covenants related to our borrowing capacity and leverage, and will also add a temporary bridge loan for \$275 million ("Bridge Commitment"). The temporary modifications will expire and the Bridge Commitment will mature nine months after the closing of the FW Portfolio into MCWR II. The Bridge Commitment combined with existing borrowing capacity under the Line will provide sufficient cash for our equity investment into MCWR II. These borrowings will raise our debt to assets leverage ratio above current levels, which could exceed the current allowable Line covenant leverage ratio of 55%. The temporary modification to the leverage covenant is intended to keep us from defaulting on the Line during the term that the Bridge Commitment is outstanding. We intend to pay-off the Bridge Commitment within the nine month term through a combination of issuing equity and selling shopping centers under our capital recycling program.

As of March 31, 2005, scheduled principal repayments on notes payable and the Line were as follows (in thousands):

Scheduled Payments by Year	Scheduled Principal Payments	Term Loan Maturities	Total Payments
Current year	\$ 4,042	175,428	179,470
2006	3,775	20,950	24,725
2007 (includes the Line)	3,542	237,076	240,618
2008	3,388	19,535	22,923
2009	3,458	53,047	56,505
2010	3,919	177,092	181,011
Beyond 5 Years	13,271	744,242	757,513
Unamortized debt premiums		3,274	3,274
Total	\$ 35,395	1,430,644	1,466,039

Our investments in real estate partnerships had unconsolidated notes and mortgage loans payable of \$692.7 million at March 31, 2005, which mature through 2028. Our proportionate share of these loans was \$175.2 million, of which 85.1% had average fixed interest rates of 5.1% and 14.9% had variable interest rates based upon a spread above LIBOR of 1.0% to 1.4%. The loans are primarily non-recourse, but for those that are guaranteed by a joint venture, our guarantee does not extend beyond our ownership percentage of the joint venture.

We are exposed to capital market risk such as changes in interest rates. In order to manage the volatility related to interest-rate risk, we originate new debt with fixed interest rates, or we consider entering into interest-rate hedging arrangements. We do not utilize derivative financial instruments for trading or speculative purposes. We account for derivative instruments under Statement of Financial Accounting Standards ("SFAS") No. 133, "Accounting for Derivative Instruments and Hedging Activities" as amended ("Statement 133"). At March 31, 2005, 83% of our total debt had fixed interest rates, compared with 82% at December 31, 2004. We intend to limit the percentage of variable interest-rate debt to be no more than 30% of total debt, which we believe to be an acceptable risk. Based upon the variable interest-rate debt outstanding at March 31, 2005, if variable interest rates were to increase by 1%, our annual interest expense would increase by \$2.4 million.

On April 1, 2005, we entered into three forward-starting interest rate swaps of approximately \$65.6 million each with fixed rates of 5.029%, 5.05% and 5.05%. We designated the \$196.8 million swaps as cash flow hedges to fix the rate on \$200 million of unsecured notes expected to be issued during July 2005, the proceeds of which will be used to repay \$100 million of fixed rate unsecured notes maturing on July 15, 2005 and reduce the Line.

#### **Equity Capital Transactions**

From time to time, we issue equity in the form of exchangeable operating partnership units or preferred units of RCLP, or in the form of common or preferred stock of Regency Centers Corporation. As previously discussed, these sources of long-term equity financing allow us to fund our growth while maintaining a conservative capital structure. The following describes our equity capital transactions as of March 31, 2005.

#### Preferred Units

We have issued Preferred Units in various amounts since 1998, the net proceeds of which we used to reduce the balance of the Line. We issued Preferred Units primarily to institutional investors in private placements. The Preferred Units, which may be called by us in 2005 and 2009, have no stated maturity or mandatory redemption, and they pay a cumulative, quarterly dividend at fixed rates ranging from 7.45% to 8.75%. Generally, the Preferred Units may be exchanged by the holders for Cumulative

Redeemable Preferred Stock at an exchange rate of one share for one unit after ten years from the date of issue or as modified and agreed to by us. The Preferred Units and the related Preferred Stock are not convertible into Regency common stock. At March 31, 2005 and December 31, 2004, the face value of total Preferred Units issued was \$104 million, with an average fixed distribution rate of 8.13%. Included in Preferred Units are original issuance costs of \$2.2 million that will be expensed as the underlying Preferred Units are redeemed in the future. We expect to redeem \$54 million of Preferred Units during 2005 and expense their related issuance costs of \$1.4 million. The redemption price will be funded from either issuing a new series of preferred stock or from drawing on the Line.

### Preferred Stock

We currently have outstanding two series of Preferred stock. The Series 3 stock was issued in 2003 and represents 3 million depositary shares or 300,000 shares of 7.45% Series 3 Cumulative Redeemable Preferred Stock. The Series 4 stock was issued in 2004 and represents 5 million depositary shares or 500,000 shares of 7.25% Series 4 Cumulative Redeemable Preferred Stock. These depositary shares are perpetual preferred stock, are not convertible into common stock of the Company, are redeemable at par upon our election five years after the issuance date, and have a liquidation value of \$25 per depositary share. The terms of the Series 3 and Series 4 Preferred Stock do not contain any unconditional obligations that would require us to redeem the securities at any time or for any purpose.

### Common Stock

On March 30, 2005, we entered into an agreement to sell 4,312,500 shares of our common stock to an affiliate of Citigroup Global Markets Inc. ("Citigroup"), in connection with a forward sale agreement (the "Forward Sale Agreement"). The Forward Sale Agreement, which closed on April 5, 2005, is expected to settle on or before August 1, 2005 at which time we will deliver the shares and receive approximately \$210 million in proceeds from Citigroup's sale of shares. The net proceeds are intended to be used to repay a portion of the balance of the \$275 million Bridge Commitment discussed above under Notes Payable.

In December 2004, the Financial Accounting Standards Board ("FASB") issued SFAS No. 123 (revised 2004), "Share-Based Payment" ("Statement 123(R)"), which is a revision of Statement 123. Statement 123(R) supersedes Accounting Principles Board Opinion No. 25, "Accounting for Stock Issued to Employees" ("Opinion 25"). Statement 123(R) requires all share-based payments to employees, including grants of employee stock options, to be recognized in the statement of operations based on their fair values. Statement 123(R) is effective for fiscal years beginning after December 15, 2005; however, early adoption is permitted. The Company adopted Statement 123(R) effective on January 1, 2005. In accordance with Statement 123(R), the Company has applied the "modified prospective" method in which compensation cost is recognized beginning with the effective date (a) based on the requirements of Statement 123(R) for all share-based payments granted after the effective date and (b) based on the requirements of Statement 123 for all awards granted to employees prior to the effective date of Statement 123(R) that remain unvested on the effective date. For further understanding of our adoption of Statement 123(R) and its impact on our financial statements, see Footnote 1(h) in the accompanying consolidated financial statements.

In summary, net cash used in financing activities was \$65.3 million and \$5.8 million for the three months ended March 31, 2005 and 2004, respectively, related to the debt and equity activity discussed above and the investing activity also discussed within Liquidity and Capital Resources.

# Critical Accounting Policies and Estimates

Knowledge about our accounting policies is necessary for a complete understanding of our financial results, and discussion and analysis of these results. The preparation of our financial statements requires that we make certain estimates that impact the balance of assets and liabilities at a financial statement date and the reported amount of income and expenses during a financial reporting period. These accounting estimates are based upon, but not limited to, our judgments about historical

results, current economic activity and industry standards. They are considered to be critical because of their significance to the financial statements and the possibility that future events may differ from those judgments, or that the use of different assumptions could result in materially different estimates. We review these estimates on a periodic basis to ensure reasonableness. However, the amounts we may ultimately realize could differ from such estimates.

Revenue Recognition and Tenant Receivables – Tenant Receivables represent revenues recognized in our financial statements, and include base rent, percentage rent, and expense recoveries from tenants for common area maintenance costs, insurance and real estate taxes. We analyze tenant receivables, historical bad debt levels, customer credit worthiness and current economic trends when evaluating the adequacy of our allowance for doubtful accounts. In addition, we analyze the accounts of tenants in bankruptcy, and we estimate the recovery of pre-petition and post-petition claims. Our reported net income is directly affected by our estimate of the collectability of tenant receivables.

Capitalization of Costs - We have an investment services group with an established infrastructure that supports the due diligence, land acquisition, construction, leasing and accounting of our development properties. All direct costs related to these activities are capitalized. Included in these costs are interest and real estate taxes incurred during construction, as well as estimates for the portion of internal costs that are incremental and deemed directly or indirectly related to our development activity. If future accounting standards limit the amount of internal costs that may be capitalized, or if our development activity were to decline significantly without a proportionate decrease in internal costs, we could incur a significant increase in our operating expenses.

Real Estate Acquisitions - Upon acquisition of operating real estate properties, we estimate the fair value of acquired tangible assets (consisting of land, building and improvements), and identified intangible assets and liabilities (consisting of above- and below-market leases, in-place leases and tenant relationships) and assumed debt in accordance with SFAS No. 141, Business Combinations ("Statement 141"). Based on these estimates, we allocate the purchase price to the applicable assets and liabilities. We utilize methods similar to those used by independent appraisers in estimating the fair value of acquired assets and liabilities. We evaluate the useful lives of amortizable intangible assets each reporting period and account for any changes in estimated useful lives over the revised remaining useful life.

Valuation of Real Estate Investments - Our long-lived assets, primarily real estate held for investment, are carried at cost unless circumstances indicate that the carrying value of the assets may not be recoverable. We review long-lived assets for impairment whenever events or changes in circumstances indicate such an evaluation is warranted. The review involves a number of assumptions and estimates used to determine whether impairment exists. Depending on the asset, we use varying methods such as i) estimating future cash flows, ii) determining resale values by market, or iii) applying a capitalization rate to net operating income using prevailing rates in a given market. These methods of determining fair value can fluctuate significantly as a result of a number of factors, including changes in the general economy of those markets in which we operate, tenant credit quality and demand for new retail stores. If we determine that the carrying amount of a property is not recoverable and exceeds its fair value, we will write down the asset to fair value for "held-and-used" assets and to fair value less costs to sell for "held-for-sale" assets.

Discontinued Operations - The application of current accounting principles that govern the classification of any of our properties as held for sale on the balance sheet, or the presentation of results of operations and gains on the sale of these properties as discontinued, requires management to make certain significant judgments. In evaluating whether a property meets the criteria set forth by SFAS No. 144 "Accounting for the Impairment and Disposal of Long-Lived Assets" ("Statement 144"), the Company makes a determination as to the point in time that it can be reasonably certain that a sale will be consummated. Given the nature of all real estate sales contracts, it is not unusual for such contracts to allow potential buyers a period of time to evaluate the property prior to formal acceptance of the contract. In addition, certain other matters critical to the final sale, such as financing arrangements often remain pending even upon contract acceptance. As a result, properties under contract may not close within the

expected time period, or may not close at all. Due to these uncertainties, it is not likely that the Company can meet the criteria of Statement 144 prior to the sale formally closing. Therefore, any properties categorized as held for sale represent only those properties that management has determined are likely to close within the requirements set forth in Statement 144. The Company also makes judgments regarding the extent of involvement it will have with a property subsequent to its sale, in order to determine if the results of operations and gain on sale should be reflected as discontinued. Consistent with Statement 144, any property sold to an entity in which the Company has significant continuing involvement (most often joint ventures) is not considered to be discontinued. In addition, any property which the Company sells to an unrelated third party, but retains a property or asset management function, is also not considered discontinued. Therefore, only properties sold, or to be sold, to unrelated third parties that the Company, in its judgment, has no continuing involvement are classified as discontinued.

Income Tax Status - The prevailing assumption underlying the operation of our business is that we will continue to operate in order to qualify as a REIT, defined under the Internal Revenue Code. We are required to meet certain income and asset tests on a periodic basis to ensure that we continue to qualify as a REIT. As a REIT, we are allowed to reduce taxable income by all or a portion of our distributions to stockholders. We evaluate the transactions that we enter into and determine their impact on our REIT status. Determining our taxable income, calculating distributions, and evaluating transactions requires us to make certain judgments and estimates as to the positions we take in our interpretation of the Internal Revenue Code. Because many types of transactions are susceptible to varying interpretations under federal and state income tax laws and regulations, our positions are subject to change at a later date upon final determination by the taxing authorities.

### **Recent Accounting Pronouncements**

In December 2004, the FASB issued Statement No. 153, Exchange of Non-monetary Assets - an amendment of APB Opinion No 29 ("Statement 153"). The guidance in APB Opinion No. 29, Accounting for Non-monetary Transactions, is based on the principle that exchanges of non-monetary assets should be measured based on the fair value of the assets exchanged. The guidance in that Opinion, however, included certain exceptions to that principle. Statement 153 amends Opinion No. 29 to eliminate the exception for non-monetary assets that do not have commercial substance. A non-monetary exchange has commercial substance if the future cash flows of the entity are expected to change significantly as a result of the exchange. Statement 153 is effective for non-monetary asset exchanges occurring in fiscal periods beginning after June 15, 2005. The impact of adopting Statement 153 is not expected to have a material adverse impact on the Company's financial position or results of operations.

### Results from Operations

Comparison of the three months ended March 31, 2005 to 2004

At March 31, 2005, on a Combined Basis, we were operating or developing 288 shopping centers, as compared to 291 shopping centers at the end of 2004. We identify our shopping centers as either development properties or stabilized properties. Development properties are defined as properties that are in the construction or initial lease-up process and have not reached their initial full occupancy (reaching full occupancy generally means achieving at least 93% leased and rent paying on newly constructed or renovated GLA). At March 31, 2005, on a Combined Basis, we were developing 29 properties, as compared to 34 properties at the end of 2004.

Our revenues increased by \$9.2 million, or 10%, to \$101.7 million in 2005. This increase was related to changes in occupancy in the portfolio of stabilized and development properties, growth in re-leasing rental rates, operating properties acquired subsequent to March 31, 2004 that had no revenues during the comparable prior year period, revenues from new developments commencing operations subsequent to March 31, 2004, less a reduction in revenues from properties sold not classified as discontinued operations. In addition to collecting minimum rent from our tenants for the GLA that they lease from us, we also collect contingent rent based upon tenant sales, which we refer to as percentage rent. Tenants are also responsible for reimbursing us for their pro-rata share of the expenses associated

with operating our shopping centers. In 2005, our minimum rent increased by \$5.5 million, or 8%, and our recoveries from tenants increased \$2.3 million, or 12%. Percentage rent was \$551,539 in 2005, compared with \$452,485 million in 2004.

Our operating expenses increased by \$6.4 million, or 13%, to \$55.2 million in 2005 related to increased operating and maintenance costs, general and administrative costs and depreciation expense, as further described below.

Our combined operating, maintenance, and real estate taxes increased by \$1.2 million, or 5%, for the three months ended March 31, 2005 to \$24.1 million. This increase was primarily due to shopping centers acquired during 2004 and new developments that only recently began operating and therefore did not incur operating expenses during the three month period ending on March 31, 2004.

Our general and administrative expenses were \$8.7 million for the three months ended March 31, 2005, compared with \$5.9 million in 2004, or 47% higher, related to an increase in the total number of employees, higher costs associated with incentive compensation and costs related to implementing new regulations for public companies imposed by the Sarbanes-Oxley Act.

Our depreciation and amortization expense increased \$1.4 million for the three months ended March 31, 2005 primarily related to new development properties placed in service and operating properties acquired subsequent to March 31, 2004 that had no operations during the comparable prior year period.

Our net interest expense was \$21.1 million for the three months ended March 31, 2005 and 2004. Average interest rates on our outstanding debt decreased to 6.41% at March 31, 2005, compared with 6.45% at March 31, 2004. Our weighted average outstanding debt at March 31, 2005 and 2004 was \$1.5 billion.

We account for profit recognition on sales of real estate in accordance with SFAS Statement No. 66, "Accounting for Sales of Real Estate." Profits from sales of real estate will not be recognized by us unless (i) a sale has been consummated; (ii) the buyer's initial and continuing investment is adequate to demonstrate a commitment to pay for the property; (iii) we have transferred to the buyer the usual risks and rewards of ownership; and (iv) we do not have substantial continuing involvement with the property. Gains from the sale of operating and development properties includes \$2.1 million in gains from the sale of six out-parcels for proceeds of \$4.2 million and \$4.4 million in gains related to the sale of two development properties. For the period ended March 31, 2004, the gains from the sale of operating and development properties included \$3.4 million from the sale of seven out-parcels for proceeds of \$9.5 million and \$629,905 in gains for properties sold. These gains are included in continuing operations rather than discontinued operations because they were either properties that had no operating income, or they were properties sold to joint ventures where we have continuing involvement through our minority investment.

Income from discontinued operations was \$9.2 million in 2005 related to two properties sold to unrelated parties for net proceeds of \$34.7 million and two properties classified as held-for-sale. Income from discontinued operations was \$1.8 million in 2004 related to the operations of shopping centers sold or classified as held-for-sale in 2005 as well as 2004. In compliance with the adoption of Statement 144, if we sell an asset in the current year, we are required to reclassify its operating income into discontinued operations for all prior periods. This practice results in a reclassification of amounts previously reported as continuing operations into discontinued operations. Our income from discontinued operations is shown net of minority interest of exchangeable partnership units and income taxes.

Minority interest of preferred units declined \$3.0 million to \$2.1 million in 2005 as a result of redeeming \$125 million of preferred units during 2004 in combination with negotiating a lower rate on \$50 million of preferred units. Preferred stock dividends increased \$2.3 million to \$3.7 million in 2005 as a result of issuing \$125 million of Series 4 preferred stock during 2004, a portion of the proceeds of which were used to redeem the preferred units.

Net income for common stockholders was \$34.7 million in 2005, compared with \$21.4 million in 2004 or a 62% increase for the reasons described above. Diluted earnings per share were \$0.55 in 2005, compared with \$0.35 in 2004, or 57% higher.

#### **Environmental Matters**

We are subject to numerous environmental laws and regulations and we are primarily concerned with dry cleaning plants that currently operate or have operated at our shopping centers in the past. We believe that the tenants who currently operate plants do so in accordance with current laws and regulations. Generally, we use all legal means to cause tenants to remove dry cleaning plants from our shopping centers or convert them to environmentally approved systems. Where available, we have applied and been accepted into state-sponsored environmental programs. We have a blanket environmental insurance policy that covers us against third-party liabilities and remediation costs on shopping centers that currently have no known environmental contamination. We have also placed environmental insurance, where possible, on specific properties with known contamination, in order to mitigate our environmental risk. We believe that the ultimate disposition of currently known environmental matters will not have a material effect on Regency's financial position, liquidity, or operations; however, we can give no assurance that existing environmental studies with respect to our shopping centers have revealed all potential environmental liabilities; that any previous owner, occupant or tenant did not create any material environmental condition not known to us; that the current environmental condition of the shopping centers will not be affected by tenants and occupants, by the condition of nearby properties, or by unrelated third parties; or that changes in applicable environmental laws and regulations or their interpretation will not result in additional environmental liability to us.

#### Inflation

Inflation has remained relatively low and has had a minimal impact on the operating performance of our shopping centers; however, substantially all of our long-term leases contain provisions designed to mitigate the adverse impact of inflation. Such provisions include clauses enabling us to receive percentage rent based on tenants' gross sales, which generally increase as prices rise; and/or escalation clauses, which generally increase rental rates during the terms of the leases. Such escalation clauses are often related to increases in the consumer price index or similar inflation indices. In addition, many of our leases are for terms of less than ten years, which permits us to seek increased rents upon re-rental at market rates. Most of our leases require tenants to pay their share of operating expenses, including common-area maintenance, real estate taxes, insurance and utilities, thereby reducing our exposure to increases in costs and operating expenses resulting from inflation.

### Item 3. Quantitative and Qualitative Disclosures about Market Risk

### Market Risk

We are exposed to interest-rate changes primarily related to the variable interest rate on the Line and the refinancing of long-term debt, which currently contain fixed interest rates. Our interest-rate risk management objective is to limit the impact of interest-rate changes on earnings and cash flows and to lower our overall borrowing costs. To achieve these objectives, we borrow primarily at fixed interest rates and may enter into derivative financial instruments such as interest-rate swaps, caps and treasury locks in order to mitigate our interest-rate risk on a related financial instrument. We have no plans to enter into derivative or interest-rate transactions for speculative purposes.

Our interest-rate risk is monitored using a variety of techniques. The table below presents the principal cash flows (in thousands), weighted average interest rates of remaining debt, and the fair value of total debt (in thousands), by year of expected maturity to evaluate the expected cash flows and sensitivity to interest-rate changes.

	2005	2006	2007	2008	2009	2010	Thereafter	Total	Fair Value
Fixed rate debt \$	147.181	24,725	29.680	22.923	56,505	181,011	757,513	1,219,538	1,249,142
Average interest rate for all fixed rate debt	7.08%	, -	- /	,		,	6.10%	_	_,,
Variable rate LIBOR debt \$	32,289	_	210,938	_	_	_	_	243,227	243,227
Average interest rate for all variable rate debt	3.04%		3.04%						

As the table incorporates only those exposures that exist as of March 31, 2005, it does not consider those exposures or positions that could arise after that date. Moreover, because firm commitments are not presented in the table above, the information presented above has limited predictive value. As a result, our ultimate realized gain or loss with respect to interest-rate fluctuations will depend on the exposures that arise during the period, our hedging strategies at that time, and actual interest rates.

### **Item 4. Controls and Procedures**

Under the supervision and with the participation of the Company's management, including the Company's Chief Executive Officer, Chief Operating Officer and Chief Financial Officer, the Company has evaluated the effectiveness of the design and operation of its disclosure controls and procedures as of the end of the period covered by this report, and, based on their evaluation, the Chief Executive Officer, Chief Operating Officer and Chief Financial Officer have concluded that these disclosure controls and procedures are effective. There have been no changes in the Company's internal controls over financial reporting identified in connection with this evaluation that occurred during the period covered by this report and that have materially affected, or are reasonably likely to materially affect, the Company's internal controls over financial reporting.

### PART II - OTHER INFORMATION

# **Item 1. Legal Proceedings**

We are a party to various legal proceedings, which arise, in the ordinary course of our business. We are not currently involved in any litigation nor to our knowledge, is any litigation threatened against us, the outcome of which would, in our judgment based on information currently available to us, have a material adverse effect on our financial position or results of operations.

# Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

(a) We sold the following equity securities during the quarter ended March 31, 2005 that we did not report on Form 8-K because they represent in the aggregate less than 1% of our outstanding common stock. All shares were issued to a total of four accredited investors in transactions exempt from registration pursuant to Section 4(2) of the Securities Act of 1933, in exchange for an equal number of common units of our operating partnership, Regency Centers, L.P.

Date	Number of Shares
<del></del>	
1/3/05	32,528
1/5/05	4,800
1/14/05	6,000
1/25/05	23,513

(b) None

(c) Issuer Purchases of Equity Securities

Period	Total number of shares purchased (1)	(a) Average price paid per share	(b) Total number of shares purchased as part of publicly announced plans or programs	(d) Maximum number or approximate dollar value of shares that may yet be purchased under the plans or programs
January 1 through January 31, 2005	<del>_</del>	_	<del>_</del>	_
February 1 through February 28, 2005	141,245	\$ 50.29	_	_
March 1 through March 31, 2005	_	_	<del>-</del>	_
Total	141,245	\$ 50.29	<del>-</del>	_

Represents shares delivered in payment of withholding taxes in connection with restricted stock vesting by participants under Regency's Long-Term Omnibus Plan.

Item 3. None

Item 4. None

Item 5. None

### Item 6. Exhibits

- (a) Exhibits:
  - 10. Material Contracts
    - 10.1 Purchase and Sale Agreement dated as of February 14, 2005 by and among Regency, Macquarie CountryWide-Regency II, LLC, Macquarie CountryWide Trust, USRP Texas GP, LLC, U.S. Retail Partners, LLC, Eastern Shopping Centers Holdings, LLC, First Washington Investment I, LLC and California Public Employees' Retirement System
    - 10.2 First Amendment dated as of March 28, 2005 to Amended and Restated Credit Agreement by and among Regency Centers, L.P., as Borrower, Regency, each of the Lenders signatory thereto, and Wells Fargo Bank, National Association, as Agent (incorporated by reference to Exhibit 10.1 to the Company's Form 8-K filed April 1, 2005)
  - 31.1 Rule 13a-14 Certification of Chief Executive Officer.
  - 31.2 Rule 13a-14 Certification of Chief Financial Officer.
  - 31.3 Rule 13a-14 Certification of Chief Operating Officer.
  - 32.1 Section 1350 Certification of Chief Executive Officer.
  - 32.2 Section 1350 Certification of Chief Financial Officer.
  - 32.3 Section 1350 Certification of Chief Operating Officer.

# SIGNATURE

Pursuant to the requirements of the Securities and Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

Date: May 9, 2005 REGENCY CENTERS CORPORATION

By: /s/ J. Christian Leavitt

Senior Vice President and Principal Accounting Officer

42

PURCHASE AND SALE AGREEMENT

**AMONG** 

MACQUARIE COUNTRYWIDE-REGENCY II, LLC

MACQUARIE COUNTRYWIDE TRUST

REGENCY CENTERS CORPORATION

USRP TEXAS GP, LLC

U.S. RETAIL PARTNERS, LLC

EASTERN SHOPPING CENTERS HOLDINGS, LLC

FIRST WASHINGTON INVESTMENT I, LLC

AND

CALIFORNIA PUBLIC EMPLOYEES' RETIREMENT SYSTEM

DATED FEBRUARY 14, 2005

# TABLE OF CONTENTS

		Page
1.	1. PURCHASE AND SALE OF MEMBERSHIP INTERESTS	1
2.	2. PURCHASE PRICE – PAYMENT	1
	2.1. Purchase Price	1
	2.2. Price Adjustments	2
	2.3. Deposit.	3
	2.4. Closing	4
	2.5. Escrows at Closing	4
	2.6. Payments at Closing	5
	2.7. Payment Allocation	5
	2.8. Tax Matters; Purchase Price Allocation	5
	2.9. Closing Costs	6
	2.10. Excluded Assets	6
	2.11. Edgewater Retail Partners LLC	6
	2.12. Company Group's Pre-Closing Accounts Receivable and Payables	8
	2.13. Dropping Properties	9
	2.14. Purchase of Option Parcels	10
	2.15. Purchase of Maryland Permanent Bank Parcel	10
3.	3. PRORATIONS	11
	3.1. Prorations	11
4.	4. BUYER'S RIGHT OF INSPECTION; INSPECTION PERIOD	18
	4.1. Right to Evaluate	18
	4.2. Inspection Obligations	19
	4.3. Sellers' Deliveries	19
	4.4. No Other Representations and Warranties Outside Agreement and Transaction Documents	19
	4.5. Limited Release of Sellers	21
5.	5. TITLE, SURVEY AND ENVIRONMENTAL MATTERS	21
	5.1. Title and Survey Matters	21
	5.2. Resolution of Other Exceptions	24
	5.3. Sellers' Obligations Regarding Title Objections	25
	5.4. Postponement of Closing	26
	5.5. Cure of Title Objections	26
	5.6. Environmental Remediation and Insurance	26
6.	5. RISK OF LOSS	31
	6.1. Condemnation	31

31

6.2.

Casualty

7.	COVENANTS. 7.1. Implementing Agreement	32 32
	7.2. General Covenants	32
	7.3. Special Covenants	33
	.s. Special Covenants	55
8.	OINT AND SEVERAL REPRESENTATIONS AND WARRANTIES OF THE SELLERS	37
	3.1. Organization; Authorization	37
	3.2. Sellers	40
	3.3. Ownership of Properties	40
	3.4. No Encumbrances	40
	3.5. No Violation	40
	3.6. Financial Statements	41
	3.7. Absence of Undisclosed Liabilities	41
	3.8. Tax Matters	41
	3.9. Litigation	43
	3.10. No Employees	43
	3.11. No Condemnation	43
	3.12. Environmental Matters	44
	3.13. Leasing Commissions and Tenant Allowances	44
	3.14. Rent Rolls; Leases	44
	3.15. Accounts Receivable Aging	44
	3.16. Service Contracts	44
	3.17. Material Contracts	45
	3.18. Existing Loans	45
	3.19. Ground Leases	45
	3.20. No Other Contracts	46
	3.21. No Brokers or Finders	46
	3.22. Insurance	46
	3.23. Compliance with Restrictions	46
	3.24. FIRPTA	46
	3.25. Permits	46
	3.26. Existing Policies	46
	3.27. Shadow Anchors	46
9.	REPRESENTATIONS AND WARRANTIES OF BUYER	47
٠.	9.1. Organization; Authorization	47
	9.2. Bankruptcy of Buyer	47
	D.3. No Brokers or Finders	47
	9.4. Investment Intent	47
	The investment ment	1,
10.	REPRESENTATIONS AND WARRANTIES OF INVESTORS	48
	10.1. Power	48
	0.2. Authorization	48
	10.3. Validity	48
	10.4. Financial Ability	48

11.	CONDITIONS TO DEPOSIT  11.1. Conditions to Obligations of Sellers at Deposit Date  11.2. Conditions to Obligations of Buyer in Connection With Deposit	48 48 49
12.	DELIVERIES AT DEPOSIT DATE  12.1. Deliveries by the Selling Parties at Time of Deposit  12.2. Deliveries by Buyer at Time of Deposit	52 52 54
13.	CONDITIONS TO CLOSING 13.1. Conditions to Obligations of Sellers at the Closing 13.2. Conditions to Obligations of Buyer at the Closing	55 55 55
14.	CLOSING 14.1. Deliveries by the Selling Parties at Closing 14.2. Deliveries by the Buyer at Closing 14.3. Deliveries by Escrow Agent at Closing 14.4. Other Instruments	56 56 58 58 58
15.	BREACHES, REMEDIES AND INDEMNIFICATION 15.1. Buyer's Indemnity for Buyer's Inspection 15.2. Liabilities of Sellers and Remedies of Buyer for Deposit Date or Pre-Closing Breaches 15.3. Post-Closing Indemnity By Sellers 15.4. Post-Closing Indemnity By Buyer 15.5. Indemnification of Third-Party Claims 15.6. Payment 15.7. Limitations on Indemnification 15.8. Release of Buyer's Other Environmental Remedies 15.9. No Waiver	59 59 59 64 66 66 67 68 69 70
16.	TERMINATION; DEFAULT  16.1. Termination By Mutual Agreement  16.2. Termination before Deposit  16.3. Termination If Closing Fails to Occur  16.4. Termination Upon Default by Buyer  16.5. Termination Upon Default by Sellers  16.6. Exclusive Remedy Prior to Closing  16.7. Effect of Termination	70 70 70 70 70 71 72 72
17.	CONFIDENTIALITY; ANNOUNCEMENTS	73
18.	ASSIGNMENT	73
19.	RESOLUTION OF DISPUTES	74
20.	MISCELLANEOUS	76

76

76

20.1. Guarantee By Investors

20.2. Joint and Several Liability of the Sellers

20.3.	Survival	76
20.4.	Notices	77
20.5.	Governing Law	79
20.6.	Headings	79
20.7.	Business Days	79
20.8.	Counterpart Copies; Facsimile	79
20.9.	Binding Effect	79
20.10.	Interpretation	80
20.11.	Entire Agreement; Amendments; Waiver	80
20.12.	Severability	80
20.13.	Exhibits and Schedules	80
20.14.	Time	80
20.15.	Prevailing Party	81
20.16.	No Recording	81
20.17.	WAIVER OF TRIAL BY JURY	81
20.18.	Further Assurances	81
20.19.	No Offer	81
20.20.	OFAC Policy	81
20.21.	Limitation of Liability	82
20.22.	Macquarie CountryWide Management Limited	82
20.23.	Venue	83
20.24.	Control of Company	83

### List of Exhibits

Exhibit A Definitions and Meanings

Exhibit B Properties

Exhibit C Form of Tenant Estoppel Exhibit D Form of Seller Estoppel

Exhibit E Form of Sellers' Ground Lease Estoppel Certificate

Exhibit F Terms of Management Agreement and Accounting Services Agreement

#### List of Schedules

Schedule 2.2(a)(v) Construction Contracts Schedule 2.3(a) Wire Transfer Instructions

Schedule 2.10 Excluded Assets
Schedule 2.14 Option Parcels
Schedule 2.15 MPB&T Parcel

Schedule 3.1(m) Rent Guaranty and Other Escrows
Schedule 5.1(a)-1 Existing Policies and Existing Surveys
Schedule 5.1(a)-2 Exceptions Not Provided to Buyer

Schedule 5.1(a)-3 Extended Coverage

Schedule 5.1(b)(ii) List of Existing Rights of First Refusal or Options to Purchase Under Leases and Approved New Leases

Schedule 5.1(b)(v) Other Exceptions

Schedule 5.6(a)(i) Ongoing Remediation Projects

Schedule 5.6(a)(ii) Ongoing Remediation Escrow Allocated Shares Schedule 5.6(f) Form of Buyer's Environmental Insurance Policy

Schedule 7.3(a)(iii) Permitted Sales

Schedule 8.1(e) Jurisdictions Where Company Group is Licensed or Qualified to Do Business

Schedule 8.1(f) Capitalization of the Companies

Schedule 8.1(g) Subsidiaries

Schedule 8.1(h) Company Organizational Documents Schedule 8.3 Properties and Title to Properties

Schedule 8.5 No Violation
Schedule 8.6 Financial Statements
Schedule 8.8(a) Taxes; Tax Returns
Schedule 8.8(d) Partnership Status

Schedule 8.9 Litigation

Schedule 8.11 Ongoing Condemnation Proceedings

Schedule 8.14 Rent Rolls; Leases Schedule 8.15 Accounts Receivable Aging

Schedule 8.16 Service Contracts
Schedule 8.17 Material Contracts
Schedule 8.18 Existing Loans
Schedule 8.19 Ground Leases
Schedule 8.20 No Other Contracts

Schedule 8.22 Insurance

Schedule 8.24 Taxpayer Identification Numbers
Schedule 8.27 Shadow Anchors
Schedule 11.2(c) Underwriting Agreement
Schedule 13.2(a) Extended Lockout Properties

### PURCHASE AND SALE AGREEMENT

THIS PURCHASE AND SALE AGREEMENT dated February 14, 2005, by and among MACQUARIE COUNTRYWIDE-REGENCY II, LLC, a Delaware limited liability company ("Buyer"), MACQUARIE COUNTRYWIDE TRUST, an Australian listed registered managed investment scheme (ARSN 093 143 965) ("MCW"), REGENCY CENTERS CORPORATION, a Florida corporation ("Regency") (MCW and Regency each an "Investor" and collectively the "Investors"), USRP TEXAS GP, LLC, a Delaware limited liability company ("USRP"), U.S. RETAIL PARTNERS, LLC, a Delaware limited liability company ("Eastern," and together with USRP and U.S. Retail, the "Companies"), FIRST WASHINGTON INVESTMENT I, LLC, a Delaware limited liability company ("First Washington"), and CALIFORNIA PUBLIC EMPLOYEES' RETIREMENT SYSTEM, a unit of the State and Consumer Services Agency of the State of California (together with First Washington, each a "Seller," collectively, the "Sellers," and the Sellers and Companies are referred to collectively herein as "Selling Parties").

### **RECITALS**

A. The Companies directly or through their Subsidiaries (the Companies and their Subsidiaries are referred to collectively as the "Company Group" and individually as entities in the Company Group) own 101 retail centers, which are more specifically identified on Exhibit B (the "Properties"). Sellers own all of the issued and outstanding limited liability company interests (together with all rights as members, the "Membership Interests") and are the sole members of the Companies.

B. Buyer desires to purchase the Membership Interests from Sellers and Sellers desire to sell the Membership Interests to Buyer, upon the terms and conditions herein set forth. The Investors, directly or indirectly, own the Buyer.

NOW THEREFORE, in connection with the foregoing and the respective representations, warranties, covenants, agreements and conditions hereinafter set forth, and intending to be legally bound, the parties agree as follows (defined terms are set forth on <u>Exhibit A</u>):

# 1. PURCHASE AND SALE OF MEMBERSHIP INTERESTS

Subject to the terms and conditions of this Agreement, at the Closing, Sellers shall sell, assign, convey, transfer and deliver to Buyer, or Buyer's permitted assignee, and Buyer, or Buyer's permitted assignee, shall purchase, acquire and take assignment and delivery from Sellers, all of the Membership Interests, free and clear of all Liens.

# 2. PURCHASE PRICE – PAYMENT

**2.1.** Purchase Price. The purchase price payable for the Membership Interests shall be Two Billion, Seven Hundred Forty One Million, Two Hundred and Twenty-Five Thousand and No/100 Dollars (\$2,741,225,000.00) (the "Purchase Price"), subject to the Price Adjustments pursuant to Section 2.2.

#### 2.2. Price Adjustments.

- (a) The Purchase Price shall be adjusted at Closing as follows (each a "Price Adjustment"):
- (i) Buyer shall receive a credit against the Purchase Price in the aggregate amount of the outstanding principal balance of the Existing Loans as of the Closing Date.
- (ii) Buyer shall receive a credit against the Purchase Price in the amount of \$6,557,200 in connection with certain anticipated future real estate or franchise tax adjustments for those Properties located in California, Cook County, Illinois and certain other states.
- (iii) The Purchase Price shall be adjusted up or down in the net amount of the Proration Items calculated pursuant to Section 3.1(n)(ii) (Closing Statement).
- (iv) Buyer shall (A) receive a credit against the Purchase Price equal to the adjustment for Agreed Losses and Disputed Losses to which the Buyer is entitled pursuant to Sections 15.2(d) (Sellers' and Buyer's Pre-Deposit Options) and 15.2(e) (Post-Deposit/Pre-Closing Breach Notice) (taking into account the limitations thereto set forth in Section 15.2(d), except to the extent Sellers expressly waive such limitations pursuant to Section 15.2(d) and Section 15.2(e)), and (B) deposit into escrow the Disputed Losses pursuant to Section 15 (Breaches, Remedies and Indemnification) and the Disputed Amount Escrow pursuant to Section 2.5 (Escrows at Closing) and other applicable provisions of this Agreement.
- (v) Buyer shall receive a credit against the Purchase Price for any unpaid balance as of the Closing Date payable under the construction contracts set forth on Schedule 2.2(a)(y).
- (vi) The Purchase Price shall be increased by the amount which the Company Group actually paid with respect to capital projects undertaken at the specific request of Buyer pursuant to Section 7.2(i) (General Covenants), provided that if such amount exceeds \$8,000,000, then the Purchase Price shall not be increased by the amount in excess of \$8,000,000, unless such amount exceeds \$13,000,000, in which event the Purchase Price shall be increased by the sum of \$8,000,000 plus the amount of the excess over \$13,000,000.
  - (vii) Buyer shall receive a credit against the Purchase Price pursuant to Section 5.2 (Resolution of Other Exceptions), if applicable.
  - (viii) Buyer shall receive a credit against the Purchase Price pursuant to Section 11.2(d) (Tenant Estoppels), if applicable.
  - (ix) The Purchase Price shall be reduced to reflect the Agreed Value of any Dropped Assets pursuant to Section 2.13 (Dropping Properties).
- (x) The Purchase Price shall be increased by (A) one-half of the cost of the Buyer's Environmental Insurance Policy pursuant to Section 5.6(f) (Buyer's Environmental Insurance Policy) plus (B) all costs of obtaining any additional coverage requested by Buyer in excess of the coverage set forth on Schedule 5.6(f).

- (xi) Buyer shall receive a credit against the Purchase Price in the aggregate amount of all closing costs that Sellers are required to pay pursuant to Section 2.9 (Closing Costs), but have not otherwise paid and that have not otherwise been credited against the Purchase Price hereunder.
- (xii) The Purchase Price shall be increased by Three Million Seven Hundred and Fifty Thousand Dollars (\$3,750,000) to fund Buyer's 25% of the Environmental Escrow, subject to adjustment if the Initial Remediation Estimate is less than Fifteen Million Dollars (\$15,000,000).
- (xiii) The Purchase Price shall be adjusted, if appropriate, for the Edgewater Buy-Sell Amount pursuant to Section 2.11(b) (Buy-Sell with Crow Holdings).
- (xiv) The Purchase Price shall be adjusted, if appropriate, for the Agreed Value of the Option Parcels sold or under contract to be sold pursuant to Section 2.14 (Purchase of Option Parcels).
- (xv) The Purchase Price shall be adjusted for the sale of the MPB&T Parcel to the extent provided in Section 2.15 (Purchase of Maryland Permanent Bank Parcel).
- (b) Notwithstanding the foregoing, to the extent any of the Price Adjustments arise from or out of any facts, circumstances, conditions or events that overlap, there shall be no duplication of counting toward the calculation of any such Price Adjustments.
- (c) Price Adjustments (other than those set forth in Section 2.2(a)(iv) and (vi)) shall not be limited by, applied against or otherwise impact any limitations, deductibles, thresholds or maximum amounts relating to indemnification obligations and damage claims for Losses contained elsewhere in this Agreement.

### 2.3. Deposit.

(a) <u>Payment of Deposit</u>. Subject to the terms and conditions of this Agreement, on the Deposit Date, Buyer shall pay to Sellers the sum of Seven Hundred Twelve Million Four Hundred Thousand Dollars (\$712,400,000) as a deposit toward payment of the Purchase Price at Closing (the "**Deposit**"). Sellers agree to pay Buyer interest on the Deposit, at an interest rate equal to the Interest Rate (defined below), simple interest, accruing daily from the Deposit Date to and including the earlier of the Closing Date or the date on which the Deposit is returned to Buyer (or its designee), with interest due and payable to Buyer, by wire transfer, on the fifteenth (15<sup>th</sup>) day of each calendar month after the Deposit Date, in arrears. Notwithstanding anything herein to the contrary, Sellers shall be entitled to all interest or other income earned from Sellers' investment of the Deposit, subject however to Sellers' obligation to pay to Buyer the interest on the Deposit as set forth in this Agreement. At the Closing or other date on which the Deposit is returned to Buyer (or its designee) or otherwise released in accordance with the terms and conditions of this Agreement, the Sellers shall pay to Buyer all accrued but unpaid interest on the Deposit. All payments under this Agreement shall be

delivered by wire transfer of immediately available funds pursuant to instructions set forth in Schedule 2.3(a). "Interest Rate" shall mean (i) during the period commencing on the Deposit Date and ending on the ninetieth (90) day after the Deposit Date, an interest rate of six and one-tenth percent (6.1%) per annum, and (ii) thereafter, during each thirty (30) day period beginning on the day after either the end of the period in clause (i) or the previous such thirty (30) day period, an interest rate per annum equal to the rate for such period in clause (i) or previous such period, less one percent (1%), provided that an interest rate for any such 30-day period shall be no less than the yield on one month U.S. Treasury Bills reported in the most applicable Federal Reserve Statistical Release H.15(159) for the first Business Day of such 30-day period under the heading "Treasury Constant Maturities – Normal – 1-month," provided further, that the running of any such period shall be extended on a day-for-day basis for such time that the Closing has been delayed or adjourned by Sellers to address title issues pursuant to Section 5.4 (Postponement of Closing).

- (b) <u>Deposit Held in Blocked Account</u>. Sellers shall hold and invest the Deposit in a blocked account pursuant to a blocked account agreement among Buyer and Sellers (the "**Blocked Account Agreement**"), which shall be in form and content reasonably acceptable to the parties and which shall secure the Sellers' obligations to deliver the Membership Interests at Closing and to return the Deposit (or a portion thereof) pursuant to the terms and conditions of this Agreement. The Deposit shall be invested in cash and cash equivalents and other investments which are reasonably acceptable to the Buyer and Sellers.
- (c) <u>Escrow of Certificates on Deposit Date</u>. On the Deposit Date, simultaneously with payment of the Deposit, Sellers shall cause to be deposited into escrow, with an escrow agent reasonably acceptable to the parties (the "**Escrow Agent**"), certificates representing all the Membership Interests, together with the other documents set forth in Section 12.1(b). The Escrow Agent, Buyer and Sellers shall enter into an escrow agreement (the "**Escrow Agreement**") in form and content reasonably acceptable to the parties and which shall secure the Sellers' obligation to deliver the Membership Interests at Closing pursuant to the terms of this Agreement.
- **2.4.** <u>Closing</u>. The closing of the purchase and sale of the Membership Interests ("Closing") shall occur on the third (3rd) Business Day after satisfaction or written waiver by Buyer, of the conditions to Closing set forth in Section 13 (Conditions to Closing) (the "Closing Date"), at 10:00 a.m. Eastern time at the offices of Foley & Lardner LLP, Jacksonville, Florida, or on such sooner or adjourned date as this Agreement may contemplate or as the parties may agree in writing.

### 2.5. Escrows at Closing.

(a) <u>Disputed Price Adjustments and Prorations</u>. If the parties disagree about any Price Adjustment (including the amount of any prorations or Loss claimed by Buyer at Closing), the disputed amount shall be held back from the balance of the Purchase Price paid at Closing, and deposited into escrow (the "**Disputed Amount Escrow"**) with the Escrow Agent, pursuant to an escrow agreement delivered at Closing, in form reasonably acceptable to the parties, pending resolution in accordance with Baseball Arbitration.

- (b) <u>Ongoing Remediation Projects</u> At the Closing, \$4,000,000 of the Purchase Price less any amounts paid for Ongoing Remediation Projects between the date hereof and the Closing Date (other than amounts paid by Sellers pursuant to the Apex Contracts) shall be paid into escrow with Escrow Agent pursuant to an escrow agreement, delivered at Closing, in form reasonably acceptable to the parties (the "**Ongoing Remediation Escrow Agreement**") to establish a separate environmental remediation escrow fund (the "**Ongoing Remediation Escrow**") which shall be disbursed to Buyer and Sellers from time to time subject to and in accordance with Section 5.6(a) (Ongoing Remediation Projects).
- (c) <u>Future Remediation Escrow</u>. Prior to the Closing, Buyer shall notify Sellers of Buyer's estimate of the cost of Future Remediation Projects (the "**Initial Remediation Estimate**"). The Initial Remediation Estimate shall equal \$15,000,000, unless Buyer, in its sole discretion, estimates the cost of the Future Remediation Projects to be a lower amount, in which case the Initial Remediation Estimate shall be such lower amount. At the Closing, an amount equal to the Initial Remediation Estimate shall be paid into escrow with Escrow Agent pursuant to an escrow agreement, delivered at Closing, in form reasonably acceptable to the parties (the "**Future Remediation Escrow Agreement**") to establish a separate environmental remediation escrow fund (the "**Future Remediation Escrow**") which shall be disbursed to Buyer and Sellers from time to time subject to and in accordance with Section 5.6(b) (Future Remediation Projects).

### 2.6. Payments at Closing.

- (a) <u>Payment of Purchase Price</u>. At the Closing, Buyer shall pay to the Sellers an amount equal to the Purchase Price adjusted by the net application of all Price Adjustments. Sellers jointly and severally direct Buyer to pay all amounts due to either of the Sellers by wire transfer to the bank account and in accordance with the wire transfer instructions set forth on <u>Schedule 2.3(a)</u>.
- (b) <u>Application of Deposit</u>. At the Closing, Sellers shall apply the Deposit (but not any interest or other income thereon) to the payment of the Purchase Price.
  - 2.7. Payment Allocation. Sellers shall be responsible for allocating between themselves the amounts paid by Buyer to Sellers hereunder.
- 2.8. Tax Matters; Purchase Price Allocation. Buyer shall prepare and present to Sellers a proposed allocation of the Purchase Price (and all other capitalized costs) among the assets of the Company Group in accordance with Section 1060 of the Internal Revenue Code of 1986, as amended (the "Code") and Treasury regulations thereunder (and any similar provision of state, local or foreign law, as appropriate) within 60 days after the Closing Date. Sellers shall notify the Buyer within 15 days of the receipt of the proposed allocation of any objections that Sellers have with respect thereto and the changes that Sellers would make as a result thereof. If the Sellers have no objections, or if the Buyer agrees to the changes proposed by Sellers, such allocation (as so modified) shall be binding upon Sellers. If Buyer and Sellers cannot resolve any disagreements with respect to the proposed allocation within 15 days after Sellers deliver to Buyer any objections thereto, the Buyer and Sellers jointly shall select an independent valuation expert to resolve such differences, with the fees and costs of such valuation expert to be shared

equally between Buyer and Sellers, and with the decision of such valuation expert as to any matters in dispute between Buyer and Sellers to be binding and conclusive on both Buyer and Sellers. Sellers and Buyer and their Affiliates shall report, act, and file Tax Returns (including, but not limited to Internal Revenue Service Form 8594) in all respects and for all purposes consistent with such allocation prepared by Buyer (with such changes thereto as shall be proposed by Sellers and agreed to by Buyer or as shall be determined by the third party valuation expert as set forth above). The Sellers and the Company Group shall timely and properly prepare, execute, file, and deliver all such documents, forms, and other information as Buyer may reasonably request in preparing such allocation. Neither Buyer, nor the Company Group nor Sellers shall take any position (whether in audits, Tax Returns, or otherwise) that is inconsistent with such allocation, as finally determined as set forth above, unless required to do so by applicable law.

- 2.9. Closing Costs. The Sellers shall pay: (i) all transfer and similar taxes incident to the sale of the Membership Interests and change in control with respect to the Properties and the Company Group; (ii) an amount not to exceed \$9,000,000 in the aggregate for (A) the costs, fees and expenses required to be incurred in connection with obtaining the consent of the holders of the Existing Loans ("Lender Consents") to the Transactions and for the Existing Loans to remain outstanding after Closing (the "Consent Fees"), (B) the prepayment fees, prepayment penalties, breakage fees, defeasance costs and similar costs and penalties with respect to the prepayment and defeasance by Buyer of any or all of the Existing Loans incurred in connection with the Transactions or after the Closing and on or before October 1, 2005 (the "Prepayment Fees") and together with the Consent Fees, the "Loan Fees"), and (C) endorsements to Lender's existing title policies required for Lender's Consents; (iii) Lender's legal fees and expenses in obtaining the Lender Consents; (iv) all title insurance premiums and costs for Owner's Title Policies obtained in accordance with this Agreement for each Property; and (v) the costs required to be incurred to obtain those new surveys or updates to existing surveys as provided in Section 5.1 (Title and Survey Matters), but not the costs of any other additional endorsements to the Owner's Title Policies which were not issued in connection with the Existing Policies (which Buyer shall pay for). The Buyer shall pay (1) the Loan Fees in excess of \$9,000,000 and (2) all costs and expenses to obtain new (unless otherwise provided in Section 5.1) surveys and other endorsements to the Owner's Title Policies not required to be paid by Sellers and to obtain any additional lender's policies or endorsements thereto for new lenders or surveys for new lenders. Buyer shall pay all costs associated with Buyer's due diligence. Each party shall be responsible for its own attorney's fees. All other costs incurred at
- **2.10.** Excluded Assets. Prior to Closing, Sellers shall cause the Company Group, at Sellers' sole liability, cost and expense, to transfer the assets listed on Schedule 2.10 to Sellers or an entity owned by Sellers that is not an entity in the Company Group and to cause the Company Group to be released from all liabilities and obligations relating thereto or to cause the transferee to indemnify the Company Group therefrom.

# 2.11. Edgewater Retail Partners LLC.

(a) <u>Adjustments for One-Half Ownership of Edgewater</u>. Notwithstanding anything in this Agreement to the contrary, it is expressly agreed and understood that Edgewater

Retail Partners, LLC (together with the property owned by it, "Edgewater") is only 50% indirectly owned by the Sellers and therefore the following shall apply unless Edgewater is bought or sold prior to Closing pursuant to the buy-sell provisions referred to in Section 2.11(b):

- (i) Edgewater shall be considered part of the Company Group, subject to the limitation that Sellers, who only own 50% of the equity, are only entitled to 50% of income and cash flow from Edgewater;
- (ii) all covenants that relate to causing Edgewater to take or refrain from taking any action shall only require that Sellers and/or the Company Group use their commercially reasonable efforts to perform in accordance with their rights under the Organizational Documents of Edgewater;
  - (iii) the prorations regarding Edgewater shall be based on one-half ownership of the Edgewater Property;
- (iv) The credit against the Purchase Price in Section 2.2(a)(i) (Price Adjustments) for the Existing Loan encumbering Edgewater shall be based on 50% of the outstanding principal balance thereof as of the Closing Date; and
- (v) The adjustments described in this Section 2.11 are not intended to be duplicative of any prorations and adjustments provided for in Section 3.1 (Prorations) of this Agreement and there shall be no double counting of any such adjustments. In the event of a conflict between the provisions of this Section 2.11 and Section 3, the specific provisions of this Section 2.11 shall control.
- (b) <u>Buy-Sell with Crow Holdings</u>. The parties acknowledge that Sellers' one-half interest in Edgewater is subject to certain consent and buy-sell provisions with Crow Holdings, the other one-half owner of Edgewater, pursuant to the Organizational Documents of Edgewater. Sellers shall use commercially reasonable efforts to obtain consent from Crow Holdings for the change of control of Edgewater contemplated by this Agreement.

If Crow Holdings does not consent to the change of control, Sellers shall implement the buy-sell provision in the Edgewater Organizational Documents using a purchase price designated by Buyer. If the Company Group's interest in the Edgewater Property shall be sold to Crow Holdings prior to the Closing using the buy-sell price designated by Buyer, the Purchase Price shall be reduced by the amount of proceeds received from Crow Holdings. If Crow Holdings elects to sell pursuant to the buy-sell, and the Company Group purchases the interest of Crow Holdings prior to Closing, the Purchase Price shall be increased by the amount paid by Sellers to Crow Holdings for its one-half interest (the "Edgewater Buy-Sell Amount"). The Buyer, Sellers and the Company Group shall cooperate reasonably and in good faith in connection with the buy-sell process. The foregoing assumes Crow Holdings will agree to make the exercise of the buy-sell contingent upon the Closing. If Crow Holdings does not agree to make closing of the buy-sell contingent upon the Closing, the Buyer and Sellers will cooperate for a reasonable alternative structure to avoid the risk that Sellers would be forced to close the buy-sell transaction with Crow Holdings without assurance that the Closing would occur.

(c) PSE&G Generating Transmission Leads. Sellers, acting on behalf of Edgewater, have negotiated an agreement in principle with PSE&G Power In-City I, LLC ("PSE&G") for the installation of Generating Transmission Leads (the "PSE&G Facilities") under the Edgewater Property pursuant to a 99 year lease which, if implemented prior to Closing, would provide a one-time, lump-sum payment to the Company Group. Sellers have advised Buyer that the proposed PSE&G Facilities shall not adversely impact the operations and use of the Edgewater Property in any material respect and that the PSE&G Facilities shall be buried under the Edgewater Property and that PSE&G shall do all drilling off-site without interfering with the Edgewater Property in any material respect. Sellers and Buyer shall cooperate reasonably and in good faith in order to finalize and implement the proposed lease agreement with PSE&G, provided that the terms of the lease do not adversely affect the operation and use of the Edgewater Property in any material respect or result in a material reduction in the value of the Edgewater Property, and that Sellers shall share the Company Group's share of the lump sum payment, net of payment of expenses related to the transaction, when actually received from PSE&G, 90% to Sellers and 10% to Buyer. Buyer and Sellers shall act reasonably and in good faith in order to facilitate entering into a contract with PSE&G. This provision shall terminate if a definitive agreement is not signed with PSE&G within two years following Closing. If Buyer acquires Crow Holdings' interest in Edgewater, whether as part of the Closing or thereafter, the Buyer shall not share with Sellers any revenues from the lease agreement attributable to the 50% interest acquired from Crow Holdings. If Crow Holdings purchases the Company Group's interest in Edgewater, Buyer shall be released from all further obligations related to the PSE&G lease.

(d) <u>Title Insurance Matters</u>. Notwithstanding anything in this Agreement to the contrary, Sellers shall have no obligation to provide an Owner's Title Policy or Survey with respect to the Edgewater Property, there shall not be any Mandatory Removal Exceptions with respect to Edgewater and it shall not be a condition for the payment of the Deposit or the Closing that Sellers obtain an Owner's Title Policy for Edgewater. Instead, the corresponding condition and delivery for Edgewater shall be that Sellers obtain and deliver a title insurance commitment from the Title Company showing that Edgewater Retail Partners, LLC owns fee simple title to the Edgewater Property, subject to all matters of record and Permitted Exceptions.

**2.12.** Company Group's Pre-Closing Accounts Receivable and Payables. Following the Closing, Sellers shall provide funds to the Company Group for payment, when due, of all of the Company Group's liabilities for payment of money (other than Proration Items and Existing Loans) which would be required to be shown on a combined balance sheet of the Company Group prepared as of the Adjustment Time in accordance with GAAP. Following the Closing, the Company Group shall turn over to the Sellers all of the Company Group's accounts receivable other than Proration Items, as of the Adjustment Time, which are collected by the Buyer or the Company Group from any Person who owes such accounts receivable, provided that payments from such Persons shall be applied to satisfy current accounts to the extent due on or before the date on which such payments are received before being applied to pre-Adjustment Time accounts receivable.

### 2.13. **Dropping Properties**.

- (a) Dropping the Dropping Party Properties. If Sellers or Buyer (the "**Dropping Party**") shall have duly elected the "Section 2.13 Rights," the following shall apply: The Dropping Party shall have the right to exclude a Property or the Subsidiary that owns such Property and only such Property ("**Dropped Asset**") from the Closing by delivering a written notice to the Buyer electing to exercise the Section 2.13 Rights on or before the Closing Date (a "**Dropped Property Notice**"). Buyer shall have five (5) Business Days prior to the Closing Date to waive the problem or condition that gave rise to the specific exercise of Section 2.13 Rights. If Buyer so elects to waive, the Property shall not be deemed a Dropped Asset and Buyer shall be deemed to have waived any right to a Price Adjustment or indemnification arising from the problem or condition that gave rise to such exercise of the Section 2.13 Rights. If Buyer does not elect to waive the condition or problem within said period, this Agreement will be deemed amended, without any further action on the part of any party, with respect to each Property or the Subsidiary that owns the Dropped Asset, as follows:
- (i) All Losses (including, Agreed Losses and Disputed Losses) associated with such Dropped Asset shall be excluded from the calculation of Agreed Losses and Disputed Losses, except to the extent the Losses continue as an Entity Claim.
- (ii) All Breach Events associated with such Dropped Asset shall be excluded from Breach Events, except to the extent any Breach Event was an Entity Claim or resulted in Claims for breaches of Entity Representations.
  - (iii) All Other Exceptions associated with such Dropped Asset shall be excluded from Other Exceptions.
- (iv) All Future Remedial Actions Expenses and all expenses arising from Recommended Future Remedial Actions associated with such Dropped Asset shall be excluded from Future Remedial Actions.
  - (v) All Existing Loans associated with such Dropped Asset shall be excluded from Existing Loans.
- (vi) There shall be no Claims with respect to such Dropped Asset, except to the extent it continues to be an Entity Claim or Claims for breaches of Entity Representations.
  - (vii) At the Closing, the Purchase Price shall be reduced by the amount of the Agreed Value for such Dropped Asset.
  - (viii) Sellers shall cause such Dropped Asset to be transferred to a Person that is not part of the Company Group prior to Closing.
  - (ix) Sellers shall be responsible for the costs of transferring Dropped Assets.

- (b) <u>Limit on Sellers' Dropped Assets</u>. Notwithstanding anything contained in this Agreement to the contrary, Properties cannot be excluded pursuant to this Section 2.13 if the total Agreed Value of such Dropped Assets (other than (i) exclusions at Buyer's request relating to Tenant Estoppels pursuant to Section 11.2(d), (ii) exclusion of the Properties listed on <u>Schedule 13.2(a)</u> pursuant to the requirements of Section 13.2(a)), (iii) the MPB&T Parcel, and (iv) if Edgewater is sold pursuant to the buy-sell (described in Section 2.11)), exceeds \$270,000,000.
- (c) <u>Postponement of Closing</u>. Sellers shall have the right, in their sole and absolute discretion, to adjourn the Closing Date, if necessary, from time to time by notice to Buyer delivered on or prior to the date then set for Closing (for a maximum adjournment period not to exceed fifteen (15) Business Days in the aggregate) to obtain any necessary consents or take other actions necessary or appropriate to effect an exclusion of Properties pursuant to this Section 2.13.
- 2.14. Purchase of Option Parcels. The parties acknowledge and agree that certain Leases may contain, or the Transactions may trigger, certain rights in favor of third Persons (all such third Persons, "Option Holders") to purchase or otherwise acquire certain of the Properties or portions of the Properties (collectively, "Option Parcels"), each as set forth on Schedule 2.14. The Sellers shall use commercially reasonable efforts to obtain waivers from the Option Holders of their purchase rights prior to Closing. If any of the Option Holders refuse to waive the potential option rights and elect to purchase or otherwise acquire an Option Parcel and the closing of such transaction occurs prior to the Closing Date, then Buyer shall be entitled to a Price Adjustment equal to the Agreed Value of the Option Parcel as set forth on Schedule 2.14 (less the principal amount of any related Existing Loans which will be repaid or assumed at the time of sale). If an Option Holder elects to purchase or otherwise acquire an Option Parcel but the transaction does not close prior to the Closing Date, then the Agreed Value of the Option Parcel, as set forth in Schedule 2.14 (less the principal amount of any related Existing Loans which will be repaid or assumed at the time of sale), shall be paid into escrow at Closing. If the sale of the Option Parcel actually closes following the Closing Date, the escrowed funds shall be paid to Buyer by the Escrow Agent and the net cash proceeds of the sale (after payment of related expenses and repayment or assumption of any related Existing Loans) shall be paid to the Sellers. If the option rights have terminated with regard to any election and the sale does not close with respect to any exercise of option rights, the escrowed funds shall be paid to Sellers by the Escrow Agent. Notwithstanding anything in this Agreement to the contrary, the rights held by the Option Holders shall be Permitted Exceptions as to rights to purchase or otherwise acquire the Option Parcel arising directly or indirectly from the Transactions, but not as to any exercise right which the Option Holder could have exercised prior to, and independent of the Transactions (unless such right qualifies as a Permitted Exception under Section 5.1(b)), and Sellers and the Company Group shall be entitled to comply with obligations with respect thereto including consummating any sale of an Option Parcel. Sale of an Option Parcel pursuant to this provision shall be treated as a Dropped Asset pursuant to Section 2.13 hereof.
- **2.15.** Purchase of Maryland Permanent Bank Parcel. The Maryland Permanent Bank & Trust has exercised its option to purchase the parcel (the "MPB&T Parcel"), more specifically described on Schedule 2.15, located at the Valley Center Property. If the Closing of the MPB&T Parcel occurs prior to the Closing Date, then Buyer shall be entitled to a Price

Adjustment equal to the Agreed Value of the MPB&T Parcel as set forth on Schedule 2.14 (less the principal amount of any related Existing Loans which will be repaid or assumed at the time of sale). If the sale of the MPB&T Parcel has not closed prior to the Closing Date, then the Agreed Value (less the principal amount of any related Existing Loans which will be repaid or assumed at the time of sale) of the MPB&T Parcel shall be paid into Escrow at Closing. If the sale of the MPB&T Parcel actually closes within 24 months after the Closing Date, the escrowed funds shall be paid to Buyer by the Escrow Agent as a Price Adjustment and the net cash proceeds of the sale (after payment of related expenses and the principal amount of any related Existing Loans) shall be paid to Sellers. If the sale does not close within 24 months after the Closing Date, the escrowed funds shall be paid to Sellers, by Escrow Agent, provided, however, that if such sale closes at a later date, Sellers shall promptly pay to Buyer an amount equal to the excess of the Agreed Value over the option price. Sale of the MPB&T Parcel shall be treated as a Dropped Asset pursuant to Section 2.13 hereof.

# 3. PRORATIONS

**3.1.** Prorations. All matters involving prorations or adjustments to be made in connection with Closing and not specifically provided for in some other provision of this Agreement shall be adjusted in accordance with this Section 3.1. All prorations, adjustments and closing payments shall be made on the basis of closing statement(s) reasonably approved by Buyer and Sellers at the Closing Date (or, if the parties fail to agree, as determined by Baseball Arbitration). Except as otherwise set forth herein, all items to be prorated and other adjustments to be made pursuant to this Section 3.1 (the "**Proration Items**") shall be prorated as of 11:59 p.m. of the day immediately preceding the Closing Date (the "**Adjustment Time**") and the net amount thereof either shall be paid by Buyer to Sellers or credited to Buyer, as the case may be, at the Closing. Buyer shall be treated as the owner of the Company Group and the Properties, for purposes of prorations of income and expenses, on and after the Adjustment Time. Buyer and Sellers acknowledge that, except as otherwise expressly provided herein, and elsewhere in the Transaction Documents, the purpose and intent as to proration and adjustments is that the Sellers shall bear all expenses of ownership and operation of the Company Group and the Properties and shall receive the benefit of all income therefrom accruing until Closing and Buyer shall bear all such expenses and receive the benefit of all income accruing after the Adjustment Time.

# (a) Taxes; Water; Sewer and other Utilities.

- (i) All real estate, personal property and franchise taxes and special assessments attributable to the Company Group and the Properties shall be prorated on an accrual basis as of the Adjustment Time. Sellers shall pay or cause the Company Group to pay, prior to Closing, all taxes to the extent levied or billed prior to the Adjustment Time.
- (ii) Water rates and charges, except to the extent required to be paid by Tenants pursuant to Leases in effect as of the Adjustment Time directly to the entity imposing same, shall be prorated as of the Adjustment Time.
- (iii) Sewer taxes and rents, except to the extent required to be paid by Tenants pursuant to Leases in effect as of the Adjustment Time directly to the entity imposing same, shall be prorated as of the Adjustment Time.

- (iv) Utility charges, including, but not limited to, electricity, gas, steam, telephone, internet, cable or satellite television and any other utilities, except to the extent required to be paid by Tenants pursuant to Leases in effect as of the Adjustment Time directly to the entity imposing same, shall be prorated based upon the most current bill charged to Seller, plus sales taxes thereon, and shall be reprorated after the Closing Date when bills for the period in which Closing occurs are received, unless actual readings are obtained as of the Adjustment Time, in which case such actual readings shall govern, and each party shall pay the amount properly allocated or billed to it, respectively.
- (v) Apportionment of taxes, water rates and charges, sewer taxes and rents shall be made on the basis of the fiscal year for which assessed assuming the maximum available discount. If the taxes, water rates and charges, sewer taxes and rents for any Property have not been set for the year in which the Closing occurs, then the proration of such item for such Property shall be made on the basis of the real property tax rate, water rates and charges, sewer taxes and rents or vault charges, as applicable, for the preceding year applied to the latest assessed valuation and such proration shall be adjusted in cash between Sellers and Buyer upon presentation of written evidence that the actual taxes, water charges and sewer rents paid for such Property for the year in which the Closing occurs differ from the amounts used at the Closing in accordance with the provisions of Section 3.1(n)(iii) (Preparation of Final Statement) hereof.
  - (vi) All increases in taxes imposed due to a change of use of any portion of any Property after the Closing Date shall be paid by the Buyer.
- (vii) Any increases in taxes directly attributable to reassessment of the Properties located in California and Cook County, Illinois for ad valorem (including real estate and franchise) tax purposes as a result of the Transactions shall be paid by Buyer.
- (viii) If any refund of any prorated item is made after the Closing Date whether for a period prior to or on and after the Closing Date, the same shall be applied first to the reasonable out-of-pocket third party costs incurred in obtaining same and the balance, if any, of such refund shall, to the extent received by Buyer, be paid to Sellers (to the extent relating to the period through the Adjustment Time), and to the extent received by Sellers, be paid to Buyer (to the extent relating to the period commencing with the Closing Date).
- (ix) Sellers shall have the right to continue the prosecution of any real estate tax certiorari or other proceedings or protests brought by Sellers, at Sellers' expense, to reduce the taxes or any portion thereof for the fiscal year prior to the fiscal year in which Closing occurs which are pending as of the Closing Date using counsel selected by Sellers, until a final determination has been rendered or a settlement reached. If Sellers elects not to pursue such matters, Buyer may continue to prosecute same. Sellers (or Buyer, if applicable) shall pay from the proceeds of any refund, all actual and reasonable legal, accounting and other expenses which may be incurred in connection with such real estate tax certiorari or other proceedings or protests, and Sellers (or Buyer, if applicable, with Sellers' reasonable consent) may settle or compromise any such proceedings. If such determination shall result in a refund or credit of taxes, or if the taxes are otherwise reduced, because of a reduction of the assessed valuation or tax rate or for any other reason, then the net amount of such refund or credit of taxes (after deducting Sellers' (and Buyer's, if applicable) out-of-pocket third party costs and any refunds or

credits of taxes owed to Tenants under Leases or expired or terminated Leases) shall be apportioned between Sellers and Buyer as of the Adjustment Time (and the reimbursement made to Sellers (and Buyer, if applicable) for their out of pocket costs to the extent of such deduction). Any amount of the refund due to Buyer for the period commencing with the Closing Date pursuant to this Section 3.1(a)(ix) shall be paid to Buyer by Sellers promptly following receipt thereof by Sellers and Buyer shall also be entitled to any amounts of the refund of taxes owed under Leases to Tenants then in occupancy, provided that Buyer shall agree to give each such Tenant a credit for its applicable portion of the amount of such refund of taxes. Seller agrees to give all applicable tenants a refund for its applicable portion of the amount of any refund of taxes received by Sellers. Any amount of the tax refund due to Sellers for the period prior to the Closing Date pursuant to this Section 3.1(a)(ix) received by Buyer after the Closing shall be paid by Buyer to Sellers promptly upon receipt of any refund on the final determination of the credit or of the reduction.

(b) <u>Fuel</u>. Proration shall be made of fuel on any buildings located on the Properties on the Adjustment Time, based upon a reading made by the supplier as close as obtainable to the Adjustment Time (reasonably adjusted to the quantity present on the Adjustment Time). The value thereof shall be calculated at Sellers' last cost (including sales tax), based upon actual invoices. If the heating, ventilation or air conditioning for any buildings located on the Properties is provided by a measurable product (<u>e.g.</u>, steam or gas) the adjustment will be based on meter readings prorated, if necessary, to the Adjustment Time.

### (c) Rents.

- (i) Ground rent and all other amounts payable under the Ground Leases shall be prorated as of the Adjustment Time. Sellers will be credited with an amount equal to all security deposits, prepaid rentals and other deposits paid or deposited under any Ground Lease, together with any interest that has accrued thereon to the extent said deposits or prepaids have not been applied to outstanding obligations in accordance with the applicable Ground Lease.
- (ii) Base or fixed rent ("**Fixed Rent**"), including any pre-paid rent, collected by Sellers or the Company Group prior to Closing shall be prorated as of the Adjustment Time. Any amounts collected by the Company Group prior to the Adjustment Time with respect to specific tenant billings for pass-through payments, charges for overtime services, work orders, other specific services or special items performed or provided at the request of a given Tenant, and all other charges payable by Tenants properly allocable to the period prior to the Adjustment Time and not subject to readjustment, shall be retained by the Company Group. All such amounts collected by Sellers or the Company Group properly allocable to the period from and after the Adjustment Time, shall be paid to and be the property of Buyer. All amounts collected by the Buyer or Seller that are subject to readjustment with respect to Tenants shall be prorated as of the Adjustment Time (and re-prorated if not finally determinable at Closing).
- (iii) Common area maintenance or "CAM" charges and "escalation rent" or additional rent based upon increases in taxes, operating expenses, labor costs, a cost of living index or porter's wages or otherwise, and all other charges required to be paid periodically by Tenants in estimated amounts and later recalculated based upon actual expenses, taxes and/or

other relevant factors, other than Percentage Rent, as hereinafter defined (all of the foregoing, collectively the "Additional Rent"), shall be prorated between Sellers and Buyer as of the Adjustment Time based on such estimated payments paid by the Tenants for the applicable accounting periods (i.e., monthly, quarterly or annually) in which the Closing occurs, as set forth in the applicable Leases (each, with respect to any period for which Rent is due under the applicable lease, an "Applicable Rent Period"), with (a) Sellers receiving the portion of such Additional Rent that the portion of such Applicable Rent Period occurring prior to the Adjustment Time bears to such entire Applicable Rent Period. If, at any time, Sellers shall receive any installments of Additional Rent attributable to Applicable Rent Periods from and after the Adjustment Time, such sum shall be credited to Buyer at the Closing or if received after Closing, immediately paid to Buyer. At the time(s) of final calculation and collection from (or refund to) each Tenant of the actual Additional Rent for an Applicable Rent Period with respect to which estimated amounts paid by such Tenant have been prorated between Sellers and Buyer, there shall be a re-proration between Sellers and Buyer.

(iv) Payments of additional rent based upon the amount by which a designated percentage of each Tenant's gross revenues achieved during the Applicable Rent Period exceeded base rent, breakpoint or some other standard ("Percentage Rent"), if any, for any Applicable Rent Period in which the Closing occurs shall be apportioned between the Sellers and Buyer. Sellers and Buyer will prorate the total Percentage Rent due from each Tenant for such Tenant's Applicable Rent Period, based upon the number of days in the Applicable Rent Period each of the parties owned the Company Group, as soon as practicable after each such Tenant has made the payment due with respect to such Applicable Rent Period. Until the Percentage Rent payable by a Tenant is prorated pursuant to the immediately preceding sentence, (A) Sellers will initially retain all monthly or quarterly interim payments of Percentage Rent made before the Closing by such Tenant for such Applicable Rent Period and (B) Buyer will initially retain all monthly or quarterly interim payments of Percentage Rent made on and after the Closing Date by such Tenant for its Applicable Rent Period. Any amounts collected by or on behalf of Buyer from a Tenant after the Closing that relate to Percentage Rents payable with respect to any Applicable Rent Period ending before the Closing or any Lease that terminated before the Closing Date will be remitted to Sellers promptly upon receipt, and any amounts collected by or on behalf of Sellers from a Tenant after Closing Date will be remitted to Buyer promptly upon receipt. In the event that any refund shall be due to any Tenants, there shall be a reproration as of the Adjustment Time between Sellers and Buyer.

(d) <u>Delinquent Rent</u>. Any Fixed Rent, Additional Rent or Percentage Rent (collectively, "**Rent**") not paid when due is "**Delinquent Rent**". During the period after Closing, any Delinquent Rent collected by the Sellers, Buyer or the Company Group from any Tenant who owes Rent for the Applicable Rent Period in which Closing occurs or for Applicable Rent Periods prior to the Closing shall be applied *first*, in payment of Rent owed by such Tenant for the Applicable Rent Period (not to exceed one month) in which the Closing occurs; *second*, in payment of Rent owed by such Tenant for all Applicable Rent Periods after the Applicable Rent Period in which the Closing occurs; and *third*, after Rent for all current Applicable Rent

Periods have been paid in full, in payment of Rent owed by such Tenant for the Applicable Rent Periods prior to the Applicable Rent Period in which the Closing occurs. Each such amount, less any reasonable out-of-pocket third party costs of collection (including reasonable attorneys' fees and expenses) reasonably allocable thereto, shall be adjusted and prorated as provided above, and the party who receives any such amount shall promptly pay over to the other party any portion thereof to which it is so entitled. Buyer agrees that it shall cause the Company Group to use commercially reasonable efforts to collect all Additional Rent and Percentage Rent payable by Tenants after Closing and any Delinquent Rent due the Company Group or Selling Parties (provided, however, that neither Buyer nor Company Group shall have any obligation to institute legal proceedings, including an action for unlawful detainer, against any Tenant owing Delinquent Rents). Notwithstanding Buyer's good faith compliance with the provisions of the immediately preceding sentence, if the Buyer or Company Group does not collect past due Rent, Sellers shall have the right, during the six (6) month period after Closing, upon prior written notice to Buyer, to commence proceedings to collect pre-closing Delinquent Rent, provided that (i) Sellers shall not be entitled to pursue any Lease termination, dispossession, eviction or similar proceedings against any such Tenant and (ii) the sole remedy which may be sought by Sellers against any such Tenant is money damages.

- (e) <u>Prepaid Items</u>. Any prepaid costs, expenses, charges, fees and any other items paid by Sellers with respect to any Property, including, fees for licenses and annual permit and inspection fees, along with any deposits thereunder, that inure to the benefit of Buyer shall be apportioned between the Sellers and Buyer as of the Adjustment Time.
- (f) <u>Declaration Assessments</u>. Any assessments and other charges paid by Company Group under any declarations, reciprocal easement agreements, covenants, restrictions or other agreements affecting the Properties shall be prorated between the Sellers and Buyer as of the Adjustment Time.
- (g) <u>Service and Material Contracts</u>. Fees and other charges under any Service Contracts or Material Contracts that are not required to be terminated by Sellers pursuant to this Agreement at or prior to Closing shall be prorated between the Sellers and Buyer as of the Adjustment Time. Seller shall be liable (for periods both before and after Closing) with respect to any Service Contracts that are required to be terminated by Sellers.
- (h) <u>Inventory</u>. The value of any building inventory and supplies in customary full case quantities (e.g., soap, cleaning powder, light bulbs, etc.) in unopened containers in the buildings at the Properties, if any, in accordance with an inventory prepared by Sellers (subject to review by Buyer), shall be credited to Sellers. Such value amount shall be determined based upon Sellers cost thereof, based upon actual invoices.
- (i) <u>Cash on Hand</u>. The parties anticipate that the Company Group shall distribute all cash on hand to Sellers prior to Closing, to the extent that such distributions are permitted by law and applicable contractual restrictions and except for deposit or reserve accounts addressed elsewhere in this Section 3.1. However, to the extent cash is not distributed prior to Closing, Sellers shall be entitled to a credit at Closing for any cash held by Company Group and not distributed to Sellers prior to Closing.

- (j) <u>Leasing Commissions and Leasing Costs</u>. The Sellers shall be responsible for all unpaid leasing and brokerage commissions and tenant improvement costs with respect to the current term of all Leases executed prior to the date hereof. To the extent that any such commissions or tenant improvement costs shall not have been paid by the Sellers prior to Closing (the "**Unfunded Leasing Expenses**"), Buyer shall be entitled to a credit against the Purchase Price at Closing in an amount equal to the Unfunded Leasing Expenses, and Buyer shall (to the extent of such credit) be responsible for payment of the Unfunded Leasing Expenses from and after the Closing. With respect to any other Lease or Lease modification entered into by the Sellers after the date hereof in accordance with the terms of Section 7.3(d) (Tenant Leases), and with respect to any renewal, expansion or extension of any Lease through the exercise of an existing option, occurring after the date hereof, all tenant improvement work, leasing commissions, legal fees or other expenses or grants of any free rent period or other concessions shall be prorated over the term of the applicable Lease, renewal or extension based on the economic benefit to the parties hereto occurring before or after the Closing.
- (k) <u>Tenant Security Deposits</u>. The Sellers shall grant to Buyer a credit against the Purchase Price in an amount equal to the aggregate of the unapplied Tenant security and other deposits under the Leases held by the Company Group in cash at the time of the Adjustment Time and not retained by the Company Group after Closing, including all accrued interest thereon to the extent any Tenants may be entitled to receive such amounts in connection with the refund of any such deposit.
- (l) Existing Loans. Interest, both payable and accrued but not yet payable, fees, costs and expenses (other than Loan Fees) under the Existing Loans shall be prorated as of the Adjustment Time. In addition, Sellers shall receive a credit on a dollar-for-dollar basis an amount equal to any reserves, escrows or similar funds held by the Lenders in connection with the Retained Loans.
- (m) Rent Guaranty Escrows. Instead of Sellers receiving a credit therefor, Buyer shall administer the rent guaranty escrows or similar funds described on Schedule 3.1(m) held by any entity in the Company Group or any escrow agent as security for rent or similar guaranties with respect to properties previously sold by such entity as required in said rent guaranty escrows or other obligations, and Buyer shall cause any amounts released from any such escrows to any entity in the Company Group to be promptly paid to Sellers.

### (n) Calculations.

(i) <u>Applicable Time Period</u>. For purposes of calculating Proration Items under this Agreement, Buyer shall be deemed to be in ownership of the Company Group and the Properties, and, therefore entitled to the income therefrom and responsible for the expenses thereof, from and after the Adjustment Time, including the entire day upon which the Closing occurs. Notwithstanding anything herein to the contrary, all such prorations shall be made on the basis of the actual number of days of the applicable time period which shall have elapsed prior to the Adjustment Time and based upon the actual number of days in the applicable time period and a three hundred sixty five (365) day year. All prorations and adjustments made pursuant to this Section 3 shall be made without duplication whatsoever.

(ii) <u>Closing Statement</u>. Sellers will be charged and credited for the amounts of all of the Proration Items relating to the period up to and including the Adjustment Time, and Buyer will be charged and credited for all of the Proration Items relating to the period after the Adjustment Time. Sellers will prepare in good faith and deliver a statement of estimated Proration Items and other credits and adjustments to the Purchase Price hereunder on a Property by Property basis, together with all relevant supporting documentation, to be submitted to Buyer no less than ten (10) Business Days before the Closing Date (the "Closing Statement"). Upon approval by Sellers and Buyer, the preliminary Proration Items and other credits and adjustments reflected in the Closing Statement will be paid at Closing by Buyer to Sellers (if the preliminary Proration Items, credits and adjustments result in a net credit to Sellers) or by Sellers to Buyer (if the preliminary Proration Items, credits and adjustments result in a net credit to Buyer) by increasing or reducing the cash to be delivered by Buyer in payment of the Purchase Price at the Closing. Notwithstanding anything herein to the contrary, in the event the actual amounts of the Proration Items, credits and adjustments are not known as of the Adjustment Time, the proration of the Proration Items, credits and adjustments will be made at Closing on the basis of the best evidence then available; thereafter, when actual amounts are determined, re-prorations will be made as provided in Section 3.1 below on the basis of the actual amounts, and a final cash settlement will be made between Sellers and Buyer. Buyer will be entitled to all deposits presently in effect with the utility providers, and Sellers shall receive a credit for same. Sellers and Buyer shall cooperate to produce the Final Closing Adjustment in accordance with Section 3.1.

(iii) Preparation of Final Statement. On or before thirty (30) days following the last day of the 2nd full calendar month following the year end reconciliations that are required to be performed under all of the Leases existing on the date hereof or such later date as is contemplated herein for various reprorations, including real property taxes, or such later date as is mutually agreed by the Sellers and the Buyer, the Buyer will prepare and deliver to the Sellers a final unaudited statement of Proration Items and other credits and adjustments to the Purchase Price as of the Adjustment Time (the "Final Statement"), based on information available as of the last day of the 2nd full calendar month following the year end reconciliations are required to be performed under all of the Leases existing on the date hereof or such later date as is mutually agreed by the Sellers and the Buyer. Sellers will afford Buyer and its representatives and auditors the opportunity at reasonable times and upon reasonable prior notice to review with Buyer all underlying financial records and work papers pertaining to the preparation of the Final Statement within Sellers' control. Subject to Section 3.1(n)(iv) below, any net adjustment in favor of Buyer will be paid in cash by Sellers to Buyer no later than ten (10) days after delivery of the Final Statement. Subject to Section 3.1(n)(iv) below, any net adjustment in favor of Sellers will be paid in cash by Buyer to Sellers no later than ten (10) days after delivery of the Final Statement. The Final Statement may be audited at either party's request. The payments made under the Final Statement will be the "Final Closing Adjustment."

(iv) <u>Audit</u>. If Buyer and Sellers do not agree on the computation of the Proration Items and other credits and adjustments to the Purchase Price in the Final Statement, then each party will prepare its own audited calculation of the Final Statement and the differences between such audited calculations shall be resolved in accordance with Section 19 (Resolution of Disputes) Buyer and Sellers will cooperate and assist each other in their

calculation of the Final Statement and in the conduct of the audits and reviews referred to in this Section 3.1(n), including making available to Buyer and Sellers to the extent necessary of books, records, work papers and personnel. If the Baseball Arbitration reflects a net adjustment in favor of Sellers, Buyer will pay to Sellers the amount of that adjustment in cash to Sellers no later than ten (10) days after the determination of the Final Statement. If the Baseball Arbitration reflects a net adjustment in favor of Buyer, Sellers will pay to Buyer the amount of that adjustment in cash to Buyer no later than ten (10) days after the determination of the Final Statement.

(v) Prorations shall not affect, be limited by, or applied against any limitations, deductibles, thresholds or maximum amounts relating to indemnification obligations and claims for Losses contained elsewhere in this Agreement.

### 4. BUYER'S RIGHT OF INSPECTION; INSPECTION PERIOD

- **4.1.** Right to Evaluate. Commencing on the date hereof and continuing until Closing (the "Inspection Period"), Buyer and its agents shall have the right during reasonable hours (with reasonable advance notice to the applicable Selling Parties and subject to the rights of the Tenants and other occupants in possession), at Buyer's sole cost and expense and at Buyer's and its agents' sole risk, to perform inspections and tests of the Properties, and to perform such other analyses, inquiries and investigations as Buyer shall reasonably deem necessary or appropriate, including, a complete physical inspection of the Properties, soil tests or sampling and environmental audits, subject to the limitations on environmental testing set forth in Section 5.6 (collectively, "Physical Testing"), All Physical Testing shall be done in accordance with all Laws, subject to the rights of Tenants under the Leases.
- (a) In no event shall any such inspection or Physical Testing (i) permanently damage any part of the Properties or any personal property owned or held by any Tenant, (ii) in any material way disrupt, disturb or interfere with the on-going operations of the Properties, the providing of any maintenance or services to the Properties or the rights of Tenants or other occupants at the Properties or interfere with their use of the Properties pursuant to their respective Leases. Buyer shall notify the applicable Selling Parties in advance and coordinate the timing of any site inspections with the applicable Selling Parties and the property manager of the applicable Property so as to reasonably minimize disruption of the operation of the Properties. Such notice shall describe the general scope of the due diligence Buyer intends to conduct during Buyer's access to the Properties. The applicable Selling Parties will make commercially reasonable efforts to have an agent available to accompany Buyer or any of Buyer's agents, and in all events the applicable Selling Parties shall have the right to have a representative present during any visits to or inspections of any Properties.

After making such tests and inspections, Buyer agrees to promptly, and in no event later than ten (10) days after such damage occurs, restore the affected Improvements and the surface of any affected Property to the same condition that existed immediately prior to such tests and inspections.

(b) Buyer may, after notice to Sellers, communicate or conduct interviews with any employee, lender, partner or joint venturer of any Selling Party, Company Group entity or any Tenant of any of the Properties without the prior consent of the applicable Selling Party.

Such interviews shall not unreasonably disrupt or disturb (i) the on-going operation of the Properties or the Selling Parties, (ii) any services to the Properties or (iii) the quiet possession of any Tenants under the Leases. The applicable Selling Party will have the right to have its representative present at all times during any interviews with any employee, lender, partner or joint venturer or any Tenant of any of the Properties.

- **4.2.** <u>Inspection Obligations</u>. In addition to their obligations under Section 4.1, Buyer and its agents and representatives shall: (i) not injure or otherwise cause bodily harm to any of the Selling Parties and the Company Group, their respective agents, contractors and employees or any Tenant or other occupant of the Properties; (ii) promptly pay when due the costs of all tests, investigations and examinations done with regard to the Properties; (iii) not permit any mechanics', materialmen's or other liens to attach to any Property by reason of acts or omissions in connection with the exercise of Buyer's rights under this Section 4; and (iv) at its sole cost and expense, comply with all applicable Laws in conducting its inspection of the Properties and the Physical Testing.
- **4.3.** Sellers' Deliveries. During the term of this Agreement, Sellers shall use commercially reasonable efforts to make available to Buyer all documents and information regarding the ownership, construction, or operation of the Properties and other information regarding the Company Group which are in the possession of the Sellers, the Company Group or their agents as Buyer shall reasonably request. Except as otherwise expressly set forth herein or in Selling Parties' deliveries pursuant to Section 12.1 (Deliveries by the Selling Parties at Time of Deposit), the Selling Parties make no representations or warranties of any kind regarding the accuracy, thoroughness or completeness of or conclusions drawn in the information contained in such Documents, if any, relating to the Properties.

### 4.4. No Other Representations and Warranties Outside Agreement and Transaction Documents.

- (a) The parties hereby expressly acknowledge and agree that, except as set forth in this Agreement and the Transaction Documents, as reliance thereon and enforcement thereof may be limited in this Agreement and the Transaction Documents, no party, nor anyone acting for or on behalf of any party, has made any oral or written representation, warranty, covenant, agreement, promise or statement, express or implied, to the other party, or to anyone acting for or on behalf of the other party, and no party has, except as provided in this Agreement and the Transaction Documents, relied on, and shall not be entitled to rely on same.
- (b) Without limiting the generality of the foregoing, Buyer hereby acknowledges and agrees that, except for the Selling Parties' representations and warranties set forth in this Agreement and the Transaction Documents, and without limiting or restricting any representation, warranty, affirmation, statement or any other provisions set forth in this Agreement or the other Transaction Documents, as reliance thereon and enforcement thereof may be limited in this Agreement, Buyer waives, and the Selling Parties' disclaim, all warranties of any type or kind whatsoever with respect to the Properties, whether express or implied, including, by way of description but not limitation, those of fitness for a particular purpose and use.

(c) WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, BUYER SPECIFICALLY ACKNOWLEDGES AND AGREES THAT (i) EXCEPT AS SET FORTH HEREIN OR IN ANY OTHER DOCUMENT DELIVERED PURSUANT HERETO, THE PROPERTIES ARE "AS IS, WHERE IS AND WITH ALL FAULTS" AND (ii) EXCEPT AS EXPRESSLY SET FORTH HEREIN AND ANY OTHER DOCUMENT DELIVERED PURSUANT HERETO, AS APPLICABLE, NONE OF THE BUYER, THE INVESTORS OR ANY OF THEIR AFFILIATES OR ANY OTHER PERSON IS RELYING ON ANY REPRESENTATIONS OR WARRANTIES OF ANY KIND WHATSOEVER, WHETHER ORAL OR WRITTEN, EXPRESS OR IMPLIED, STATUTORY OR OTHERWISE, FROM EITHER SELLER OR ANY DIRECT OR INDIRECT PARTNER, OFFICER, DIRECTOR, TRUSTEE, MEMBER, EMPLOYEE, AFFILIATE, ATTORNEY, AGENT OR BROKER OF EITHER SELLER, AS TO ANY MATTER CONCERNING THE PROPERTY OR SET FORTH, CONTAINED OR ADDRESSED IN ANY DUE DILIGENCE MATERIALS (INCLUDING, THE COMPLETENESS THEREOF), INCLUDING (A) the quality, nature, habitability, merchantability, use, operation, value, marketability, adequacy or physical condition of the Properties or any aspect or portion thereof, including, structural elements, foundation, roof, appurtenances, access, landscaping, parking facilities, electrical, mechanical, HVAC, plumbing, sewage, water and utility systems, facilities and appliances, soils, geology and groundwater; (B) the dimensions or lot size of the Properties or the square footage of any of the improvements thereon or of any tenant space therein; (C) the development or income potential, or rights of or relating to, the Properties, or the fitness, suitability, value or adequacy of a Property for any particular purpose; (D) the zoning or other legal status of any Property; (E) the compliance of any Property or its operation with any applicable codes, laws, regulations, statutes, ordinances, covenants, conditions and restrictions of any Governmental Authority or of any other person or entity (including, the Americans with Disabilities Act of 1990, as amended); (F) the ability of Buyer or any of its Affiliates to obtain any necessary governmental approvals, licenses or permits for the use or development of any Property; (G) the presence, absence, condition or compliance of any Hazardous Substances or Waste on, in, under, above or about any Property or any adjoining or neighboring property; (H) the quality of any labor and materials used in any improvements at any Property; or (I) the economics of, or the income and expenses, revenue or expense projections or other financial matters, relating to the operation of, any Property. Without limiting the generality of the foregoing, Buyer expressly acknowledges and agrees that, except as set forth herein or in any other document delivered pursuant hereto, it is not relying on any representation or warranty of either Seller or any direct or indirect partner, member, director, trustee, officer, employee, affiliate, attorney, agent or broker of either Seller, whether implied, presumed or expressly provided, arising by virtue of any statute, regulation or common law right or remedy in favor of any of them.

(d) BUYER AND INVESTORS ACKNOWLEDGE AND AGREE THAT ANY REPORTS OBTAINED BY BUYER OR ANY OF ITS AFFILIATES ARE THE SOLE RESPONSIBILITY OF BUYER AND INVESTORS AND, EXCEPT TO THE EXTENT EXPRESSLY REQUIRED PURSUANT TO THIS AGREEMENT OR ANY OTHER DOCUMENT DELIVERED PURSUANT HERETO, NONE OF SELLERS OR ANY OF THEIR AFFILIATES HAS ANY OBLIGATION TO MAKE ANY CHANGES, ALTERATIONS OR REPAIRS TO ANY PROPERTY OR ANY PORTION THEREOF OR TO CURE ANY VIOLATIONS OF LAW OR TO COMPLY WITH THE REQUIREMENTS OF ANY INSURER. THE BUYER AND INVESTORS ACKNOWLEDGE AND AGREE THAT,

EXCEPT AS OTHERWISE EXPRESSLY PROVIDED IN THIS AGREEMENT, OR IN ANY OTHER DOCUMENT DELIVERED PURSUANT HERETO, BUYER IS SOLELY RESPONSIBLE FOR OBTAINING, AS IT DEEMS NECESSARY OR APPROPRIATE, ANY APPROVAL OR PERMIT NECESSARY FOR ACCEPTANCE BY IT OF ANY PROPERTY OR INTEREST AND FOR ANY REPAIRS OR ALTERATIONS NECESSARY TO OBTAIN THE SAME, ALL AT BUYER'S SOLE COST AND EXPENSE.

### 4.5. Limited Release of Sellers.

(a) Without limiting the provisions of Section 4.4, from and after the Closing, except for rights, remedies and other provisions (including the representations, warranties and covenants) set forth in this Agreement and the Transaction Documents, Buyer irrevocably and absolutely waives its right to recover from, and forever releases and discharges, and covenants not to file or otherwise pursue any legal action against Sellers or their Affiliates or any direct or indirect partner, member, manager, trustee, director, shareholder, controlling person, Affiliate, officer, attorney, employee, agent or broker of any of the foregoing, and any of their respective heirs, successors, personal representatives and assigns, with respect to any and all suits, actions, proceedings, investigations, demands, claims, liabilities, fines, penalties, liens, judgments, losses, injuries, damages, settlement expenses or costs of whatever kind or nature, whether direct or indirect, known or unknown, contingent or otherwise (including any action or proceeding brought or threatened or ordered by any Governmental Authority), including, without limitation, attorneys' and experts' fees and expenses, and investigation and remediation costs that may arise on account of attorneys' and experts' fees and expenses, and investigation and remediation costs that may arise on account of attorneys' and experts' fees and expenses, and investigation and remediation costs that may arise on account of attorneys' and experts' fees and expenses, and investigation and remediation costs that may arise on account of attorneys' and experts' fees and expenses, and investigation and remediation costs that may arise on account of attorneys' and experts' fees and expenses, and investigation and remediation costs that may arise on account of attorneys' and experts' fees and expenses, and investigation and remediation costs that may arise on account of attorneys' and experts' fees and expenses, and investigation and remediation costs that may arise on account of attorn

(b) In connection with this Section 4.5, Buyer expressly waives the benefits of any provision or principle of federal or state Law that may limit the scope or effect of the waiver and release provisions of the preceding paragraph.

# TITLE, SURVEY AND ENVIRONMENTAL MATTERS

### 5.1. Title and Survey Matters.

(a) Prior to the execution and delivery of this Agreement, Sellers delivered or provided access to the Buyer or its counsel, for each Property, copies of: (a) the most recent title insurance policy, commitment, or pro forma whether owner's or lender's, as set forth on <u>Schedule 5.1(a)-1</u> (each, an "Existing Policy"); (b) except as set forth on <u>Schedule 5.1(a)-2</u>; the

recorded documents for each exception to title listed in each such Existing Policy; and (c) the most recent survey as set forth on Schedule 5.1(a)-1 (each, an "Existing Survey"). The Sellers have ordered from the Title Company, for delivery to the Buyer's and the Sellers' counsel, title insurance commitment(s) (the "Title Commitments") obligating the Title Company to issue for each Property an Owner's Title Policy (or leasehold policy, as applicable), in the amount of the Agreed Value allocated to such Property, together with all endorsements and affirmative coverage provided, issued or committed to be issued in connection with the Existing Policies (or if the Existing Policy was a lenders' policy, the equivalent if available and typically provided in an owner's policy), and the additional endorsements and coverage set forth on Schedule 5.1(a)-3, as are available in the applicable jurisdiction and indicated on Schedule 5.1(a)-3 (collectively, the "Extended Coverage") with a copy of all exception documents set forth therein. Sellers shall take such actions (including obtaining new surveys or updates to the Existing Surveys and providing such affidavits reasonably requested by the Title Company) to cause the Title Company to delete the general survey exception in the Title Commitments and Owner's Title Policies issued at Closing, and in its place provide a specific survey exception brought current as of the Closing Date. To the extent the Title Company requires an update to any Existing Survey or a new survey for any Property in order to delete the general survey exception in any Title Commitment or Owner's Title Policy or as a condition to obtaining a Lender's Consent, the costs of such updates and new surveys shall be paid by Sellers (which as to Lender's Consents shall be subject to the limitations set forth in Section 2.9(ii)(A)), and in all other instances the Buyer shall pay for any such new survey or updates to any Existing Survey. The Title Company shall be permitted to provide primary coverage up to an insured amount agreeable to Buyer for all Properties; however, Buyer shall have the right to require re-insurance from Fidelity National Financial, Inc. and its affiliates (collectively, the "FNF Entities"), or such other national underwriters identified by Buyer as needed, requested or as required by Buyer or its lenders. Any additional premiums incurred as a result of any re-insurance requested by Buyer (including as set forth above) shall be at Buyer's expense. Each Owner's Title Policy to be issued pursuant to the Title Commitments shall be (i) an ALTA 1992 Owner's policy form if available and permitted by the local Law of the jurisdiction where a particular Property is located, (ii) if not so available and permitted, the most comparable form permitted by such local Law, (iii) with the Extended Coverage and (iv) in the amount of the Agreed Value allocated to such particular Property (collectively, the "Owner's Title Policy"). Any endorsements required by any new lender shall be Buyer's responsibility to obtain and pay for, if and to the extent, there is any additional cost and/or liability imposed upon Sellers for obtaining such endorsements. Within ten (10) Business Days after receipt by Buyer's counsel of each respective Title Commitment, together with copies of all exception documents listed therein for the applicable Property, but in no event later than the Closing Date, Buyer's counsel shall notify the Selling Parties' counsel in writing of any title exceptions identified in any Title Commitment received by Buyer which are unacceptable to Buyer and that are not Permitted Exceptions. Buyer shall not be deemed to have received a Title Commitment until Buyer has received copies of all exception documents listed therein.

(b) Buyer and Sellers hereby agree that, as to each Property, the following items shall be deemed acceptable to Buyer and all such items shall constitute "**Permitted Exceptions**" hereunder to which Buyer may not object:

(i) all unpaid personal property, real estate and excise taxes, and all water, sewer, utility, trash and other similar charges, in each case that are not yet due and payable as of the Closing Date and/or the Deposit Date, as applicable but may become or give rise to a lien on all or any portion of such Property (it being understood that such items may be subject to apportionment at the Closing as provided herein);

- (ii) the rights of the Tenants as tenants only pursuant to Leases and Approved New Leases now in effect or which may be in effect on the Closing Date, as applicable with respect to such Property, which except as disclosed on <u>Schedule 5.1(b)(ii)</u>, do not contain any rights of first refusal or offer or options to purchase, all as disclosed in the Rent Rolls or Title Affidavits;
- (iii) all matters created or caused by or on behalf of, or with the written consent of, Buyer, including (A) any documents or instruments to be recorded as part of any financing for the acquisition of the Membership Interests by Buyer or (B) those for any Ongoing Remediation or Deferred Maintenance Conditions that Buyer is to perform or remedy under this Agreement;
  - (iv) any documents evidencing or securing the Existing Loans;
- (v) each Schedule B item shown on or referenced in each Existing Policy (excluding general exceptions which will be deleted pursuant to Title Affidavits) is a "Permitted Exception" except for the matters which are listed on Schedule 5.1(b)(v) (unless indicated as a "Permitted Exception" thereon without any reference to any further agreement or obligation by Seller or the Title Company), which (a) if not a missing document shall be deemed an "Other Exception" until the resolution with respect to such "Other Exception" as set forth on Schedule 5.1(b)(v) occurs, such matters shall then be deemed a "Permitted Exception", and (b) if a missing document shall be deemed an "Other Exception" only if its substance is such that it does not constitute a "Permitted Exception," including by reason of being a minor irregularity in title or other similar encumbrance that does not, individually or together with other minor irregularities in the aggregate, materially impair the use of the Property or the operation of the business conducted thereon by the applicable Property owner as of the date hereof;
- (vi) all laws, regulations and ordinances including all environmental, building and zoning restrictions affecting the Property or the ownership, use or operation thereof adopted by any Governmental Authority having jurisdiction over any Property or the ownership, use or operation thereof, and all amendments or additions thereto now in effect or which may be in force and effect on the Closing Date and/or the Deposit Date, as applicable with respect to such Property (collectively, the "Governmental Restrictions"); provided, however, that the existing Improvements thereon or use of the applicable Property as improved, used and operated as of the date hereof, do not violate the Governmental Restrictions;
- (vii) any lien, encumbrance or other matter identified to the Buyer in any Title Commitment or update thereof prior to the Closing pursuant to this Section 5.1 and not objected to by the Buyer in writing within the time allowed in this Section 5.1 or otherwise waived by the Buyer as provided in this Section 5.1;

(viii) any public record filings by mechanics, materialmen or other workmen or suppliers employed by any Tenant at its own expense to provide services at the Property to the extent applicable to any Tenant's leasehold interest only (to the extent, if any, that such filings constitute a lien, encumbrance or other matter);

- (ix) liens securing obligations of the relevant Property owner for which a credit will be given to Buyer at Closing; and
- (x) if not included in clauses (i) through (ix) above, any easements, rights of way, restrictions, exceptions, reservations, conditions, limitations, covenants, adverse rights or interests, licenses, minor irregularities in title and other similar encumbrances disclosed in the Title Commitments which were not previously disclosed in the Existing Policy and/or Existing Survey, and do not in the aggregate materially impair the use of such Property or the operation of the business conducted thereon by the applicable Property owner as of the date hereof.
- (c) Sellers shall have the Title Commitments updated from time to time up to and through the Closing Date as may be reasonably requested by the Buyer, and Buyer may give the Selling Parties' counsel written notice of any additional exceptions, objections and encumbrances to title that are not Permitted Exceptions within five (5) Business Days after receipt of any updated Title Commitments. Any exceptions, objections and encumbrances to title that are raised by Buyer on a timely basis and which are not Permitted Exceptions or Mandatory Removal Exceptions, shall be referred to as collectively, "Other Exceptions." Buyer shall not be entitled to raise any title or survey objections after the Closing except for those which are Mandatory Removal Exceptions.
- 5.2. Resolution of Other Exceptions. Sellers shall, and shall cause the Company Group to, use commercially reasonable efforts to satisfy or cure each duly and timely raised Other Exception, including those set forth on Schedule 5.1(b)(y), at Sellers' sole cost and expense. Notwithstanding the foregoing, except for Mandatory Removal Exceptions (and the timely filing of any claims under the Existing Policies and compliance with the terms thereof), Selling Parties shall not be required and are not obligated to bring any action or proceedings, convey or acquire any interest in real property, incur any expense or liability in excess of \$5,000,000 in the aggregate with respect to the removal or cure of Other Exceptions or, except as set forth in the following sentence, take any other action of any kind or nature to render title to any of the Properties free and clear of any title or survey exceptions, objections or encumbrances, and Buyer shall have no right of specific performance or other relief against the Selling Parties to cause any Other Exceptions to be satisfied or cured. If after Sellers have used good faith and commercially reasonable efforts, including trying to obtain title insurance endorsements, providing affidavits and certificates and indemnities to the title insurer (but subject in all events to the \$5,000,000 aggregate potential liability set forth above), the Sellers are unable, or fail to remove, satisfy or cure any such Other Exceptions, Sellers shall notify Buyer in writing, no later than fifteen (15) Business Days prior to Closing of those Other Exceptions which Sellers are unable to satisfy or cure, then Buyer shall have the right to elect in writing, no later than ten (10) Business Days prior to Closing, to (a) waive any rights with respect to such Other Exceptions (in which event Buyer shall be obligated to and shall consummate the Closing without abatement of the Purchase Price or any claim for damages or

liabilities against Sellers by reason of such Other Exceptions, and such Other Exceptions shall become part of and deemed Permitted Exceptions) or (b) indicate Buyer's good faith estimate of the loss or other diminution in value caused by such Other Exception(s) for the particular Property aggregated with all Other Exceptions on all other Properties ("Alleged Diminution of Value"). Upon receipt of such estimate, Sellers shall have five (5) Business Days (and the Closing shall be adjourned during such period) to elect by written notice to Buyer to: (i) exclude any individual Property in accordance with and subject to Section 2.13 if the Alleged Diminution of Value with respect to any one Property alone exceeds ten percent (10%) of the Agreed Value of the particular Property; (ii) if the aggregate Alleged Diminution of Value as estimated by Buyer for all Other Exceptions on all the Properties is equal to or more than \$250,000, cause Buyer to Close with a credit against the Purchase Price equal to the Alleged Diminution of Value less \$250,000; (iii) if the aggregate Alleged Diminution of Value estimated by Buyer is less than \$250,000, cause Buyer to Close or (iv) submit the Alleged Diminution of Value to Baseball Arbitration and proceed to Closing in accordance with (ii) above except that the Alleged Diminution of Value less \$250,000 shall be held in escrow until the amount is finally determined by Baseball Arbitration. For purposes of determining the Alleged Diminution of Value, all relevant factors shall be considered, including the following: (a) the magnitude of the matter; (b) the impact of the matter upon the fair market value and future transferability of the Property; (c) the likelihood the exception will be enforced or pursued by a third party; and (d) the likelihood of success by said third party. The inability of Buyer and Sellers to resolve any disputes regarding Other Exceptions or of Sellers to complete the steps to remove, satisfy or cure Other Exceptions prior to the Deposit Date shall not be groun

5.3. Sellers' Obligations Regarding Title Objections. Notwithstanding the provisions of Section 5.2 hereof, the Selling Parties covenant and agree that at or prior to Closing, the Selling Parties shall cause the Company Group, at Sellers' sole cost and expense, to: (i) pay in full and cause to be canceled and discharged, bonded or otherwise cause the Title Company to insure over, all mechanics' and contractors' Liens which encumber any Property as of the Closing Date, other than such liens which only encumber a Tenant's leasehold estate or are reimbursable by a Tenant under a Lease; (ii) pay all Liens or encumbrances against such Property or any personal property that is used in connection therewith which evidence monetary obligations of any Seller or any entity in the Company Group (excluding the Existing Loans) including Liens for past due taxes and assessments; (iii) cause to be released any loan security documents of any Seller or any entity in the Company Group which encumbers any Property or personal property, except for the loan documents evidencing and securing the Existing Loans; and (iv) pay and obtain the release of all judgments or Liens against any Seller or any entity in Company Group on such Property or personal property that is used in connection therewith. The Liens, judgments and encumbrances specifically referred to in this Section 5.3 are collectively referred to as ("Mandatory Removal Exceptions"). Mandatory Removal Exceptions do not have to be paid or satisfied at the Deposit Date in order to satisfy the condition in Section 11.2(f) to payment of the Deposit or the corresponding delivery requirement in Section 12.1(a)(xii), but must be satisfied by Sellers prior to or at Closing, and if not so satisfied, shall be satisfied at Closing out of the proceeds otherwise payable to Sellers without adjustment to the Purchase Price.

- **5.4.** <u>Postponement of Closing</u>. Sellers shall have the right, in their sole and absolute discretion, to adjourn the Closing Date, if necessary, from time to time by notice to Buyer delivered on or prior to the date then set for Closing (for a maximum adjournment period not to exceed fifteen (15) Business Days in the aggregate) to permit the cure of any duly and timely made Other Exceptions.
- 5.5. Cure of Title Objections. Notwithstanding anything herein to the contrary, Sellers shall be deemed to have removed, satisfied or cured each objection that is not a Permitted Exception if, in Sellers' discretion and at their sole cost and expense and without the Buyer or Company Group incurring any liability or obligations in connection therewith (monetary or otherwise), Sellers either: (a) take such actions as are necessary to remove, satisfy or cure (of record or otherwise, as appropriate) such objection; (b) cause the Title Company to remove such objection as an exception to title in the Owner's Title Policy issued at Closing or affirmatively insure against the same (which insurance shall be satisfactory to Buyer in form and substance), in either case without any additional cost to Buyer or Company Group, whether such removal or insurance is made available in consideration of payment, bonding, indemnity of Sellers or otherwise; or (c) deliver (i) their own funds (or direct that a portion of the Purchaser Price be delivered), in an amount needed to fully discharge any such objection, to the Title Company with instructions for the Title Company to apply such funds to fully discharge such objection, and (ii) if required by the Title Company, such instruments, in recordable form, as are necessary to enable the Title Company to discharge such Other Exception(s) of record. To the extent that funds are needed to cure any Mandatory Removal Exception or Other Exception, Sellers shall have the right to cause any portion of the Purchase Price being paid at Closing to be delivered to the Title Company to cure same.

### 5.6. Environmental Remediation and Insurance.

# (a) Ongoing Remediation Projects.

(i) <u>Completion of Ongoing Remediation Projects</u>. Prior to Closing, the Sellers shall cause the Company Group to use commercially reasonable efforts to continue the ongoing environmental remediation projects identified on <u>Schedule 5.6(a)(i)</u> ("**Ongoing Remediation Projects**") at the Properties identified on <u>Schedule 5.6(a)(i)</u> ("**Ongoing Remediation Sites**"). Following Closing, Buyer shall cause the Company Group to use commercially reasonable efforts to promptly commence and diligently pursue to completion the Ongoing Remediation Projects and seek appropriate "no further action" letters or equivalent determinations from the appropriate Governmental Authority for each Ongoing Remediation Project.

### (ii) Funding.

(a) Seller has existing contracts with Apex to perform work in connection with the four Ongoing Remediation Projects identified with an asterisk on Schedule 5.6(a)(i) ("Apex Contracts"). The work to be performed under the Apex Contracts is described in the proposals for work from Apex listed on Schedule 5.6(a)(i). Seller shall pay the Remediation Costs of completing the work described in the proposals listed on Schedule 5.6(a)(i) out of its own funds.

- (b) Buyer shall be entitled to draw upon the funds held in the Ongoing Remediation Escrow to pay Remediation Costs incurred by Buyer in performing the Ongoing Remediation Projects. As set forth on <a href="Schedule 5.6(a)(ii)">Schedule 5.6(a)(ii)</a>, each Ongoing Remediation Project shall be allocated a share of the Ongoing Remediation Escrow.
- (iii) <u>Disbursements to Sellers</u>. On a quarterly basis, beginning July 1, 2005, Buyer and Seller shall review the status of the Ongoing Remediation Projects. With respect to any Ongoing Remediation Project for which Buyer has obtained "no further action" letters or equivalent determinations prior to the end of such quarter (including prior to the Closing Date) from the appropriate Governmental Authorities, Sellers shall be entitled to draw upon the funds held in the Ongoing Remediation Escrow an amount equal to 80% of the difference between such Ongoing Remediation Project's allocated share of the Ongoing Remediation Escrow, as reflected on <u>Schedule 5.6(a)(ii)</u>, and the amounts paid from the Ongoing Remediation Project Escrow for such Ongoing Remediation Project.
- (iv) <u>Liquidation of Ongoing Remediation Project Escrow</u>. Once the Buyer has obtained "no further action" letters or equivalent determinations from appropriate Governmental Authorities with respect to all Ongoing Remediation Projects, the Escrow Agent shall disburse to Sellers the remaining balance in the Ongoing Remediation Project Escrow. In the event one or more Ongoing Remediation Projects are not complete and "no further action" letters or equivalent determinations for all of the Ongoing Remediation Projects have not been obtained by the second anniversary of the Closing, the Buyer and Sellers shall make a reasonable, good faith estimate of the total future costs and expenses (including a reasonable contingency reserve, if appropriate) necessary to complete all remaining Ongoing Remediation Projects and obtain "no further action" letters or equivalent determinations for all remaining Ongoing Remediation Projects. If Buyer and Sellers cannot agree on such amount, it shall be determined by Baseball Arbitration. An amount equal to the estimated total future costs for Remediation with respect to the remaining Ongoing Remediation Projects shall be disbursed from the Ongoing Remediation Project Escrow to Buyer. To the extent additional funds remain in the Ongoing Remediation Project Escrow after such disbursement to Buyer, then the remaining balance shall be distributed 100% to Sellers.

# (b) Future Remediation Projects.

(i) <u>Designation</u>. On or before 5:00 p.m. Eastern Time on February 18, 2005, Buyer, by written notice to Sellers, shall designate not more than 15 Properties as potential "**Future Remediation Sites**" as to which Buyer plans to perform additional environmental investigation prior to Closing including, but not limited to, sampling and analysis of soil and groundwater ("**Phase II Investigation**"). After such Phase II Investigation and within 10 days prior to Closing, Buyer's consultant shall prepare and deliver to Sellers reports ("**Phase II Reports**") which identify the conditions detected and any recommended Remediation with respect to such detected environmental conditions ("**Future Remediation Projects**").

Following Closing, Buyer shall cause the Company Group to use commercially reasonable efforts to promptly commence and diligently pursue to completion the Future Remediation Projects and seek appropriate "no further action" letters or equivalent determinations from the appropriate Governmental Authority for each Future Remediation Site.

- (ii) <u>Cap on Seller's Remediation Liability for Future Remediation Sites</u>. Sellers' responsibility for the cost of Future Remediation Projects shall be limited to the amounts Sellers contribute to the Initial Remediation Estimate (<u>i.e.</u>, 75% of the Initial Remediation Estimate) plus the amounts actually recovered by Buyer, Sellers or Company Group, if any, pursuant to the American International Specialty Lines Insurance Company Commercial Pollution Legal Liability Insurance Policy, Policy Number PLC 195 42 69, with a policy period from February 1, 2001 through May 1, 2006 ("Sellers' Environmental Insurance").
- (iii) Adjustment of Initial Remediation Estimate. 180 days after the Closing, Buyer and Sellers shall review the status of the Future Remediation Sites and shall make a reasonable, good faith estimate of the total future costs and expenses (including a reasonable contingency reserve, if appropriate) necessary to complete all remaining Future Remediation Projects and obtain "no further action" letters or equivalent determinations for all remaining Future Remediation Sites ("Future Remediation Estimate"). If the parties cannot agree on the Future Remediation Estimate, it shall be determined by Baseball Arbitration. The Future Remediation Estimate agreed to by the parties, or, if the parties do not agree, an amount equal to 125% of the Future Remediation Estimate determined through Baseball Arbitration, shall be retained in the Future Remediation Escrow for subsequent disbursement and the Escrow Agent shall disburse the remaining balance of the Future Remediation Escrow 75% to Sellers and 25% to Buyer.
- (iv) Final Disbursements from Future Remediation Escrow. Once the Future Remediation Projects are completed and the Buyer has obtained "no further action" letters or equivalent determinations from appropriate Governmental Authorities with respect to all Future Remediation Sites and all Remediation Costs for the Future Remediation Projects have been paid, Escrow Agent shall disburse the remaining balance of the Future Remediation Escrow 75% to Sellers and 25% to Buyer. In the event all Future Remediation Projects are not complete and "no further action" letters or equivalent determinations for all of the Future Remediation Projects have not been obtained by the second anniversary of the Closing, the Buyer and Sellers shall make a reasonable, good faith estimate of the total future costs and expenses ("Post Escrow Future Remediation Costs") (including a reasonable contingency reserve, if appropriate) necessary to complete all remaining Future Remediation Projects and obtain "no further action" letters or equivalent determinations for all remaining Future Remediation Sites. If Buyer and Sellers cannot agree on such amount, it shall be determined by Baseball Arbitration. Post Escrow Future Remediation Costs shall not include Remediation Costs which the insurer under Sellers' Environmental Insurance has agreed in writing are insured under Sellers' Environmental Insurance and for which such insurer has agreed to reimburse Buyer directly. An amount equal to the Post Escrow Future Remediation Costs shall be disbursed to Buyer from the Future Remediation Escrow. To the extent funds are available in the Future Remediation Escrow after such disbursement, the remaining balance shall be distributed 75% to Sellers and 25% to Buyer.

(c) Performance of Remediation. With respect to any Remediation Costs (whether with respect to Future Remediation Sites or Ongoing Remediation Sites) for which Buyer seeks or will seek reimbursement pursuant to this Section 5.6, Buyer shall: (i) provide written notification to Sellers that it intends to perform Remediation prior to commencing any such Remediation except that such notice may be provided following commencement of such Remediation in the event of an emergency; (ii) provide Sellers with a reasonable opportunity to comment in advance upon any material written communications, filings, reports, correspondence or other writings given to any Governmental Authority in connection with such Remediation provided that Sellers will provide their comments in a timely fashion and Buyer will consider timely provided comments in good faith; (iii) to the extent practical, provide Sellers with a reasonable opportunity to participate in any meetings with any Governmental Authority regarding the Remediation; (iv) use commercially reasonable efforts to perform the Remediation in accordance with Environmental Laws; (v) use commercially reasonable efforts to minimize costs in conducting the Remediation in accordance with Environmental Laws; (vi) use commercially reasonable efforts to employ cost-effective remedial methods that are commercially reasonable under the circumstances; (vii) use commercially reasonable efforts to apply risk based clean-up standards; if available, (viii) to the extent available, employ deed restrictions, engineering controls and institutional controls consisting of prohibitions on the use of groundwater, notifications to contractors or utility workers and maintaining existing pavement; (ix) use cleanup standards predicated on the non-residential, non-agricultural use of the Property to the extent available; (x) allow Sellers or their agents reasonable access to the Property for purposes of observing the Remediation, so long as Sellers or such agents do not interfere with the Remediation or the Company Group's operations; (xi) keep Sellers reasonably informed of the progress of any such Remediation and the schedule for completing such Remediation; (xii) provide to Sellers copies of all material written communications, filings, reports, correspondence or other writings, photographs or materials received from any Person (including any Governmental Authority) in connection with any such Remediation and such other written materials or information concerning the Remediation that Sellers reasonably request; and (xiii) to the extent notified of same in writing, use commercially reasonable efforts to comply with all terms, covenants and conditions required or reasonably requested by the provider of Sellers' Environmental Insurance in order for Sellers to get reimbursed for the costs of the Future Remediation by such provider. Notwithstanding anything in the above to the contrary, Buyer's failure to comply with the provisions set forth in this paragraph shall not limit or reduce Buyer's right to reimbursement from Sellers, the Ongoing Remediation Escrow or the Future Remediation Escrow for Remediation, except to the extent that Sellers are actually prejudiced by such noncompliance.

(d) <u>Reimbursement for Remediation</u>. With regard to Remediation Costs for which Buyer seeks reimbursement under this Section 5.6, Buyer shall first present such request for reimbursement to Sellers along with invoices reflecting the actual third party costs and expenses in performing such Remediation. If Sellers have not reimbursed Buyer within thirty (30) days for such Remediation Costs, Buyer may then seek disbursement of such Remediation Costs from the Ongoing Remediation Escrow or the Future Remediation Escrow, as appropriate. The disbursement of such funds from the Ongoing Remediation Escrow or the Future Remediation Escrow shall occur within ten (10) days of request for disbursement by Buyer. Buyer and Sellers acknowledge that Sellers may seek to recover from Sellers' Environmental Insurance certain of these Remediation Costs. In the event the Sellers or the Company Group

receive any payments with respect to the Sellers' Environmental Insurance for costs of Future Remediation previously paid from the Future Remediation Escrow, such funds shall be deposited into the Future Remediation Escrow if received prior to the dissolution of the Future Remediation Escrow. Any payments with respect to the Sellers' Environmental Insurance for costs of Future Remediation received after dissolution of the Future Remediation Escrow (other than payments directly to Buyer which are excluded from calculation of Post Escrow Future Remediation Costs pursuant to Section 5.6(b)(iv)) shall be disbursed to Buyer and Seller as if the funds were in the Future Remediation Escrow at the time of its dissolution pursuant to Section 5.6(b)(iv). Buyer agrees to use commercially reasonable efforts to pursue available state trust funds to recover Remediation Costs with respect to the Riverside Square Ongoing Remediation Project. Buyer shall deposit into the Ongoing Remediation Escrow or the Future Remediation Escrow, as applicable, the amounts received by Buyer from any state trust fund with respect to the Ongoing Remediation Projects or the Future Remediation Projects, less Buyer's actual costs in pursuing such amounts. Any Remediation Costs recovered from state trust funds after the liquidation of the applicable Escrow shall be retained by Buyer.

- (e) <u>Estimates for Remediation Costs</u>. All cost estimates required under this Section 5.6 for Ongoing Remediation Projects and for Future Remediation Projects shall be based upon cleanup standards, to the extent available, predicated on the non-residential, non-agricultural use of the real property, and shall assume that, to the extent commercially reasonable, all Remediations will employ (i) cost-effective remedial measures, and (ii) to the extent available, (A) risk-based clean-up standards, and (B) deed restrictions, institutional controls and engineering controls consisting of prohibitions on the use of groundwater, notifications to contractors and utility personnel, and maintaining existing pavement.
- (f) <u>Buyer's Environmental Insurance Policy</u>. Sellers and Buyer agree that at Closing Buyer shall obtain (at Sellers' expense with one-half of the cost of the premium and any surplus lines tax to be added to the Purchase Price) an environmental insurance policy in substantially the form attached as <u>Schedule 5.6(f)</u>, with applicable endorsements, covering the period from the Closing and ending on the 10<sup>th</sup> anniversary date of the Closing, with a \$50,000,000 Policy Limit and \$100,000 deductible per Property, issued by American International Specialty Lines Insurance Company or another comparable insurance company with at least as strong an A.M. Best rating, selected solely in Buyer's discretion, insuring Buyer and the Company Group, covering all the Properties except (i) any matter subject to an Ongoing Remediation until such time as Buyer obtains a "no further action" letter or equivalent determination for the relevant Ongoing Remediation Project and (ii) any matter at the Future Remediation Sites until Buyer's Phase II Reports determine that no remediation is recommended, or if remediation is recommended, until the Buyer obtains a "no further action" letter or equivalent determination with respect to such Future Remediation Project ("Buyer's Environmental Insurance Policy"). Notwithstanding the foregoing, except as provided in the next sentence, in no event shall Sellers be obligated to purchase the Buyer's Environmental Insurance Policy if the cost of such policy exceeds \$10,000,000. Buyer may request that the Buyer's Environmental Insurance Policy contain additional terms, conditions or coverage (including lower deductible, increased policy limits or longer term), provided that the additional cost resulting therefrom shall increase the Price Adjustment pursuant to Section 2.2(a)(x).

### 6. RISK OF LOSS

#### 6.1. Condemnation.

(a) If, prior to the Closing Date, all or any portion of any Property is taken by condemnation or eminent domain, or is the subject of a pending taking which has not been consummated, Selling Parties shall notify Buyer of such fact promptly after Selling Parties obtain knowledge thereof. Buyer shall have no right to terminate this Agreement for any condemnation. Buyer shall receive at the Closing all condemnation proceeds (or, if such have not been awarded, all of Selling Parties' and the Company Group's right, title and interest therein). Seller shall not enter into any settlement with respect to any condemnation proceeding, deed or settlement in lieu of condemnation, insurance settlement or similar agreement without Buyer's prior written consent. Buyer shall have the right, with Seller's participation, to lead and conduct all negotiations with respect to any such settlement. The provisions of this Section 6.1 shall be the sole remedy for Buyer for losses arising from any condemnation and Buyer shall not be entitled to any Price Adjustments or indemnifications pursuant to Section 15 by reason of Losses arising from any condemnation.

### 6.2. Casualty.

- (a) If, prior to the Closing Date, all or any portion of the Property is destroyed or damaged by fire, earthquake, hurricane or other casualty, Selling Parties shall notify Buyer of such fact promptly after Selling Parties obtain knowledge thereof. Buyer shall have no right to terminate this Agreement for any casualty. Unless required to do so by Law or pursuant to the terms of any Leases or Existing Loans, Selling Parties shall not be obligated to repair any damage or destruction (but they may do so and use insurance proceeds for same, as required to protect against injury to person or property).
- (b) If the reasonably estimated economic impact following the Closing and diminution in the value of the Property as a result of any casualty is agreed by the parties to be less than 20% of the Agreed Value of the particular Property, then Selling Parties shall cause Company Group to assign, without recourse, and turn over to Buyer all of the insurance proceeds, net of any costs of repairs and net of reasonable collection costs (or, if such have not been awarded, all of the Selling Parties' and the Company Group's right, title and interest therein) payable with respect to such casualty, including any business or rental interruption insurance proceeds for such casualty applicable to the period after Closing, and the parties shall proceed to Closing pursuant to the terms hereof without abatement of the Purchase Price except for a credit to Buyer in the amount of (x) any uninsured or underinsured portion of the loss including the applicable insurance deductible or self-insurance amount, if any and (y) the amount of casualty proceeds paid to and retained by a Lender, Tenant or other party in interest.
- (c) Notwithstanding the foregoing, if Buyer believes that the reasonably estimated economic impact following the Closing and diminution in the value of a Property as a result of a casualty is more than 20% of its Agreed Value (determined in accordance with applicable factors for determination of value set forth in Section 6.2(d) below), Buyer, by written notice to Sellers, may elect to reduce the Purchase Price by such reasonably estimated economic impact following the Closing and diminution in the value of the Property as a result of the casualty and Sellers shall be entitled to retain the related insurance proceeds.

- (d) The reasonably estimated economic impact following the Closing and diminution in the value of a Property as a result of a casualty shall be determined by Baseball Arbitration if the parties are unable to agree and any disputed amounts shall be held in escrow from and after the Closing. The determination of diminution in value shall include consideration of factors such as whether the Property can be rebuilt in its present size and use under current zoning laws and the length of time necessary to rebuild the Property and resulting loss of rental income, the sufficiency of insurance proceeds paid or available to Buyer (plus any deductible paid to Buyer) to repair and/or replace any improvements damaged by the applicable casualty and/or the sufficiency of business interruption and/or rental insurance paid or available to Buyer to replace rental income that would have been calculated during the period of time of the repair and/or restoration.
- (e) The provisions of this Section 6.2 shall be the sole remedy for Buyer for losses arising from any casualty event and except as provided in this Section 6.2, Buyer shall not be entitled to any Price Adjustment or indemnification pursuant to Section 15 (Breaches, Remedies and Indemnification) by reason of losses arising from a casualty event.

### 7. COVENANTS.

- **7.1.** <u>Implementing Agreement</u>. Subject to the terms and conditions hereof, each party hereto shall use good faith, commercially reasonable efforts: (i) to perform in all material respects its obligations under this Agreement; (ii) to obtain the Lender Consents with respect to all Existing Loans; (iii) to cause the conditions to the payment of the Deposit and the Closing to be satisfied; and (iv) to facilitate the consummation of the sale of the Membership Interests and other Transactions.
- **7.2. General Covenants**. Except as otherwise provided for or contemplated by this Agreement, from the date of this Agreement to the Closing, the Selling Parties shall cause the Company Group:
- (a) to operate the Properties in the ordinary and usual course of business and consistent in all material respects with past practice (it being understood that this Section 7.2 shall not be interpreted as requiring the Company Group to continue any expansion, renovation or other capital expenditure programs, except as provided in Section 7.2(i));
- (b) to use its commercially reasonable efforts to preserve the Company Group's good will and advantageous relationships with tenants, customers, suppliers, independent contractors and other persons material to the operation of the Properties;
- (c) to confer on a regular basis with one or more representatives of the Buyer to report on material operational matters and any proposals to engage in material transactions;
- (d) to promptly notify the Buyer (to the extent Sellers have Knowledge of same) of any material emergency or other material change in the condition (financial or otherwise) of the business, Properties, of any material governmental complaints, investigations

or hearings (or written communications indicating that the same may be contemplated), or of any breach in any material respect of any representation, warranty or covenant contained in this Agreement or any event or development which would constitute a breach in any material respect of any representation or warranty if such event or development existed as of the date hereof;

- (e) to provide the Buyer with a reasonable opportunity to review and comment on any federal income tax returns filed by the Companies or any entity of the Company Group;
- (f) to keep the Buyer informed in a timely manner regarding any communications to or filings with any state or federal environmental regulatory authorities regarding the Properties;
- (g) not to submit any written communication or filing to any state or federal environmental authority without Buyer's Consent unless required by applicable Laws or pursuant to a request of an insurance carrier or pursuant to the reasonable recommendation of an environmental professional or advisor;
- (h) if the Sellers decide to discuss or negotiate with any Person regarding a Competing Transaction, promptly advise Buyer of the existence of such discussions or negotiations and any material developments in such discussions or negotiations (for purposes herein, a "Competing Transaction" means the sale or assignment, or other transfer, directly or indirectly, of the ownership or economic benefits of all or substantially all the Properties or equity interests in any entity in the Company Group, whether through direct sale, merger, consolidation, asset sale, exchange, recapitalization, other business combination, liquidation, or other action out of the ordinary course of business, provided, however, that contracts for sales and transfers after Properties have been duly designated as Dropped Assets shall not constitute a Competing Transaction); and
- (i) to cooperate reasonably with Buyer regarding undertaking specified projects relating to Deferred Maintenance Conditions requested by Buyer, provided the Purchase Price is subject to adjustment for such costs pursuant to Section 2.2(a)(vi) (Price Adjustments).
  - **7.3. Special Covenants**. (The covenants in this Section 7.3 are referred to hereinafter as "**Special Covenants**".)
- (a) <u>Prohibitions</u>. Except as otherwise provided for or contemplated by this Agreement (or pursuant to Buyer's Consent), from the date of this Agreement until the Closing, the Selling Parties shall cause the Company Group and, as to the Membership Interests, Sellers shall:
- (i) Except as set forth in <u>Schedules 8.17</u> (Material Contracts) and <u>8.20</u> (No Other Contracts), not to acquire, enter into a contract or option to acquire or exercise an option or contract to acquire, additional real property or any equity or ownership interest in any entity;
- (ii) not to incur additional indebtedness (including refinancing any existing indebtedness); provided, however, that the Company Group shall have the right, without

Buyer's Consent, to borrow money under their existing lines of credit and otherwise in the ordinary course of business on condition that all such borrowings be fully repaid and all security interests and agreements documenting same be fully satisfied and released prior to Closing;

- (iii) except for the permitted sales and other dispositions set forth on <u>Schedule 7.3(a)(iii)</u>, not to sell, or otherwise dispose of (a) any Properties or any portion thereof or any of the capital stock of or partnership, membership or other interests in any of the entities comprising the Company Group or (b) except in the ordinary course of business, any of its other assets which, in the aggregate, are material in nature;
- (iv) not to make any loans or capital contributions to investments in, or guarantee the liabilities of, any Person other than within the Company Group;
- (v) not to mortgage or create a lien, security interest or other security for the payment of money on the Properties or the other material assets of the Company Group that will not be fully satisfied and released prior to Closing other than Permitted Exceptions;
- (vi) not to enter into any agreement with any officer, director, consultant or Affiliate of the Company Group or Sellers or any family member thereof that will survive Closing;
  - (vii) not to amend or modify the terms or conditions of any Existing Loan or the documents evidencing or securing any Existing Loan;
- (viii) not to authorize for issuance, issue, sell, deliver or agree or commit to issue, sell or deliver (whether through the issuance or granting of options, warrants, convertible or exchangeable securities, commitments, subscriptions, rights to purchase or otherwise) any shares of capital stock, partnership interests, limited liability company or membership interests or other securities or equity interests or any right or interest therein;
  - (ix) not to admit any Person as a member or partner (other than admitting Buyer as member of the Companies at the Closing);
- (x) not to amend or otherwise modify any certificate or articles of incorporation or formation, any by-laws, any limited partnership agreement or limited liability company or operating agreement, or other Organizational Documents;
- (xi) not to amend or otherwise modify any of the terms of any capital stock, partnership interests, limited liability company or membership interests or other securities or equity interests;
- (xii) not to declare, set aside or pay any dividend or distribution of securities or property or any combination thereof (except for cash distributions to any member of the Company Group and/or Sellers paid before Closing) in respect of, or redeem or otherwise acquire, capital stock, partnership interests, limited liability company or membership interests or other securities or equity interests;

(xiii) not to dissolve, merge, consolidate, convert or otherwise change entity form;

(xiv) not to sell, transfer, or permit any security interest or other encumbrance that will not be fully satisfied and released prior to Closing on any of the Membership Interests or other shares of capital stock, partnership interests, limited liability company or membership interests or other securities or equity interests of any member of the Company Group;

(xv) not to take or permit any action that would cause the representations and warranties in Sections 8.1(c) (Authorization), 8.2(c) (Sellers' Authorization), 8.2(e) (Title), the first sentence of Section 8.3 (Ownership of Properties), Section 8.8 (Tax Matters), or 8.11 (No Condemnation) to cease to be true as of the Deposit Date or the Closing Date.

(xvi) not to make any Tax election without Buyer's Consent, which shall not be unreasonably withheld; and

(xvii) not to undertake any development or new construction project for a new shopping center or for the development of additional Improvements creating rentable square feet on a Property or engage in any new line of business.

- (b) <u>Affirmative Covenants</u>. Except as otherwise provided for in or contemplated by the Agreement (or pursuant to Buyer's Consent), from the date hereof to Closing, Selling Parties shall cause the Company Group:
- (i) to maintain all insurance policies set forth on Schedule 8.22 (or policies that provide coverage that is not materially less favorable to Company Group by insurance companies with an A.M. Best rating that is not materially less strong, to the extent reasonably available in the insurance market), in full force and effect up to and including the Closing Date, to add Buyer as an additional insured, as its interests may appear under the insurance policies up until the Closing and to discuss with Buyer in advance any change in or claim made under any insurance coverage; and
  - (ii) to pay, when due, all principal and interest payments on the Existing Loans.
- (c) New Contracts. Except as otherwise provided for in or contemplated by the Agreement (or with Buyer's Consent), from the date hereof until the Closing, Selling Party shall cause the Company Group to refrain from entering into new contracts, or entering into modifications, amendments, extensions or renewals of any existing contract (including without limitation Service Contracts). Notwithstanding the foregoing, Sellers and/or the Company Group shall have the right, without prior notice to Buyer or Buyer's Consent, to enter into new contracts and/or modifications, amendments extensions or renewals of existing Contracts provided that such new contract and/or modification, amendment, extensions or renewals of an existing Contract either:
  - (i) will not be an obligation affecting the Properties or binding on the Company Group after the Closing, and/or

- (ii) was entered into in the ordinary course of business and either involves payments following the Closing which total less than \$10,000 in the aggregate (for each applicable contract modification, amendment, extension or renewal) or are terminable for any reason without any termination fee either upon less than thirty-two (32) days' notice or upon the Closing.
- (d) <u>Tenant Leases</u>. Except as otherwise provided for or contemplated by this Agreement (or pursuant to Buyer's Consent), from the date hereof until the Closing, Selling Parties shall cause the Company Group:
  - (i) not to provide a written waiver of any rights under the Leases;
- (ii) not to terminate, modify, amend, extend, renew or expand any Lease with a Major Tenant or any Lease in excess of 5,000 square feet (except for terminations, amendments, modifications, extensions, renewals or expansions required by the existing terms of any such Leases); or
  - (iii) not to enter into any new Lease with a Major Tenant or in excess of 5,000 square feet.

Selling Parties shall have the right, but not the obligation, without Buyer's Consent, to cause the Company Group to enter into any new Lease of less than 5,000 square feet (or any modification, amendment, extension, renewal or expansion of any existing Lease that is not with a Major Tenant or a Lease that is not more than 5,000 square feet) that satisfies all of the following conditions (collectively, the "Leasing Parameters"): (i) the tenant improvement allowance for such proposed transaction shall have a value equal to or less than ten dollars per square foot (\$10.00/sq. ft.); (ii) leasing commissions for such proposed transaction shall be at a rate equal to or less than six percent (6%) of the base rent due under any such proposed transaction; (iii) the base rent for the applicable space in the applicable Property shall not be less than the scheduled base rent indicated for such space in the Argus data file contained on the CD accompanying the Selling Parties' October 2004 Portfolio Executive Summary prepared by Sellers and delivered to Buyer; and (iv) the free rent period for the applicable space on the applicable Property shall not be greater than 60 days (and each such modification, amendment, extension, renewal or expansion of existing Leases and/or new Lease, together with any such Lease and modifications, amendments, extensions, renewals or expansions of existing Leases entered into with Buyer's Consent, shall also be deemed to be an "Approved New Lease"). Selling Parties shall provide to Buyer copies of all amendments, modifications, new Leases, notices of default or events of default received or given by any Selling Party within three (3) Business Days of the date such notice is given or received by Selling Party and; provided, further, that Selling Parties shall not declare any Major Tenant in default under its Lease or commence any legal proceedings related thereto without Buyer's consent.

(e) Ground Leases. Sellers shall cause the Company Group not to terminate, amend or modify any Ground Lease without Buyer's Consent.

- (f) <u>Taxable REIT Subsidiary Election</u>. Buyer may make, or require Sellers to make or cause to be made, an election under Section 856(1) of the Code, effective as of the Deposit Date, to treat any entities in the Company Group that are treated as corporations for federal income tax purposes as "taxable REIT subsidiaries" with respect to each of Regency and Macquarie CountryWide (US) No. 2 Corporation or another REIT designated by MCW that is a subsidiary of MCW, and Sellers and Company Group shall cooperate in connection with making such election; <u>provided</u>, <u>however</u>, Sellers shall only be required to take such actions with respect to any such election as shall be specifically prescribed by Buyer in writing to the Sellers and Sellers shall not in any event have any liability to Buyer by reason of such actions not being effective for purposes of causing one or more of the designated entities to be treated as a taxable REIT subsidiary provided that Sellers have timely taken such actions.
- (g) <u>Buyer's Consent</u>. "**Buyer's Consent**" shall mean (a) Buyer's actual written consent or (b) with respect to Sellers' requests for consent made reasonably and in good faith regarding leasing matters and property maintenance and management matters, Buyer's deemed consent which shall be deemed given if Buyer does not respond within four (4) Business Days to a written request for Buyer's Consent with a written explanation of its denial of its consent. Buyer's Consent shall not be unreasonably withheld. In order for the notice to Buyer be sufficient for deemed consent within four (4) Business Days, such notice request to Buyer must be marked in bold lettering with the following language: "BUYER'S RESPONSE IS REQUIRED WITHIN FOUR (4) BUSINESS DAYS OF RECEIPT OF THIS NOTICE PURSUANT TO THE TERMS OF THE PURCHASE AGREEMENT OR BUYER'S CONSENT SHALL BE DEEMED GIVEN". A copy of such notice must be sent to Linda Y. Kelso, Esq., Foley & Lardner LLP, One Independent Drive, Suite 1300, Jacksonville, Florida 32202.
- (h) <u>Conveyance of Management and Southside Nominee</u>. Selling Parties shall cause the Company Group to transfer outside of the Company Group prior to Closing the ownership of First Washington Management, Inc. and Southside Nominee, Inc., if permitted by applicable contractual obligations.

## 3. JOINT AND SEVERAL REPRESENTATIONS AND WARRANTIES OF THE SELLERS

Any reference in this Agreement to Sellers making available documents to Buyer shall be deemed to mean at the offices of First Washington Realty, Inc. located at 4350 East-West Highway, Suite 400, Bethesda, Maryland 20814. The Sellers, jointly and severally, represent and warrant to Buyer that, as of the date hereof:

### 8.1. Organization; Authorization.

- (a) <u>Organization</u>. Each entity in the Company Group is a limited liability company, partnership or corporation duly organized, validly existing and in good standing under the laws of the state of its organization.
- (b) <u>Power</u>. Each entity in the Company Group has all requisite corporate, partnership or limited liability company power and authority to own, operate and lease its

properties and to carry on its business as and where such is now being conducted. Each entity in the Company Group has full power, legal right and authority to enter into, execute and deliver this Agreement (if a party thereto) and the Ancillary Instruments (if a party thereto), and, as applicable, to carry out the Transactions.

- (c) <u>Authorization</u>. The execution and delivery of this Agreement by each entity in the Company Group (if a party hereto) and the Ancillary Instruments (if a party thereto) and the full performance of its obligations hereunder and thereunder, have been duly authorized by all requisite corporate, partnership or limited liability company action of such entity.
- (d) <u>Validity</u>. This Agreement has been duly and validly executed and delivered by each entity in the Company Group (if a party hereto) and is, and when executed and delivered by each entity in the Company Group (that is or will be a party thereto) each Ancillary Instrument will be, the legal, valid and binding obligation of such entity in the Company Group, enforceable in accordance with its terms, except as such may be limited by bankruptcy, insolvency, reorganization or other laws affecting creditors' rights generally, and by general equitable principles.
- (e) <u>Qualification</u>. Each entity in the Company Group is duly licensed or qualified to do business as a foreign corporation, limited partnership or limited liability company, and is in good standing, in each jurisdiction wherein the character of the properties owned or leased by it, or the nature of its business, makes such licensing or qualification necessary. The jurisdictions in which each entity in the Company Group is licensed or qualified to do business are listed on <u>Schedule 8.1(e)</u>.
- (f) <u>Capitalization of the Companies</u>. The Membership Interests are the only issued and outstanding limited liability company interests, securities or other equity interests, of any class or nature, of the Companies. The Memberships Interests are owned of record and beneficially by the Sellers. All such Membership Interests are validly issued, fully paid and nonassessable. There are no: (i) securities or other interests convertible into or exchangeable for either Membership Interests or other equity ownership interests of the Companies, (ii) options, warrants or other rights to purchase or subscribe to Membership Interests or other equity interests of the Companies or securities which are convertible into or exchangeable for securities or other equity interests of the Companies, or (iii) contracts, commitments, agreements, understandings or arrangements of any kind relating to the issuance or sale of any limited liability company interests or other securities or equity interests of any of the Companies to any Person other than Sellers, or (iv) except as set forth in their respective Organizational Documents, contracts, commitments, agreements, understandings or arrangements of any kind relating to the voting or transfer of any limited liability company interests. Except as set forth in <u>Schedule 8.1(f)</u>, there have never been any members or other owners of the Companies except the Sellers. No Person other than Sellers has the right to become a stockholder, partner, owner or member of any of the Companies.
- (g) <u>Subsidiaries</u>. <u>Schedule 8.1(g)</u> sets forth the name, type of entity, jurisdiction of organization and ownership of each entity in which any Company has a direct or indirect equity interest ("**Subsidiary**") and the jurisdictions in which each Subsidiary is qualified or licensed to do business as a foreign entity. Except as listed in <u>Schedule 8.1(g)</u>, neither

Company owns, directly or indirectly, any equity securities of any entity or has any direct or indirect equity or other ownership interest in any entity or business. Except as set forth in Schedule 8.1(g), each Company, directly or indirectly, owns all the securities and other equity interests of all classes and types of its Subsidiaries and such securities and equity interests are free and clear of any Lien, security interest, restriction, option, voting trust or agreement, proxy or encumbrance and are validly issued, fully paid and nonassessable. Schedule 8.1(g) sets forth the names and percentage interests in capital and profits of each Person who owns an interest in Edgewater Retail Partners, LLC or US Retail Limited Partnership. There are no (i) securities convertible into or exchangeable for the equity interests or other securities of any Subsidiary; (ii) options, warrants or other rights to purchase or subscribe to equity interests or other securities of any Subsidiary or securities which are convertible into or exchangeable for equity interests or other securities of any Subsidiary; (iii) contracts, commitments, agreements, understandings or arrangements of any kind relating to the voting, issuance, sale or transfer to any Person other than Sellers of any securities or other equity interests or other securities of any Subsidiary, any such convertible or exchangeable securities or any such options, warrants or other rights; or (iv) except as set forth in their respective Organizational Documents made available to Buyer, contracts, commitments, agreements, understandings or arrangements of any kind relating to the voting or transfer of any securities or other equity interests of the Subsidiaries. No Person other than another entity in the Company Group has a right to become a stockholder, partner or member of any entity in the Company Group. Each Subsidiary (1) is a general or limited partnership, limited liability company or company duly organized, validly existing and in good standing under the laws of its state of organization, (2) has full general or limited partnership, limited liability company or corporate power (as applicable) and authority to carry on its business as it is now being conducted and to own and lease the properties and assets it now owns and leases, and (3) is in good standing and is duly qualified or licensed to do business as a foreign entity in each of the jurisdictions in which such Subsidiary is required to be so qualified or licensed. There are no options, warrants or other rights to purchase or subscribe to the equity interests of the Subsidiaries or securities which are convertible into or exchangeable for securities or other equity interests of the Subsidiaries. No entity in the Company Group has succeeded to the liabilities of any other Person by operation of law pursuant to a purchase of assets, securities or other equity interests or stock, merger, consolidation or similar transaction.

(h) <u>Organizational Documents, etc.</u> Sellers have made available to Buyer true, correct and complete copies of the Organizational Documents of the Companies, including any amendments thereto. To the extent within Sellers' possession and control, Sellers have made available to Buyer true, correct and complete copies of the Organizational Documents of the Subsidiaries and Sellers have made available to Buyer true, correct and complete copies of the stockholder, partner, member and ownership records of all entities of the Company Group. <u>Schedule 8.1(h)</u> sets forth a complete list of all Organizational Documents of Edgewater Retail Partners, LLC and US Retail Limited Partnership and all other documents relating to (i) the rights of the members or partners thereof other than any entity in the Company Group (the "**Outside Partners**") and (ii) the obligations of any entity in the Company Group with respect to the Outside Partners, and Sellers have furnished to Buyer true, correct and complete copies thereof. Except as set forth on <u>Schedule 8.1(h)</u>, there are no agreements or understandings between any entity in the Company Group and any Outside Partner with respect to these two non-wholly owned Subsidiaries or the Properties owned by such Subsidiaries.

### 8.2. Sellers.

- (a) Organization. Each Seller is duly organized, validly existing and in good standing under the laws of the state of its organization.
- (b) Power. Each Seller has all requisite power and authority to enter into, execute and deliver this Agreement and the Ancillary Instruments to which it is or shall be a party and to carry out the Transactions.
- (c) <u>Authorization</u>. The execution and delivery of this Agreement and the Ancillary Instruments to which any Seller is or shall be a party and the full performance of the obligations of such Seller hereunder and, if applicable, thereunder, have been duly authorized by all requisite legal approval of such Seller and any other necessary approval or authorization.
- (d) <u>Validity</u>. This Agreement has been duly and validly executed and delivered by each Seller and is, the legal, valid and binding obligation of such Seller, enforceable in accordance with its terms, except as such may be limited by bankruptcy, insolvency, reorganization or other laws affecting creditors' rights generally, and by general equitable principles. Each Ancillary Instrument to which each Seller is a party or shall be a party when duly and validly executed and delivered by each Seller will be the legal, valid and binding obligation of such Seller, enforceable in accordance with its terms, except as such may be limited by bankruptcy, insolvency, reorganization or other laws affecting creditors' rights generally, and by general equitable principles.
- (e) <u>Title</u>. Each Seller has record and beneficial ownership of and shall convey to Buyer at Closing the Membership Interests to be sold by such Seller hereunder, free and clear of all Liens, including voting trusts or agreements, proxies, marital or community property interests, rights of first refusal, rights of first offer or options to purchase (except transfer restrictions set forth in the Organizational Documents of the Companies).
- **8.3.** Ownership of Properties. An entity in the Company Group owns title to (or leases to the extent shown on Schedule 8.3) each of the Properties and Exhibit B correctly identifies which entity in the Company Group owns each Property. Neither the Company Group nor any Affiliate of any Selling Party owns or has any interest in any real property directly abutting or adjoining or in the immediate vicinity of any Property. For purposes hereof, the phrase "immediate vicinity of any Property" means any real property which would be deemed to be a material or integral part of the overall shopping center development on the Property.
- **8.4.** No Encumbrances. Except as otherwise contemplated by this Agreement, the Membership Interests, the equity interests of all the Subsidiaries and the material non-real estate property and assets they purport to own, as reflected on the Recent Balance Sheet are free and clear of all Liens and Encumbrances that will remain outstanding after the Closing Date. There are no outstanding judgments against any entity in the Company Group and none of their material assets which are subject to any judgment liens.
- **8.5. No Violation**. Except as set forth on Schedule 8.5 or otherwise contemplated by or provided for in this Agreement, neither the execution and delivery of this Agreement or the Ancillary Instruments nor the consummation by Selling Parties of the Transactions (a) will

violate any Law or Order, (b) will require any authorization, consent, approval, exemption or other action by or notice to any Governmental Authority, or (c) will constitute a default (or an event which, with notice or lapse of time, or both, would constitute a default) under, or will result in the termination of, or accelerate the performance required by, or result in the creation of an additional third party right or any Lien upon any of the assets of the Sellers or the Company Group (or the Membership Interests) under any term or provision of the Organizational Documents of the Sellers or the Company Group.

**8.6.** Financial Statements. Included as Schedule 8.6 are true and complete copies of (a) the consolidated financial statements of the Company Group (or separate consolidated financial statements for each Company and its Subsidiaries) consisting of balance sheets as of December 31, 2003 and the related statements of income and cash flows for the years then ended (including the notes contained therein or annexed thereto), which financial statements have been reported on, and are accompanied by, the signed opinions of PricewaterhouseCoopers LLP, independent auditors for the Company Group for such year, and (b) the unaudited consolidated financial statements of the Company Group (or separate consolidated financial statements of each Company and its Subsidiaries), consisting of an unaudited balance sheet of the Company Group as of December 31, 2004 (the "Recent Balance Sheet"), and the related unaudited statements of income and cash flows for the twelve (12) months then ended and for the corresponding period of the prior year (including any notes and schedules contained therein or annexed thereto). Such financial statements and notes fairly present the financial condition and the results of operations and cash flow of the Company Group as at the respective dates of and for the periods referred to in such financial statements, all in accordance with GAAP, subject in the case of the unaudited financial statements, to normal year-end adjustments and the absence of notes (that, if presented, would not differ materially from those included in the audited financial statements). Sellers acknowledge and agree that Buyer intends to execute and deliver an engagement letter with PricewaterhouseCoopers LLP, pursuant to which PricewaterhouseCoopers, LLP shall provide to Buyer, at Buyer's expense, a certification of such financial statements as Buyer or Investor may reasonably require in connection with filing a Form 8-K or other documents.

**8.7.** Absence of Undisclosed Liabilities. As of the date hereof, the entities in the Company Group have no liabilities or obligations of any nature (whether known or unknown, whether absolute or contingent, whether liquidated or unliquidated and whether due or to become due), except for (i) liabilities stated or adequately reserved against on the Recent Balance Sheet, (ii) liabilities that arose in the ordinary course of business after the date of the Recent Balance Sheet consistent with past practice, (iii) liabilities for brokers, counsel, other consultants and other costs incurred in connection with the Transactions and this Agreement (all of which shall be paid by Sellers), and (iv) liabilities that would not be required to be disclosed on a combined balance sheet of all of the entities in the Company Group or in the footnotes thereto, if such balance sheet were prepared in accordance with GAAP as of the date hereof.

### 8.8. Tax Matters.

(a) <u>Taxes; Tax Returns</u>. Except as set forth on <u>Schedule 8.8(a)</u>: (i) no entity in the Company Group is taxed as a corporation under Section 7701 of the Code or for any state or local tax purposes; (ii) all material Tax Returns required to be filed by or on behalf of any entity in the Company Group have been timely filed; (iii) all material Taxes shown as due on

such Tax Returns or otherwise due and payable by any entity in the Company Group have been timely paid; (iv) all such Tax Returns are true, correct and complete in all material respects; (v) no adjustment or deficiency relating to any of such Tax Returns or otherwise, has been proposed in writing by any Tax authority; (vi) there are no outstanding summons or other document requests with respect to any Tax Returns of the Company Group or the Taxes reflected on such Tax Returns or any other Tax liability imposed on the Company Group; (vii) to the Knowledge of the Selling Parties, there are no pending or threatened actions or proceedings for the assessment or collection of Taxes against the Company Group; (viii) there are no Liens for Taxes on any assets of the Company Group other than liens for Taxes not yet due or payable or which the Company Group is diligently contesting in good faith through appropriate proceedings, as set forth on Schedule 8.8(a); (ix) all material Taxes required to be withheld, collected or deposited by the Company or any of its Subsidiaries, have been timely withheld, collected or deposited and, to the extent required prior to the date hereof, have been paid to the relevant Tax authority; (x) any adjustment of Taxes of the Company Group made by the U.S. Internal Revenue Service that is required to be reported to any state, local or foreign Tax authority has been timely reported; (xi) the Recent Balance Sheet reflects reserves that are adequate for the payment of all material Taxes not yet due and payable that are properly accruable thereon as of the date hereof (including Taxes being contested), and there is no material difference between the amounts of the book basis and the tax basis of assets (net of liabilities) that is not accounted for by an accrual on the Recent Balance Sheet for federal income tax purposes; (xii) no written notice of claim has been made by any authority in a jurisdiction where the Company Group does not file Tax Returns that any of them is or may be subject to taxation by that jurisdiction; and (xiii) no entity in the Company Group has been obligated to report, to disclose or to maintain a list of investors in connection with a "tax shelter" as defined by any Section of the Code. For purposes hereof, "Taxes" means any and all federal, state, local, or foreign income, gross receipts, license, payroll, employment, excise, severance, stamp, occupation, premium, windfall profits, environmental (including taxes under Section 59A of the Code), customs duties, capital stock, franchise, profits, withholding, social security (or similar), unemployment, disability, real property, personal property, sales, use, transfer, registration, value added, alternative or add-on minimum, estimated, or other tax of any kind whatsoever, including any interest, penalty, or addition thereto, whether disputed or not and "Tax Return" means any return, declaration, report, claim for refund or information return or statement relating to Taxes, including any schedule or attachment thereto, and including any amendment thereof, required to be filed with the Internal Revenue Service or any other governmental body or tax authority or agency, whether domestic or foreign, including any consolidated, combined or unitary tax return.

(b) Extension of Statute of Limitations; Rulings. There are no outstanding waivers or agreements extending the statute of limitations for any period with respect to any Tax to which the Company Group may be subject; (ii) no entity in the Company Group has participated in or cooperated with an international boycott within the meaning of Section 999 of the Code; (iii) no entity in the Company Group has had any income reportable for a period ending after the date hereof but attributable to a transaction (e.g., an installment sale) which resulted in a deferred reporting of income from such transaction or from such change in accounting method; (iv) the Company Group has made no written requests for rulings that are currently outstanding that could affect the Taxes of the Company Group or any similar matters pending with respect to any Tax authority; and (v) no entity in the Company Group has ever made an election under Section 1017 of the Code.

- (c) <u>Tax Audits</u>. The federal and state income Tax Returns of the Company Group have not been audited by the Internal Revenue Service and appropriate state taxing authorities, and the Company Group has not received from the Internal Revenue Service or from the tax authorities of any state, county, local or other jurisdiction any notice of underpayment of Taxes or other deficiency which has not been paid nor any objection to any return or report filed by the Company Group. There are outstanding no agreements or waivers extending the statutory period of limitations applicable to any Tax Return.
- (d) <u>Partnership Status</u>. Except as set forth on <u>Schedule 8.8(d)</u>, each entity in the Company Group is qualified, and since the date of its formation has been qualified, to be treated as a partnership or disregarded entity for federal, state and local income tax purposes.
- (e) <u>Foreign Person</u>. None of the Selling Parties are (a) foreign persons within the meaning of Section 1445(1) of the Code, nor (b) disregarded entities under Section 1445-2(b)(2)(iii) of the Code and each Selling Party agrees to execute any and all documents necessary or required by the Internal Revenue Service or Buyer in connection with such declaration(s).
- (f) <u>Section 754 Elections</u>. There are in effect, or Buyer will after the Closing have the power and authority to make elections, under Code Section 754 for U.S. Retail Partners Limited Partnership and Edgewater Retail Partners, LLC to adjust the basis of partnership property pursuant to Code Sections 754 and 743(b), or comparable provisions of state or local Law, in connection with direct or indirect transfers of interests in such entities hereunder.
- **8.9.** <u>Litigation</u>. Except for the litigations disclosed on <u>Schedule 8.9</u> hereto and other litigations where the amount of the claim is less than \$10,000 or where the claim is either fully covered by insurance or where the claim is covered by insurance subject to a deductible, but where Sellers have agreed to be responsible for the deductible, there is no action, suit or proceeding pending or, to the Knowledge of the Company Group threatened against the Sellers, the Companies, the Subsidiaries or any Property in any court or before or by any Governmental Authority. No such litigation would, individually or in the aggregate, materially adversely affect any Selling Party's ability to perform its obligations under the Transaction Documents.
- **8.10.** No Employees. At no time has any entity in the Company Group (other than First Washington Management, Inc.) had any employees, nor has there been any "defined benefit plan" within the meaning of the Employee Retirement Income Security Act of 1974, as amended ("ERISA"), maintained by any entity in the Company Group or any predecessor thereto or any person, firm or corporation which is under "common control (within the meaning of Section 7001(b) of ERISA) with any entity in the Company Group.
- **8.11.** No Condemnation. As of the date hereof, except as set forth in Schedule 8.11, (a) there are no pending or, to the Knowledge of the Company Group, threatened condemnation, expropriation, requisition or similar proceedings against any Property or any portion thereof and (b) no Selling Party or Company has received written notice that any such proceeding is contemplated.

- **8.12.** Environmental Matters. To the Knowledge of the Company Group, the Selling Parties have made available to Buyer in First Washington's office all environmental studies, investigations, reports, audits, assessments, licenses, permits and agreements relating to each of the Properties' compliance or noncompliance with Environmental Laws within the Selling Parties' and Company Group's possession or control (collectively, "Existing Environmental Reports"). The Selling Parties have made available to Buyer in First Washington's office complete copies of the most recent Phase I environmental report for each Property and any Phase II reports obtained as a result of conditions noted in such Phase I report. Except for the Ongoing Remediation and as set forth in the Existing Environmental Reports made available to Buyer in First Washington's office, no Seller nor any Company Group has made any report or disclosure to any Governmental Authority relating to a release or threatened release of Hazardous Substances to or from any Property, in each case except to the extent such matters would not reasonably be expected to have individually or in the aggregate a material adverse effect on the value of any individual Property. Except as set forth in the Existing Environmental Reports, Sellers have no Knowledge of any underground storage tanks located on any Property.
- **8.13.** <u>Leasing Commissions and Tenant Allowances</u>. Except for leasing commission and tenant improvement allowances incurred prior to the date hereof (for which Sellers are responsible) or incurred in connection with Approved New Leases or Lease modifications entered into pursuant to Section 7.3(d) (Tenant Leases), there are no rental, lease, or other commission now due and payable or which will become due or payable with respect to the current term of any of the Leases or in connection with any renewals or extensions of such terms, and no unpaid or pending Tenant improvement allowances or other concessions now due or payable in connection with any of the Leases or which will become due or payable.
- **8.14.** Rent Rolls; Leases. The information set forth in the rent rolls described in Schedule 8.14 (the "Rent Rolls") is accurate as of the date set forth therein except for the information related to pass-through expenses to Tenants, which is accurate as of the date set forth in the separate spreadsheet thereof also described in Schedule 8.14. There are no Leases in force for the Properties other than as identified in the Rent Rolls. The Leases and the Guaranties made available to Buyer are true, correct and complete copies of all the existing Leases and Guaranties. Except as indicated in Schedule 8.14, no Tenant has paid any rent in advance except for the current month and no Tenant is entitled to any concession, rebate, abatement or other similar benefit. There are no written (or binding oral) commitments between the Selling Parties or any entity in the Company Group and any Tenant other than as set forth in such copies. To the Knowledge of Selling Parties, except as indicated in Schedule 8.14, neither the Selling Parties nor any entity in the Company Group have delivered or received any written notice of default or termination under any Lease with a Major Tenant within the six (6) month period prior to the date hereof other than notices of alleged defaults that have been cured.
- **8.15.** Accounts Receivable Aging. The accounts receivable aging analysis dated as of January 12, 2005 attached as Schedule 8.15 contains a complete and accurate list of all accounts receivable for Tenants and accurately sets forth the aging of such accounts receivable as of such date.
- **8.16.** Service Contracts. Schedule 8.16 sets forth a complete list of all maintenance and service agreements relating to the management, maintenance, and operation of the Properties

other than contracts which can be terminated within thirty (30) days without penalty or premium (collectively, the "Service Contracts"), and the copies of such Service Contracts made available to Buyer are true, correct and complete copies of all such Service Contracts. Except as set forth in Schedule 8.16 for the six (6) month period of time prior to the date hereof, neither the Selling Parties nor any entity in the Company Group have delivered or received any written notice of default or termination under any Service Contract which remains uncured or has not been rescinded and, to the Knowledge of Selling Parties, no event has occurred which, with the giving of notice or the passing of time or both, would constitute a default under any Service Contracts. Except as set forth on Schedule 8.16, all Service Contracts were entered into in the ordinary course of business with entities which are not Affiliates of Sellers.

- **8.17.** Material Contracts. Schedule 8.17 sets forth a complete list of all warranties and contracts for indemnification by any entity in the Company Group arising in connection with the sale or disposition of assets, all covenants not to compete binding upon any entity in the Company Group, all guaranties of third party obligations by any entity in the Company Group and all material contracts or agreements not terminable on thirty (30) days' notice without penalty or premium to which any entity in the Company Group is a party (other than Existing Loans, Leases, Ground Leases and Services Contracts), including material government contracts (collectively, the "Material Contracts"), and the copies of such Material Contracts made available to Buyer are true, correct and complete copies of all such Material Contracts. Except as set forth in Schedule 8.17, through the six (6) month period of time prior to the date hereof, neither the Selling Parties nor any entity in the Company Group have delivered or received any written notice of default or termination under any Material Contract which remains uncured or has not been rescinded and, to the Knowledge of Selling Parties, no event has occurred which, with the giving of notice or the passing of time or both, would constitute a default under any Material Contract. Except as set forth in Schedule 8.17, all Material Contracts were entered into in the ordinary course of business with entities which are not Affiliates of Sellers.
- **8.18.** Existing Loans. Schedule 8.18 sets forth a complete list of all loans affecting the Properties and all other loans to, and debt for borrowed money of, each entity in the Company Group (the "Existing Loans") and provides (a) the names of the original lender and current holder (to the extent that the Company Group has received a written notice of the assignment thereof); (b) outstanding principal balances as of December 31, 2004; and (c) the current monthly payments as of January 31, 2005. The information contained in Schedule 8.18 is complete to the extent provided above and accurate in all respects, and the copies of the loan documents evidencing the Existing Loans made available to Buyer are true, correct and complete of all such documents evidencing and securing such Existing Loans. Except as set forth in Schedule 8.18, neither the Selling Parties nor any entity in the Company Group have delivered or received any written notice of any uncurred default or breach under the Existing Loans, and to the Knowledge of Selling Parties, no event has occurred which, with the giving of notice or the passing of time or both, would constitute a default under any of the Existing Loans.
- **8.19.** Ground Leases. Attached hereto as Schedule 8.19 is a list of all Ground Leases. The copies of the Ground Leases made available to Buyer are true, correct and complete copies of all such Ground Leases. Except as set forth in Schedule 8.19, through the date hereof, neither the Selling Parties nor any entity in the Company Group have delivered or received any written notice of default or termination under any Ground Lease which remains uncured. Except as set forth in Schedule 8.19, the Transactions do not require any consent or approval of or any notice to any ground lessor or lender of any ground lessor.

- **8.20.** No Other Contracts. Except as set forth in Schedule 8.20, neither the Sellers nor any entity in the Company Group has entered into any contract for the sale of the Properties, or any of them, that is outstanding nor do there exist any outstanding rights of first refusal, rights of first offer or options to purchase any of the Properties or any portion thereof.
- **8.21.** No Brokers or Finders. No Seller or entity in the Company Group, nor any of their directors, officers, members, partners, principals, shareholders, representatives or agents have retained, employed or used or made any oral or written promise to any broker or finder other than Secured Capital, LLC in connection with the transactions provided for herein or in connection with the negotiation thereof ("Sellers' Broker Rep"). Sellers shall pay any and all fees payable to Secured Capital, LLC in connection with the Transactions.
- **8.22.** <u>Insurance</u>. The Sellers, each entity in the Company Group and the Properties are currently insured with the coverage, in the amounts and with the deductibles, shown in <u>Schedule 8.22</u> hereof. Neither Sellers nor the Company Group have received any notice within the prior six (6) month period of time from any insurer declining to provide insurance coverage for any Property.
- **8.23.** Compliance with Restrictions. Neither the Sellers nor any entity in the Company Group has received any written notice within the six (6) months prior to the date hereof that the existing improvements or current use, operation or parking at any Property does not materially comply with any Law, restriction, declaration, covenant or similar agreement affecting such Property which remains uncured. To the Knowledge of the Sellers the current use and operation and the parking at the Properties comply in all material respects with all Laws, restrictions, declaration, covenants, and similar agreements affecting the Properties, except for such noncompliance that, individually or in the aggregate, would not reasonably be expected to have a material adverse effect on the use and operation or value of any of the individual Properties.
- **8.24. FIRPTA**. None of the Sellers is a "foreign person" or "disregarded entity" within the meaning of the Code. The taxpayer identification number of each Company Group is listed in <u>Schedule 8.24</u>.
- **8.25.** <u>Permits.</u> Neither the Sellers nor any entity in the Company Group has received within the last six (6) months any written notice from any Governmental Entity that any of the Properties are not in substantial compliance with any material governmental permit, license or certificate or any agreement, easement or similar instrument applicable to any of the Properties which has not been fully cured.
- **8.26.** Existing Policies. Each Existing Policy is, as of the date hereof, in full force and effect and no material claim has been made by the insured under any such policy.
- **8.27.** Shadow Anchors. Schedule 8.27 is a list of each Property identified in the Selling Parties' October 2004 National Portfolio Executive Summary as having a "shadow anchor," and the identity of each such "shadow anchor." Sellers have made available to Buyer

true, correct and complete copies of each agreement between any "shadow anchor" and any entity in the Company Group (collectively, the "**Shadow Anchor Agreements**"). To the Knowledge of Jeffrey S. Distenfeld, except as set forth on <u>Schedule 8.27</u> through the date hereof, neither the Selling Parties nor any entity in the Company Group have delivered or received any written notice of default or event of default under, any Shadow Anchor Agreement.

## 9. REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer represents and warrants to the Sellers as follows:

### 9.1. Organization; Authorization.

- (a) <u>Organization</u>. Buyer is a limited liability company, duly organized, validly existing and in good standing under the laws of the state of its organization.
- (b) <u>Power</u>. Buyer has all requisite limited liability company power and authority to enter into, execute and deliver this Agreement and the Ancillary Instruments to which Buyer is or shall be a party, and to carry out the Transactions.
- (c) <u>Authorization</u>. The execution and delivery of this Agreement and the Ancillary Instruments to which Buyer is or shall be a party, and full performance thereunder, have been duly authorized by the members of the Buyer and no other or further limited liability company act on the part of the Buyer is necessary therefore.
- (d) <u>Validity</u>. This Agreement has been duly and validly executed and delivered by Buyer and is, and when executed and delivered each Ancillary Instrument to which Buyer is or shall be a party when duly and validly executed and delivered by each Seller, will be, the legal, valid and binding obligation of Buyer, enforceable in accordance with its terms, except as such may be limited by bankruptcy, insolvency, reorganization or other laws affecting creditors' rights generally, and by general equitable principles.
- **9.2.** <u>Bankruptcy of Buyer</u>. Buyer has not made a general assignment for the benefit of creditors, filed any voluntary petition in bankruptcy or suffered the filing of an involuntary petition by Buyer's creditors, suffered the appointment of a receiver to take possession of all, or substantially all, of Buyer's assets, suffered the attachment or other judicial seizure of all, or substantially all, of Buyer's assets, admitted in writing its inability to pay its debts as they come due or made an offer of settlement, extension or composition to its creditors generally.
- **9.3.** <u>No Brokers or Finders</u>. Neither Buyer nor any of its members, partners, directors, officers, employees, principals, shareholders, representatives or agents have retained, employed or used, or made any oral or written promises to, any broker, finder or like agent other than Macquarie Capital Partners, LLC ( "Broker") in connection with the Transactions or in connection with the negotiation thereof ("Buyer's Broker Rep"). Buyer shall pay any and all fees payable to Broker in connection with the Transactions.
  - 9.4. <u>Investment Intent</u>. The Membership Interests are being acquired by Buyer for investment only and not with the view to resale or other distribution.

### 10. REPRESENTATIONS AND WARRANTIES OF INVESTORS

Each Investor, severally, solely with respect to such Investor, represents and warrants to Sellers as follows:

- **10.1. Power**. Such Investor has all requisite power and authority to enter into, execute and deliver this Agreement and to carry out the Transactions.
- **10.2.** <u>Authorization</u>. The execution and delivery of this Agreement has been duly authorized by the respective boards of directors or trustees of such Investor, and no other or further corporate or trust act on the part of such Investor is necessary therefor.
- **10.3.** <u>Validity</u>. This Agreement has been duly and validly executed and delivered by each Investor and is the legal, valid and binding obligation of such Investor, enforceable in accordance with its terms, except as such may be limited by bankruptcy, insolvency, reorganization or other laws affecting creditors' rights generally, and by general equitable principles.
- **10.4.** <u>Financial Ability</u>. Such Investor has the financial resources or reasonably and in good faith believes it can obtain the financial resources, including binding commitments from its prospective lenders, for its pro rata portion of the Purchase Price, necessary to consummate the Transactions. MCW has provided to Sellers a copy of the form of underwriting agreement described in Section 11.2(c).

#### 11. CONDITIONS TO DEPOSIT

- **11.1.** <u>Conditions to Obligations of Sellers at Deposit Date</u>. The obligations of Sellers under this Agreement to accept the Deposit on the Deposit Date shall be subject to the satisfaction of the following conditions on or before the Deposit Date. Any of such conditions may be waived by Sellers in writing on or before the Deposit Date.
- (a) <u>Agreement in Force</u>. This Agreement shall not have been terminated (or if terminated shall have been duly reinstated by Sellers) pursuant to Section 15.2(d) (Sellers' and Buyer's Pre-Deposit Options) or Section 16 (Termination; Default).
- (b) <u>No Injunctions</u>. No temporary restraining order, preliminary or permanent injunction, cease and desist order or other order issued by any Governmental Authority prohibiting the completion of all or a material portion of the Transactions shall be in effect.
- (c) <u>Environmental Insurance Policy</u>. The total cost of Buyer's Environmental Insurance Policy pursuant to Section 5.6(f), without any additional coverage requested by Buyer, does not exceed \$10,000,000.

If, by 3:00 p.m. Eastern time on the Deposit Date, any of the forgoing conditions are not satisfied for any reason whatsoever or, alternatively, are not expressly waived in writing by Sellers, Sellers shall have no obligation to accept the Deposit and shall have the right to terminate this Agreement. Sellers' acceptance of the Deposit shall constitute an irrevocable waiver of all conditions to acceptance of the Deposit set forth in this Section 11.1.

- 11.2. <u>Conditions to Obligations of Buyer in Connection With Deposit</u>. The obligations of Buyer under this Agreement to provide the Deposit on the Deposit Date and to purchase the Membership Interests and consummate the other Transactions shall be subject to the satisfaction of the following conditions on or before the Deposit Date. Any of such conditions may be waived by Buyer in writing on or before the Deposit Date. Buyer's tendering of the Deposit on the Deposit Date shall constitute an irrevocable waiver of all conditions to payment of the Deposit set forth in this Section 11.2.
- (a) <u>Agreement in Force</u>. This Agreement shall not have been terminated (or if terminated shall have been duly reinstated by Seller) pursuant to Section 15.2(d) (Sellers' and Buyer's Pre-Deposit Options) or Section 16 (Termination; Default).
- (b) <u>No Injunctions</u>. No temporary restraining order, preliminary or permanent injunction, cease and desist order or other order issued by any Governmental Authority prohibiting the completion of all or a material portion of the Transactions shall be in effect.
- (c) <u>Underwriting Agreement</u>. MCW's underwriting agreement with Macquarie Equity Capital Markets Limited, Citigroup Global Markets Australia Pty Limited and Merrill Lynch International (Australia) Limited dated on or about the date hereof shall not have been terminated for any of the reasons set forth in <u>Schedule 11.2(c)</u>.
- (d) <u>Tenant Estoppels</u>. Buyer shall have received Tenant estoppel certificates ("**Tenant Estoppels**") addressed to Buyer from each Tenant leasing 10,000 square feet or more and from any national drugstore chain (such as Walgreens, CVS or Rite-Aid) regardless of the actual square feet leased by any such national drugstore chain (collectively, the "**Major Tenants**") plus additional Tenants leasing in the aggregate not less than sixty percent (60%) of the occupied rentable space for all Properties excluding space leased to Major Tenants but in no event less than fifty percent (50%) of the occupied rentable space as to any one Property (the "**Estoppel Threshold**"). The Tenant Estoppels shall be either (i) substantially in the form attached hereto as <u>Exhibit C</u>; (ii) in the form substantially similar to that provided for under the applicable Lease; or (iii) in the form customarily used by an applicable Major Tenant or any other national or regional tenant. The Sellers shall cause the Company Group to deliver to Buyer copies of the signed Tenant Estoppels promptly following the receipt thereof. Buyer shall have the right to disapprove a Tenant Estoppel only if (A) such Tenant Estoppel alleges a set of facts, circumstances, conditions or events relating to the applicable Lease that would constitute a default or event of default or breach of representation or warranty under this Agreement or the Lease; and (B) Buyer delivers written objection notice to the Sellers within ten (10) Business Days following the date of Buyer's receipt thereof (each a "**Non-Conforming Tenant Estoppel**"). Each Tenant Estoppel received by Buyer prior to the Deposit Date that is a Non-Conforming Tenant Estoppel shall not be deemed to be a "**Conforming Tenant Estoppel**" and shall not be included within the determination of whether or not the Estoppel Threshold has been satisfied. Sellers shall deliver the originals of Tenant Estoppels and Ground Lease Estoppels to Buyer within three (3) Business Day(s) after Sellers' receipt to the following a

Regency Centers Corporation 121 Forsyth Street, Suite 200 Jacksonville, Florida 32202 Attn: David A. McNulty

Tel.: (904) 598-7000 Fax: (904) 354-6094 E-mail: DMcnulty@regencycenters.com

If Sellers fail to deliver conforming Tenant Estoppels sufficient to achieve the Estoppel Threshold by no later than five (5) Business Days prior to the Deposit Date, Buyer may elect, by notice delivered to Sellers no later than three (3) Business Days prior the Deposit Date, to either (i) waive the Estoppel Threshold as a condition to Buyer's obligation to deliver the Deposit or (ii) require that Sellers deliver statements (the "Seller Estoppel") in the form attached hereto as Exhibit D with respect to such Leases as are required to satisfy the Estoppel Threshold. It is hereby expressly acknowledged and agreed that except as set forth immediately below in this Section 11.2(d): (A) Seller Estoppels are intended to and shall be identical in scope and consequence to Sellers representations and warranties set forth in Sections 8.13 (Leasing Commissions and Tenant Allowances), 8.14 (Rent Rolls; Leases) and 8.15 (Accounts Receivable Aging) hereof and (B) in the event that Buyer receives a Conforming Tenant Estoppel after the Deposit Date and for which the Sellers previously delivered a Seller Estoppel, the Sellers shall be automatically released from any liability or obligation under the applicable Seller Estoppel.

In the event Buyer delivers a notice to Sellers not less than five (5) Business Days prior to Closing Date stating that: (i) Buyer has received Non-Conforming Tenant Estoppels from any Major Tenant(s) for which Sellers delivered a Seller Estoppel or from any Tenant which is not a Major Tenant and (ii) Buyer alleges aggregate Losses that would constitute Breach Events arising out of facts, circumstances, conditions or events disclosed in any Non-Conforming Tenant Estoppel(s) from any Major Tenant(s) at any Property in excess of ten percent (10%) of the Agreed Value of the applicable Property (a "Non-Conforming Property"); and (iii) Property-Related Adjustment Losses (excluding any Losses attributable to Breach Event(s) arising from the representations and warranties of Sellers in Section 8.13 (Leasing Commissions and Tenant Allowances), 8.14 (Rent Rolls; Leases) and/or 8.15 (Accounts Receivable Aging), that overlap with the applicable facts, circumstances, conditions or events alleged in such Non-Conforming Tenant Estoppel(s) relating to the applicable Non-Conforming Property that Buyer has raised in any other Buyer Breach Notice) exceed \$18,442,800, then Sellers shall, with respect to each Non-Conforming Property, elect to either (A) give Buyer a credit against the Purchase Price equal to the amount of such Losses arising out of the facts, circumstances, conditions or events disclosed in such Non-Conforming Tenant Estoppel(s) from the Major Tenant(s) of a Non-Conforming Property or (B) exclude such Non-Conforming Property from the Transactions as provided for in Section 2.13 hereof and such Property shall be deemed to be a Dropped Asset.

All disputes concerning whether a Tenant Estoppel is a Non-Conforming Tenant Estoppel and the determination of Losses arising from any applicable facts, circumstances, conditions or events shall be subject to Baseball Arbitration.

(e) <u>Ground Lease Consents and Estoppels</u>. Buyer shall have received estoppel certificates from each ground lessor, which estoppel certificates shall include any

required consents to the Transactions, as applicable (the "**Ground Lease Estoppels**") addressed to Buyer. Each Ground Lease Estoppel shall either: (i) be in the form attached to the applicable Ground Lease; (ii) contain the information required to be provided by lessor under any specific Ground Lease if no form of estoppel certificate is stipulated; or (iii) shall not reflect any uncured default and shall not reflect information that is inconsistent with applicable information regarding the applicable Ground Lease provided for in Section 8.19. Buyer shall have the right to disapprove a Ground Lease Estoppel only if (A) such Ground Lease Estoppel alleges a set of facts, circumstances, conditions or events relating to the applicable Ground Lease that could constitute a default or breach of representation or warranty under this Agreement or the Ground Lease or reflect information inconsistent with the information provided in Section 8.19 and (B) Buyer delivers written objection notice to the Sellers within ten (10) Business Days following the date of Buyer's receipt thereof (each a "**Non-Conforming Ground Lease Estoppel**"). Each Ground Lessor Estoppel received prior to the Deposit Date by Buyer that is not a Non-Conforming Ground Lease Estoppel shall be a "**Conforming Ground Lease Estoppel**". Sellers shall deliver to Buyer, at the address set forth in subsection (c) hereof originals of the signed Ground Lease Estoppels promptly following receipt thereof.

If by no later than five (5) Business Days prior to the Deposit Date, Buyer has not received Conforming Ground Lease Estoppels from all Ground Lessors, Buyer shall deliver a notice to Sellers by no later than three (3) Business Days prior to the Deposit Date electing to either (i) waive the requirement for Conforming Ground Lease Estoppels set forth in this Section 11.2(e) as a condition to Buyer's obligation to deliver the Deposit or (ii) require that Sellers deliver statements (the "Seller GL Estoppel") in the form attached hereto as Exhibit E with respect to any the Ground Lease for which a Conforming Ground Lease Estoppel has not been received for a Ground Lease. It is hereby expressly acknowledged and agreed that except as set forth immediately below in this Section 11.2(e), (A) Seller GL Estoppels are intended to and shall be identical in scope and consequence to Sellers representations and warranties set forth in Section 8.19 (Ground Leases) and (B) in the event that Buyer receives a Conforming Ground Lease Estoppel after the Deposit Date and for which the Sellers previously delivered a Seller GL Estoppel, the Sellers shall be automatically released from any liability or obligation under the applicable Seller GL Estoppel.

If the Title Company issuing a Title Commitment with respect to a Property takes an exception based on the validity of the ground lessee's estate, then Sellers shall indemnify Buyer against any Loss arising directly out of such exception (which indemnification obligation shall be limited to the Agreed Value for the Property) or in lieu of Sellers' indemnity, Buyer may elect to exclude such Property or affected ground leased portion thereof from the Transactions as provided for in Section 2.13 (Dropping Properties) hereof and such Property or ground leased portion thereof shall be deemed a Dropped Asset.

(f) <u>Title Insurance Commitments</u>. The Title Company shall have issued a signed Title Commitment for each Property dated as of the Deposit Date obligating the Title Company to issue at Closing, upon payment of the applicable premium therefor (at standard premium rates or less) and performance by the parties of their respective obligations under the Transaction Documents, in the name of a party designated by the Buyer the Owner's Title Policies, with deletion of general survey exceptions (and the specific survey exceptions brought current to the Closing Date), insuring the party designated by the Buyer as owner of good,

marketable and indefeasible fee simple (or leasehold, to the extent indicated in <u>Schedule 8.3</u>) title to each Property, subject only to the Permitted Exceptions and the resolution of Other Exceptions and Mandatory Removal Exceptions in accordance with the provisions of Sections 5.1(b)(v), 5.2 and 5.3.

(g) <u>Environmental Insurance Policy</u>. The Insurer is prepared to issue a binding commitment to issue Buyer's Environmental Insurance Policy pursuant to Section 5.6(f), without any additional coverage requested by Buyer, for a total cost not to exceed \$10,000,000.

If, by the Deposit Date, any of the foregoing conditions are not satisfied for any reason whatsoever or, alternatively, is not expressly waived by Buyer in writing, Buyer may terminate this Agreement by written notice to Sellers.

### 2. DELIVERIES AT DEPOSIT DATE

# 12.1. Deliveries by the Selling Parties at Time of Deposit.

- (a) The Selling Parties shall execute (or cause to be executed) and deliver (or cause to be delivered) to Buyer at the Deposit Date, simultaneously with and as a condition to MCW's payment of the Deposit, the following certificate, other information and documents. Notwithstanding the foregoing, it is hereby expressly acknowledged and agreed that any certificate required to be delivered by Sellers pursuant to Section 12.1(a)(i) and (iii), are not intended to and shall not in any way affect the respective rights and obligations of the parties under this Agreement (including expanding or modifying any representation, warranty or covenant set forth in this Agreement) and shall not relieve Sellers for liability, if any, for any Breach Event:
- (i) A certificate executed by Selling Parties dated as of the Deposit Date (A) reaffirming that (1) the Sellers' representations and warranties were true and correct as of the date of this Agreement and, (2) the Sellers' representations and warranties in Section 8, other than those referred to in Section 15.2(a)(vi) (Property Breach), are true and correct as of the date of the Deposit Date, or, if Sellers have Knowledge that any representations and warranties are inaccurate in any material respect, the matters which make them inaccurate; (B) stating that Selling Parties have fully performed all the covenants contained in this Agreement to be performed by Selling Parties up to such date, or, if Sellers have Knowledge that Sellers have not performed all such covenants in any material respect, describing such non-performance.
- (ii) If and to the extent required, evidence reasonably satisfactory to the Title Company of the legal existence and good standing of Selling Parties and the power and authority of the individual(s) executing and delivering this Agreement and the instruments and certificates described herein on behalf of Sellers to act for and bind Sellers;
- (iii) A rent roll for each Property, dated as of the date of the Rent Roll, certified by Sellers to be complete and accurate in all material respects, as of the date of the Rent Roll;
- (iv) A written opinion of Hogan & Hartson, LLP, counsel to the Sellers, dated as of the Deposit Date, addressed to Buyer, in customary form reasonably

acceptable to the parties addressing, with respect to the Sellers (i) organization; (ii) power; (iii) due authorization; (iv) execution and delivery; (v) no conflict with or default under Organizational Documents; and (vi) enforceability. Hogan & Hartson LLP may reasonably rely upon the opinion of California counsel as to California Public Employees' Retirement System and the opinion of First Washington's counsel as to First Washington;

- (v) Copies of the Organizational Documents of each such entity (or a parent Person) in the Company Group, certified by the Secretary or other appropriate Person, and copies of the Certificate or Articles of Incorporation, Limited Partnership, Formation or Organization of each entity in the Company Group, certified by the Secretary of State of the state of its organization as of a recent date,
- (vi) Certificates of good standing of First Washington Investment I, LLC and each entity in the Company Group, as of a recent date, from the Secretary of State of the state of its organization and each jurisdiction in which a Property owned by such entity is located;
- (vii) Incumbency certificates relating to each person executing (as an officer or otherwise on behalf of another person) any document executed and delivered to Buyer pursuant to the terms hereof;
  - (viii) A counterpart of the Escrow Agreement;
  - (ix) A counterpart of the Blocked Account Agreement;
- (x) Title Affidavit(s) with respect to each individual Property executed by the applicable Selling Parties and in a form satisfactory to the Title Company, which Title Affidavit(s) shall be released to the Title Company at the Closing;
  - (xi) A binding commitment to issue Buyer's Environmental Insurance Policy to Buyer at Closing.
- (xii) The Title Company shall have issued a signed Title Commitment for each Property dated as of the Deposit Date obligating the Title Company to issue at Closing, upon payment of the applicable premium therefor (at standard premium rates or less) and performance by the parties of their respective obligations under the Transaction Documents, in the name of a party designated by the Buyer the Owner's Title Policies, with deletion of general survey exceptions (and the specific survey exceptions brought current to the Closing Date), insuring the party designated by the Buyer as owner of good, marketable and indefeasible fee simple (or leasehold, to the extent indicated in <a href="Schedule 8.3">Schedule 8.3</a>) title to each Property, subject only to the Permitted Exceptions and the resolution of Other Exceptions and Mandatory Removal Exceptions in accordance with the provisions of Sections 5.1(b)(v), 5.2 and 5.3.
- (xiii) Evidence reasonably satisfactory to the Title Company of the legal existence and good standing of Selling Parties and each entity in the Company Group and the power and authority of the individual(s) executing and delivering this Agreement and the instruments and certificates described herein on behalf of Sellers to act for and bind Sellers; and

(xiv) All other documents, instruments or writings required to be delivered to Buyer or either Investor at or prior to the Deposit Date pursuant to this Agreement and such other certificates of authority and documents as Buyer may reasonably request, which do not expand any liability or obligation of the Sellers.

- (b) <u>Deliveries to Escrow Agent</u>. The Selling Parties shall deliver or cause to be delivered to Escrow Agent pursuant to the Escrow Agreement at the Deposit Date, simultaneously with and as a condition to payment of the Deposit, the following:
  - (i) Certificates representing the Membership Interests accompanied by duly executed transfer power executed by Sellers in favor of Buyer;
  - (ii) Documentation regarding Sellers' resignations as members of the Companies and Buyer's admission as a member of the Companies;
  - (iii) A Management Agreement duly executed by First Washington Realty, Inc., containing the terms set forth on Exhibit F hereto;
  - (iv) The Accounting Services Agreement duly executed by First Washington Realty, Inc., containing the terms set forth on Exhibit F hereto;
  - (v) The resignations of each of the officers, directors and managers of the Company Group;
- (vi) General release, to be effective as of Closing, by Sellers and First Washington Realty, Inc. releasing each entity in the Company Group from all claims arising prior to the Closing Date and containing waivers of any right of contribution or other recourse against the Company Group with respect to representations, warranties and covenants made by the Companies herein; and
  - (vii) A counterpart of the Escrow Agreement.
- **12.2.** <u>Deliveries by Buyer at Time of Deposit</u>. At the Deposit Date, Buyer shall execute (or cause to be executed) and deliver to Sellers, or cause to be delivered to Sellers, the following documents and items:
- (a) A certificate as of the Deposit Date restating and reaffirming that Buyer's representations and warranties set forth in Section 10 are true and correct as of the Deposit Date, or, if there have been changes, describing such changes;
  - (b) A counterpart of the Blocked Account Agreement, duly executed by Buyer;
  - (c) Immediately available funds in the full amount of the Deposit;
- (d) A written opinion of Foley & Lardner LLP, counsel to the Buyer, dated as of the Deposit Date, addressed to the Sellers, in customary form reasonably acceptable to the parties addressing (i) organization; (ii) power; (iii) due authorization; (iv) due execution and delivery; (v) no conflict with Organizational Documents; and (vi) enforceability;

- (e) A certified copy of the resolutions of the members of Buyer authorizing and approving this Agreement and the consummation of the Transactions;
- (f) Incumbency certificates relating to each person executing (as an officer or otherwise on behalf of another person) any document executed and delivered by Buyer pursuant to the terms hereof;
  - (g) A counterpart of the Escrow Agreement; and
- (h) All other documents, instruments or writings required to be delivered to Sellers at or prior to the Deposit Date pursuant to this Agreement and such other certificates of authority and documents as Sellers may reasonably request, which do not expand any liability or obligation of the Buyer or Investors.

## 13. CONDITIONS TO CLOSING

- **13.1.** Conditions to Obligations of Sellers at the Closing. The obligations of Sellers under this Agreement to sell the Membership Interests and consummate the other Transactions shall be subject to the satisfaction, or waiver by Sellers in writing, of all the following conditions on or before the Closing Date:
- (a) <u>No Injunction</u>. No temporary restraining order, preliminary or permanent injunction, cease and desist order or other order issued by any Governmental Authority prohibiting the completion of all or a material portion of the Transactions shall be in effect.

Other than the foregoing, upon payment of the Deposit and delivery by the Buyer of the Purchase Price (as adjusted hereunder) and the other documents required to be delivered by Buyer at the Closing), there are no conditions to Sellers' obligations to close the sale of the Membership Interests and the other Transactions.

- **13.2.** <u>Conditions to Obligations of Buyer at the Closing</u>. The obligations of Buyer to purchase the Membership Interests and consummate the other Transactions shall be subject to the satisfaction, or waiver by the Buyer in writing, of all the following conditions on or before the Closing Date:
- (a) <u>Lender Consents</u>. The Lenders shall have provided (i) the Lender Consents from the holders of all Existing Loans (other than any Existing Loan that can be prepaid as of the Closing Date in accordance with its terms (after giving effect to any applicable prepayment notice periods)), and (ii) the Lender Retention Documents in connection with the Retained Loans and all other Existing Loans may be prepaid in accordance with their terms. If the Closing has not yet occurred on October 1, 2005, Sellers shall cause the Company Group to exclude all Properties listed on Schedule 13.2(a) (or the Subsidiaries owning such Properties, as indicated on Schedule 13.2(a)) for which Lender's Consent has not been obtained pursuant to Section 2.13 (Dropping Properties), in which case such Properties shall be "**Dropped Assets**."

- (b) <u>Value of Dropped Assets</u>. Notwithstanding anything herein to the contrary, the total Agreed Value of all Dropped Assets pursuant to Section 2.13 (other than (1) exclusions at Buyer's request relating to Tenant Estoppels pursuant to Section 11.2(d), (ii) the exclusion of the Properties listed on <u>Schedule 13.2(a)</u> pursuant to Section 13.2(a), (iii) the MPB&T Parcel, and (iv) Edgewater, if it is sold pursuant to the buy-sell (pursuant to Section 2.11)) shall not exceed \$270,000,000.
- (c) <u>No Injunction</u>. No temporary restraining order, preliminary or permanent injunction, cease and desist order or other order issued by any Governmental Authority prohibiting the completion of all or a material portion of the Transactions shall be in effect.

## 14. CLOSING

- 14.1. Deliveries by the Selling Parties at Closing. The Selling Parties shall deliver or cause to be delivered to Buyer at Closing the following:
- (a) The Lender Consents and Lender Retention Documents to be executed by the applicable entities in the Company Group in connection with maintaining the Existing Loans in place after Closing;
- (b) Letter executed by each Lender confirming the outstanding principal balance and accrued but unpaid interest and costs for the Existing Loans to be paid off at Closing;
- (c) A certificate executed by Sellers dated as of the Closing Date (A) reaffirming that (1) the Sellers' representations and warranties were true and correct as of the date of this Agreement and (2) the Sellers' representations and warranties in Section 8, other than those referred to in Section 15.2(a)(vi) (Property Breach), are true and correct as of the date of the Closing, or, if Sellers have Knowledge that the representations and warranties are inaccurate, describing the matters which make then inaccurate. It is hereby expressly acknowledged and agreed that such Seller's certificate is not intended to and shall not in any way increase or decrease the respective rights and obligations of the parties under this Agreement (including expanding or modifying any representation, warranty or covenant set forth in this Agreement) and shall not relieve Sellers from liability for any Breach Events, if any;
- (d) A certificate executed by Selling Parties dated as of the Closing Date that Sellers have fully performed all their covenants contained in this Agreement, or, if Sellers have not performed all their covenants, describing such matters. It is hereby expressly acknowledged and agreed that such Seller's certificate is not intended to and shall not in any way increase or decrease the respective rights and obligations of the parties under this Agreement (including expanding or modifying any representation, warranty or covenant set forth in this Agreement) and shall not relieve Sellers from liability for any Breach Events, if any;
- (e) Title affidavit(s) (the "**Title Affidavits**") with respect to each individual Property executed by the applicable Selling Parties and in a form satisfactory to the Title Company and containing the minimum representations which the Title Company and containing the minimum representations which the Title Company shall require, and sufficient to delete the general survey exception (and sufficient to bring the specific survey exception current to the

Closing Date), mechanic's lien exception and parties in possession exception (and in its place identify only the specific tenants in possession pursuant to Leases for the applicable Property), and such other instruments, such as lien waivers and mechanics' lien and "gap" indemnities, as the Title Company shall reasonably require, in order to issue the Owner's Title Policies in favor of the applicable entity of the Company Group insuring good, marketable and indefeasible fee simple title (or leasehold, to the extent identified in Schedule 8.3) to the applicable Property and the Improvements subject only to the Permitted Exceptions;

- (f) Owner's Title Policies (or marked and signed Title Commitments therefor dated as of the Closing Date), together with all applicable Extended Coverages, each in an amount not less than the Agreed Value for the applicable Property insuring the applicable entity in the Company Group as owner of good, marketable and indefeasible fee simple title (or leasehold, to the extent shown on <a href="Schedule 8.3">Schedule 8.3</a>) to the applicable Property and Improvements subject only to the Permitted Exceptions, effective as of the Closing Date;
- (g) Evidence reasonably satisfactory to the Title Company of the legal existence and good standing of Selling Parties and each entity in the Company Group and the power and authority of the individual(s) executing and delivering this Agreement and the instruments and certificates described herein on behalf of Sellers to act for and bind Sellers;
- (h) A rent roll for each Property, in the same format as the Rent Roll and dated as of the date of the Rent Roll, certified by the applicable Selling Party to be complete and accurate in all material respects;
- (i) To the extent required by law, an affidavit from Sellers concerning compliance with, or exemption from, any applicable withholding or transfer laws of the state in which any Property is located;
- (j) An affidavit from Sellers that each Seller is not a "foreign person" or "disregarded entity," as such term is defined in Section 1445 of the Code, as amended;
- (k) A certificate executed by Selling Parties dated as of the Closing Date confirming that the Company Group Organizational Documents, as in effect on the Deposit Date, have not been modified or amended and remain in full force and effect;
- (l) Certificates of good standing of First Washington Investment I, LLC and each entity in the Company Group, as of a recent date, from the Secretary of State of the state of its organization and each jurisdiction in which a Property owned by such entity is located;
- (m) Incumbency certificates relating to each person executing (as an officer or otherwise on behalf of another person) any document executed and delivered to Buyer at the Closing;
  - (n) Payment of the accrued but unpaid interest on the Deposit;
- (o) A counterpart of the joint instructions with Buyer under the Blocked Account Agreement and the Escrow Agreement to release the items in Escrow and the Deposit ("Joint Instructions");

- (p) The Closing Statement setting forth the source and disposition of the Purchase Price, after Price Adjustments, all prorations and all other funds transferred at Closing;
  - (q) All documents necessary to remove the existing signatories on all bank and securities accounts of the Company Group;
  - (r) The Buyer's Environmental Insurance Policy with evidence of payment in full of all premiums for the 10-year term thereof;
  - (s) A counterpart of the Disputed Escrow Agreement, if applicable.
  - (t) A counterpart of the Ongoing Remediation Escrow Agreement; and
  - (u) A counterpart of the Future Remediation Escrow Agreement.
  - 14.2. Deliveries by the Buyer at Closing. The Buyer, shall deliver or cause to be delivered to Sellers at Closing the following:
    - (a) Immediately available funds in the full amount of the Purchase Price, less the amount of the Deposit, after the Price Adjustments; and
    - (b) A counterpart of the Closing Statement;
    - (c) A counterpart of the Joint Instructions;
    - (d) A counterpart of the Disputed Escrow Agreement, if applicable.
    - (e) A counterpart of the Ongoing Remediation Escrow Agreement; and
    - (f) A counterpart of the Future Remediation Escrow Agreement.
- **14.3.** <u>Deliveries by Escrow Agent at Closing</u>. The parties shall cause the Escrow Agent to deliver or cause to be delivered to Buyer at Closing the documents placed in escrow pursuant to Section 12.1(b) (Deliveries to Escrow Agent).
- 14.4. Other Instruments. In addition to the documents and instruments to be delivered as herein provided, each of the parties hereto shall, from time to time at the request of the other party, execute and deliver to the other party such other instruments of transfer, conveyance and assignment and shall take such other action as may be reasonably required to effectively carry out the terms and conditions of this Agreement, provided that no such additional document or action shall expand any obligation, covenant, representation or warranty of either party or result in any new or additional obligations, covenants, representations or warranties by any party beyond those expressly set forth in this Agreement.

## 15. BREACHES, REMEDIES AND INDEMNIFICATION

## 15.1. Buyer's Indemnity for Buyer's Inspection.

- (a) Buyer shall and does hereby agree from and after the date of this Agreement to indemnify, defend and hold the Selling Parties and the Company Group, their respective members, partners, principals, officers, directors, employees, agents and attorneys, and their respective successors and assigns, harmless from and against any and all loss, cost, damage, liability and expense (including attorneys' fees and costs), whether arising out of injury or death to persons or damage to any property, arising out of Buyer's or Buyer's agents' actions taken in, on or about the Properties in the exercise of the inspection and testing rights granted pursuant to Section 4.1 (Right to Evaluate), caused by: (i) Buyer's or any of its consultants' entry upon the Properties and any due diligence activities pursuant to Section 4.1 including the costs of remediation, restoration and other similar activities, mechanic's and materialmen's liens and attorneys fees, arising out of or in connection with the exercising of Buyer's rights under Section 4.1 (Right to Evaluate); (ii) claims made by any Tenant against any entity in the Company Group for Buyer's or Buyer's agent's, entry into such Tenant's premises or any interference with any Tenant's use or damage to its premises or property; and (iii) material disturbances, disruptions or interference with the on-going business operations of the applicable Selling Parties; provided, however, that Buyer shall have no duty to defend or indemnify any Selling Party or any other person for any loss, cost, damage, liability or expense arising out of conditions merely discovered, but not caused or contributed to, by Buyer or its consultants.
- (b) Notwithstanding anything in this Agreement to the contrary, including Section 15.7 (Limitations on Indemnification), the provisions of this Section 15.1 shall not be subject to any limitation of liability set forth in this Agreement.

# 15.2. Liabilities of Sellers and Remedies of Buyer for Deposit Date or Pre-Closing Breaches.

- (a) <u>Certain Definitions</u>. The following terms have the indicated meanings:
- (i) "Losses" means the adverse economic consequences of any Breach Event or Deferred Maintenance Condition (but without duplication to the extent there are multiple Breach Events or Deferred Maintenance Conditions which arise from, out of or relate to overlapping facts, circumstances, conditions or events) equal to either (A) if the Breach Event is of a type that cannot be cured by performing work or satisfying financial obligations that can be liquidated as to amount ("Fixed Financial Obligations"), the aggregate diminution of value of the particular Property, Properties or entity in the Company Group (as applicable) to the extent directly arising from the Breach Event, including diminution of value by reason of the loss of rent or others items of recurring income (and for purposes of clarification hereof, in no event shall the determination of diminution of value include any consideration of consequential damages, other than to the extent set forth in the following two sentences of this Section 15.2(a)(i), and items such as loss of rent or other items of recurring income arising from the Breach Event) or (B) if the Breach Event or Deferred Maintenance Condition is of a type that can be cured by performing work or satisfying Fixed Financial Obligations, the aggregate reasonable cost of curing such Breach Event or Deferred Maintenance Condition (together with,

in the case of a Breach Event, the loss of rent or other items of recurring income during the cure period for such Breach Event to the extent arising from the Breach Event). To the extent one or more Breach Events relating to Section 8.14 (Rent Rolls; Leases) constitute an overstatement or understatement of Fixed Rent or Additional Rent properly included in calculating recurring net operating income from a Property, the calculation of Losses would include multiplying the net overstatement or understatement of such base rent or recoveries by a factor of 14.918 (with the product of 14.918 and any net overstatement being included within the calculation of Losses) (as so calculated, "**NOI Losses**"). Losses also shall include reasonable legal fees and expenses in connection with a Breach Event, including legal fees and expenses in defending any litigation that constitutes or relates to a Breach Event. Any dispute as to the determination of a Loss shall be determined by Baseball Arbitration.

- (ii) "**Breach Event**" shall mean any set of facts, circumstances, conditions or events that constitutes or results in the breach of any representation or warranty under, or a breach of a covenant or obligation to be performed and complied with prior to the Closing Date made by Sellers in, this Agreement or in any Transaction Document (including any Seller Estoppel and Seller GL Estoppel).
- (iii) "Deferred Maintenance Losses" shall mean the aggregate Losses arising out of all Deferred Maintenance Conditions to the extent the same exceed \$8 million.
  - (iv) "Entity Breach" shall mean all Breach Events that are not Property Breaches.
  - (v) "Excess Losses" shall mean the amount of Losses equal to the sum of:
    - (a) the extent to which the sum of (x) the aggregate Losses arising from all Breach Events (other than Breach Events giving rise to Property-Related Adjustment Losses) and (y) all Property-Related Adjustment Losses (excluding the amount described in (b)) exceeds \$18,442,800 and
    - (b) the extent to which aggregate Deferred Maintenance Losses exceed \$5,000,000.
- (vi) "Property Breach" shall mean any Breach Event relating to those representations, warranties and/or covenants set forth in Sections 8.3 (Ownership of Properties), 8.11 (No Condemnation), 8.12 (Environmental Matters), 8.13 (Leasing Commissions and Tenant Allowances), 8.14 (Rent Rolls; Leases), 8.15 (Accounts Receivable Aging), 8.16 (Service Contracts), 8.19 (Ground Leases), 8.20 (No Other Contracts); 8.22 (Insurance), 8.23 (Compliance with Restrictions), 8.25 (Permits), 8.26 (Existing Policies), 8.27 (Shadow Anchors) and Section 7.2 (General Covenants).
- (vii) "**Property-Related Adjustment Losses**" shall mean (i) NOI Losses, to the extent the same exceed \$5,000,000 in the aggregate, if any, (ii) Deferred Maintenance Losses, if any, and (iii) Losses (other than NOI Losses) arising from any and all Property Breaches, subject to the \$250,000 threshold set forth in Section 15.7(c)(ii). For the avoidance of doubt, it is possible for there to be no Property-Related Adjustment Losses.

(viii) "Special Covenants" shall mean those covenants specifically set forth in Section 7.3 (Special Covenants) hereof.

(b) Exclusive Liabilities and Remedies. Sellers' sole and exclusive liabilities and obligations with respect to any Breach Event and Buyer's, Buyer's Affiliates' and the Company Group's sole and exclusive remedy with respect to the same shall be as set forth in this Section 15 and in Section 16.5 (Termination Upon Default by Sellers) except with respect to fraud. Buyer's, Buyer's Affiliates and the Investors' sole and exclusive liabilities and obligations with respect to any breach of a representation or warranty under, or any breach of a covenant or obligation to be performed and complied with prior to the Closing Date, under this Agreement and/or any and all Transaction Documents and the Sellers' and Sellers' Affiliates and Company Group's sole and exclusive remedy with respect to same shall be as set forth in this Section 15 and in Section 16.4 (Termination Upon Default by Buyer) except with respect to fraud.

#### (c) Determination of Losses.

(i) If Buyer becomes aware of any Breach Events and/or any Deferred Maintenance Conditions prior to the Deposit Date, or the Closing Date, as applicable, Buyer shall deliver to the Sellers, no later than six (6) Business Days before the Deposit Date or the Closing Date, as applicable, or if Buyer first becomes aware, on or after such sixth Business Day, promptly upon becoming so aware, but no later than the Deposit Date or Closing Date, as applicable, written notice thereof (a "Buyer's Breach Notice") setting forth a reasonably detailed description of the claimed Breach Events or Deferred Maintenance Conditions and Buyer's good faith calculation of each of the following categories of Losses, if any: (A) the Losses resulting from each of the claimed Breach Events (other than Breach Events giving rise to Property-Related Adjustment Losses); and (B) the Losses arising from Breach Events or Deferred Maintenance Conditions giving rise to Property-Related Adjustment Losses (all of the foregoing being collectively called "Alleged Losses").

(ii) Sellers shall, within five (5) Business Days of receipt of the Buyer's Breach Notice, provide Buyer with a written notice (a "Sellers' Response Notice") setting forth (A) which Alleged Losses, if any, Sellers agree with (the "Agreed Losses") and which Alleged Losses Sellers dispute (the "Disputed Losses"), and (B) Sellers' good faith reason for disputing the occurrence of a Breach Event or Deferred Maintenance Condition and/or Sellers' good faith calculation of the applicable Losses. The Alleged Losses set forth in a Buyer's Breach Notice shall be deemed Disputed Losses except to the extent a timely delivered Sellers' Response Notice deems the same to be Agreed Losses. If Sellers receive Buyer's Breach Notice fewer than six (6) Business Days before the Deposit Date or Closing Date, as applicable, then the Alleged Losses will be deemed Disputed Losses unless Sellers specify otherwise in their Sellers' Response Notice.

#### (d) Sellers' and Buyer's Pre-Deposit Options.

(i) If as of the Deposit Date both (A) the sum of (x) the Agreed Losses and the Disputed Losses arising from Breach Events (excluding Breach Events giving rise to Property-Related Adjustment Losses) and (y) Agreed Losses and Disputed Losses which are or may constitute Property-Related Adjustment Losses is less than or equal to \$18,442,800, and (B) the Agreed Losses and the Disputed Losses which are or may constitute Deferred Maintenance Losses are less than or equal to \$5 million, subject to the other terms and conditions of this Agreement, Buyer shall be obligated to and shall pay the Deposit to Sellers, and Buyer shall not have any claim for damages, liabilities or Losses against Sellers by reason of any of the facts, circumstances, conditions, or events underlying the Breach Events or Deferred Maintenance Conditions raised in Buyer's Breach Notice, provided that at Closing: (1) there shall be a Price Adjustment equal to the sum of such Agreed Losses and Disputed Losses, and (2) such Disputed Losses shall be deposited by Buyer in escrow pursuant to Section 2.5(a) (Disputed Price Adjustments and Prorations.) and such escrow deposit shall be subject to Baseball Arbitration and the provisions of Section 2.13 (Dropping Properties). Buyer shall have no other right or remedy with respect to the Losses arising from any Breach Event or Deferred Maintenance Conditions set forth in Buyer's Breach Notice other than to the extent of the applicable Price Adjustment. The amount of the Disputed Losses in escrow that are determined to be Losses pursuant to the Baseball Arbitration shall be disbursed to the Buyer and the amount of the Disputed Losses in escrow that are determined not to be Losses pursuant to the Baseball Arbitration shall be disbursed to the Sellers.

(ii) If as of the Deposit Date, (A) the sum of (x) the Agreed Losses arising from Breach Events (excluding Breach Events giving rise to Property-Related Adjustment Losses) and (y) the Agreed Losses which are Property-Related Adjustment Losses exceeds \$18,442,800 or (B) the Agreed Losses that constitute the Deferred Maintenance Losses exceed \$5,000,000 (all such Agreed Losses in clause (A) herein being referred to as the "Aggregate Loss Amount"), and in either such case there are no Disputed Losses (if there are Disputed Losses, Section 15.2(d)(iii) would apply), then Buyer shall have the right (by notice to Sellers delivered at least six (6) Business Days prior to the Deposit Date or if Buyer first becomes aware of such Breach Event on or after such sixth (6th) Business Day, promptly upon becoming so aware, but no later than the Deposit Date) to either: (1) subject to the other terms and conditions of this Agreement, make the Deposit and consummate the Closing, except that the Price Adjustment shall only be equal to the Aggregate Loss Amount less the Excess Loss (in which event the Buyer shall be deemed to have irrevocably waived any and all rights to claim a Breach Event or Deferred Maintenance Condition giving rise to such Excess Loss and irrevocably waived any other right or remedy with respect to the Losses arising from any Breach Event or Deferred Maintenance Condition set forth in Buyer's Breach Notice other than to the extent of the applicable Price Adjustment); or (2) terminate this Agreement as of the Deposit Date, whereupon none of the parties hereto shall have any further obligations to the other (with the exception of those obligations which expressly survive the termination of this Agreement). If Buyer shall fail to deliver such notice by such date, subject to the other terms and conditions of this Agreement, Buyer shall be deemed to have elected to make the Deposit, and Buyer shall be obligated to consummate the Closing pursuant to clause (1) of this Section 15.2(d)(ii). If Buyer elects in accorda

giving Buyer notice at least two (2) Business Days before the Deposit Date (provided, that if Sellers receive Buyer's notice of termination less than two (2) Business Days prior to the Deposit Date, the Deposit Date shall be adjourned to the second (2nd) Business Day after the day on which Sellers received Buyer's notice of termination), that Sellers elect either (X) to provide Buyer with a Price Adjustment equal to the Aggregate Loss Amount, and subject to the other terms and conditions of this Agreement, proceed with the Deposit and the Closing under this Agreement (in which event, subject to the other terms and conditions of this Agreement, Buyer shall be obligated to make the Deposit, and Buyer shall be obligated to consummate the Closing pursuant to this clause (X) of Section 15.2(d)(ii)) or (Y) exercise the rights of Sellers as provided in Section 2.13 (Dropping Properties) of this Agreement (the "Section 2.13 Rights"), to the extent that the exercise of such rights would cause the provisions of this Section 15.2(d)(ii) to be inapplicable. If Sellers shall fail to timely elect to reinstate this Agreement, this Agreement shall terminate as provided for in clause (2) of this Section 15.2(d)(ii).

(iii) If as of the Deposit Date (A) the sum of (x) the Agreed Losses and Disputed Losses arising from Breach Events (excluding Breach Events giving rise to Property-Related Adjustment Losses) and (y) the Agreed Losses and Disputed Losses which are or may constitute Property-Related Adjustment Losses exceeds \$18,442,800 or (B) the total of the Agreed Losses and Disputed Losses which are or may constitute Deferred Maintenance Losses exceeds \$13,000,000, then Buyer shall have the right (by notice to Sellers delivered at least six (6) Business Days prior to the Deposit Date or if Buyer first becomes aware of such Breach Event on or after such sixth (6th) Business Day, promptly upon becoming so aware, but no later than the Deposit Date) to either: (1) subject to the other terms and conditions of this Agreement, make the Deposit and consummate the Closing in accordance with Section 15.2(d)(i), provided that the Price Adjustment shall only be equal to the total amount of (X) Agreed Losses and Disputed Losses arising from all Breach Events not to exceed \$18,442,800 (minus the amount of any Deferred Maintenance Losses not to exceed \$5,000,000) and (Y) the Deferred Maintenance Losses in an amount not to exceed \$5,000,000 (in which event the Buyer shall be deemed to have irrevocably waived any and all rights to claim a Breach Event or Deferred Maintenance Losses giving rise to such Excess Loss and irrevocably waived any other right or remedy with respect to the Losses arising from any Breach Event or Deferred Maintenance Losses set forth in Buyer's Breach Notice other than to the extent of the applicable Price Adjustment); or (2) terminate this Agreement as of the Deposit Date, whereupon none of the parties hereto shall have any further obligations to the other (with the exception of those obligations which expressly survive the termination of this Agreement). If Buyer shall fail to deliver such notice by such date, subject to the other terms and conditions of this Agreement, Buyer shall be deemed to have elected to make the Deposit and consummate the Closing pursuant to clause (1) of this Section 15.2(d)(iii). If Buyer elects in accordance with this Section 15.2(d)(iii) to terminate this Agreement as of the Deposit Date, then, Sellers shall have the right, but not the obligation, to reinstate this Agreement by giving Buyer notice, at least two (2) Business Days before the Deposit Date (provided, that if Sellers receive Buyer's notice of termination less than two (2) Business Days prior to the Deposit Date, the Deposit Date shall be adjourned to the second (2nd) Business Day after the day on which Sellers received Buyer's notice of termination), that Sellers have elected either to (X) submit the determination of Disputed Losses to Baseball Arbitration (in which event, subject to the other terms and conditions of this Agreement, Buyer shall make the Deposit and to consummate the Closing in accordance with Section 15.2(d)(i) except that the Price Adjustment shall equal the sum of Agreed Losses and Disputed Losses, and any Disputed

Losses shall be held in escrow until same are determined to be, or not to be, Losses by the Baseball Arbitration) or (Y) exercise the Section 2.13 Rights to the extent that the exercise of such rights will cause the provisions of this Section 15.2(d)(iii) to be inapplicable. If Sellers shall fail to timely reinstate this Agreement, this Agreement shall terminate as provided for in clause (2) of this Section 15.2(d)(iii). If Sellers elect to subject the determination of Disputed Losses to Baseball Arbitration, once the amount of the Disputed Losses that are Losses have been determined by Baseball Arbitration, said amount shall be deemed "Agreed Losses" for purposes of this Agreement. The amount determined to be Agreed Losses shall be released to Buyer (or, if the dispute is resolved prior to Closing, credited as a Price Adjustment described above) and the amount determined not to be Losses shall be paid or released to Sellers as applicable.

- (e) <u>Post-Deposit/Pre-Closing Breach Notices</u>. If Buyer delivers a Buyer's Breach Notice after the Deposit Date and before the Closing Date, Buyer shall be entitled to a Price Adjustment equal to the Agreed Losses, subject to the limitations as to amount set forth in Section 15.7 (assuming for this purpose that the Buyer had asserted the claim regarding the applicable Breach Event after the Closing), with respect to any Breach Event for which a Price Adjustment may be taken. If Sellers elect to subject the determination of Disputed Losses to Baseball Arbitration, once the amount of the Disputed Losses that are Losses have been determined by Baseball Arbitration, said amount shall be deemed "Agreed Losses" for purposes of this Agreement. The amount determined to be Agreed Losses shall be released to Buyer (or, if the dispute is resolved prior to Closing, credited as a Price Adjustment described above) and the amount determined not to be Losses shall be paid or released to Sellers as applicable.
- **15.3.** <u>Post-Closing Indemnity By Sellers</u>. From and after the Closing, subject to the terms, conditions and limitations of this Section 15, each Seller, jointly and severally, hereby agrees to indemnify, defend and hold harmless Buyer, its directors, officers, employees, members and controlled and controlling persons (hereinafter "Buyer's Affiliates") and the Company Group from and against all Claims asserted against, resulting to, imposed upon, or incurred by Buyer, Buyer's Affiliates, the Company Group, their subsidiaries or the Properties, directly or indirectly, by reason of, arising out of or resulting from (except to the extent the Loss was subject to a Buyer's Breach Notice and was either expressly waived in writing or was the subject of a Price Adjustment or Baseball Arbitration):
- (a) The inaccuracy or breach of any representation or warranty of Sellers contained in or made pursuant to this Agreement or any other Transaction Document (without regard to any materiality qualifications set forth therein);
- (b) The material breach of any covenant of Sellers contained in this Agreement or any other Transaction Document or the material breach before Closing of any covenant of the Companies contained in this Agreement or any other Transaction Document, other than in each case the Special Covenants;
  - (c) The breach of any Special Covenants;
- (d) Debts, liabilities or obligations of any entity in the Company Group arising from facts, circumstances, events or agreements which occurred, or which otherwise

accrued, prior to the Adjustment Time, but excluding obligations accruing prior to the Adjustment Time under the Leases, Ground Leases, Material Contracts, the Existing Loans and other obligations and liabilities for which Buyer receives a credit against the Purchase Price pursuant to Section 2.2 (Price Adjustments), Section 3 (Prorations) or elsewhere in this Agreement (collectively, "Entity Claims");

- (e) Any Claim brought by or on behalf of any broker or finder retained, employed or used by Sellers or the Company Group in connection with this Agreement, the Transactions or the negotiations thereof, whether or not disclosed herein;
- (f) Any Claim arising from any inaccuracy in any Sellers' Estoppels or Sellers' GL Estoppel provided by Sellers pursuant to Sections 11.2(d) and 11.2(e);
- (g) Any Claim asserted against any entity in the Company Group (x) arising or accruing prior to Closing Date for (i) violations under ERISA, (ii) violations under securities Laws, (iii) breaches of fiduciary duties owed by any entity in the Company Group or their Affiliates to Persons other than any entity in the Company Group, and/or (iv) fraud, or (y) for Taxes attributable to any Tax period ending on or before the Closing Date or, in the case of any Tax period which includes, but does not end on, the Closing Date, the portion of such period up to and including the Closing Date (collectively, "Special Entity Claims");
- (h) Claims by third parties arising from the presence of Hazardous Substance or Waste at, on, about, under or within or emanating from any Ongoing Remediation Site or Future Remediation Site which is reported by Buyer to Seller prior to Buyer obtaining a "no further action" letter or equivalent determination from the appropriate Governmental Authorities with respect to such Property, or in the case no Future Remedial Action is recommended in the Phase II Report obtained by Buyer for a Future Remediation Site, the date upon which the final written Phase II Report for such Property is delivered to Buyer. This indemnification obligation shall include the investigation and remediation costs resulting from such third party claims (collectively, "**Third Party Environmental Claims**"). (For purposes of avoiding confusion, claims related to Hazardous Substance or Waste arising from real property sold or otherwise disposed of by Company Group prior to Closing and from Dropped Assets pursuant to Section 2.13 (Dropping Properties) shall be treated as Entity Claims subject to indemnification pursuant to Section 15.3(d).)
- (i) The costs and expenses arising from the litigation matters set forth on Schedule 8.9 (Litigation), including reasonable legal fees and costs of performing settlements (provided that Buyer shall bear the expense of corrective construction activity), and provided further that Sellers shall have the right to control and settle all such litigation (other than the litigation relating to Village Commons), and provided further that Buyer's consent, which shall not be unreasonably withheld, shall be required to settle any action that will impose material expenses or burdens on Buyer or the Company Group (collectively, "ADA Litigation Matters").

As used in this Agreement, the term "Claim" shall mean (i) all debts, liabilities and obligations; (ii) all Losses, damages, judgments, awards, settlements, costs and expenses (including interest (including prejudgment interest in any litigated matter), penalties, court costs

and attorneys fees and expenses); and (iii) all demands, claims, suits, actions, investigations, causes of action, proceedings and assessments. (For purposes of clarification, any and all claims for Losses arising from Breach Events made by Buyer after the Closing are included in the definition of "Claim.")

No amount of indemnity Claim shall be payable under this Section 15.3 with respect to any Loss or Claim that was previously taken into account in calculating the amount of any Price Adjustment or Proration Item (or if the Loss or Claim was the subject of Buyer's Breach Notice even if no credit was given for such Loss or Claim, for example, because the applicable amount limitation was exceeded).

- **15.4.** <u>Post-Closing Indemnity By Buyer</u>. From and after the Closing, Buyer shall indemnify, defend and hold harmless each Seller and their Affiliates or any direct or indirect partner, member, manager, trustee, director, shareholder, controlling person, Affiliate officer, attorney, employee, agent or broker of the foregoing, and any of their respective heirs, successors, personal representatives and assigns (collectively, "**Seller Indemnified Party**") from and against all Claims asserted against, resulting to, imposed upon or incurred by any such person, directly or indirectly, by reason of or resulting from:
- (a) the inaccuracy or breach of any representation or warranty of Buyer contained in or made pursuant to this Agreement or other Transaction Document;
  - (b) the material breach of any covenant of Buyer contained in this Agreement or other Transaction Document;
- (c) other than matters subject to indemnification by Sellers, debts, liabilities or obligations of the Company Group and Claims against any Seller Indemnified Party with regard to the Properties or ownership of the Membership Interests arising from facts, circumstances, events or agreements which occurred, or which otherwise accrued, from and after the Adjustment Time ("Sellers' Post Closing Occurrence Claims"); and
- (d) any Claim brought by or on behalf of any broker or finder retained, employed or used by Buyer or Buyer's Affiliates in connection with this Agreement, the Transactions or the negotiations thereof whether or not disclosed herein.
- **15.5.** <u>Indemnification of Third-Party Claims</u>. The obligations and liabilities of any party to indemnify any other under this section with respect to Claims relating to third parties shall be subject to the following terms and conditions:
- (a) Notice and Defense. The party or parties to be indemnified (whether one or more, the "Indemnified Party") will give the party from whom indemnification is sought (the "Indemnifying Party") prompt written notice of any such Claim, and the Indemnifying Party will undertake the defense thereof, at its own expense, by representatives chosen by it. Failure to give such notice shall not affect the Indemnifying Party's duty or obligations under this section, except to the extent the Indemnifying Party is prejudiced thereby. So long as the Indemnifying Party is defending any such Claim actively and in good faith, the Indemnified Party shall not settle such Claim. The Indemnified Party shall make available to the Indemnifying Party or its representatives all records and other materials required by them and in the possession or under

the control of the Indemnified Party, for the use of the Indemnifying Party and its representatives in defending any such Claim, and shall in other respects give reasonable cooperation in such defense.

- (b) <u>Failure to Defend</u>. If the Indemnifying Party, within a reasonable time after notice of any such Claim, fails to defend such Claim actively and in good faith, the Indemnified Party will (upon further notice) have the right to undertake the defense, compromise or settlement of such Claim or consent to the entry of a judgment with respect to such Claim, on behalf of and for the account and risk of the Indemnifying Party, and the Indemnifying Party shall thereafter have no right to challenge the Indemnified Party's defense, compromise, settlement or consent to judgment therein. In any event, the Indemnified Party shall have the right to participate in the defense of the Claim.
- (c) <u>Indemnified Party's Rights</u>. Anything in this Section 15.5 to the contrary notwithstanding, (i) if there is a reasonable probability that a Claim may materially and adversely affect the Indemnified Party other than as a result of money damages or other money payments, the Indemnified Party shall have the right to defend, compromise or settle such Claim, or (ii) if the amount of the Claim (when aggregated with any previous amounts paid or other Claims threatened or pending, have a reasonable likelihood of exceeding the maximum amount payable under either paragraph 15.7(c), 15.7(d) or 15.7(e) if applicable, the Indemnified Party shall have the right to defend (at the Indemnifying Party's expense), and, with the Indemnifying Party's Consent, which shall not be unreasonably withheld or delayed, compromise or settle such Claim and the Indemnifying Party shall have the right to participate in such defense, and (iii) the Indemnifying Party shall not, without the written consent of the Indemnified Party, settle or compromise any Claim or consent to the entry of any judgment which does not include as an unconditional term thereof the giving by the claimant or the plaintiff to the Indemnified Party of a release from all liability in respect of such Claim.

15.6. Payment. The Indemnifying Party shall promptly pay the Indemnified Party any amount due under this section, which payment may be accomplished in whole or in part, at the option of the Indemnified Party, by the Indemnified Party setting off any amount owed to the Indemnifying Party by the Indemnified Party. To the extent set-off is made by an Indemnified Party in satisfaction or partial satisfaction of an indemnity obligation under this section that is disputed by the Indemnifying Party, upon a subsequent determination by final judgment not subject to appeal that all or a portion of such indemnity obligation was not owed to the Indemnified Party, the Indemnified Party shall pay the Indemnifying Party the amount which was set off and not owed together with interest from the date of set-off until the date of such payment at an annual rate equal to the "prime rate" as published in The Wall Street Journal on the date the set-off is made. Upon judgment, determination, settlement or compromise of any third party Claim, the Indemnifying Party shall pay promptly on behalf of the Indemnified Party, and/or to the Indemnified Party in reimbursement of any amount theretofore required to be paid by it, the amount so determined by judgment, determination, settlement or compromise and all other Claims of the Indemnified Party with respect thereto, unless in the case of a judgment an appeal is made from the judgment. If the Indemnifying Party desires to appeal from an adverse judgment, then the Indemnifying Party shall post and pay the cost of the security or bond to stay execution of the judgment pending appeal. Upon the payment in full by the Indemnifying Party of such amounts, the Indemnifying Party shall succeed to the rights of such Indemnified Party, to the extent not waived in settlement, against the third party (other than Tenants) who made such third party Claim.

### 15.7. Limitations on Indemnification.

(a) <u>Unlimited Claims</u>. There is an unlimited survival period for (i) Broker Claims; (ii) Sellers' Post-Closing Occurrence Claims; and (iii) Third Party Environmental Claims (except as Third Party Environmental Claims may otherwise be limited by Section 15.3(h) (Post-Closing Indemnity By Sellers) of this Agreement) (collectively, the "Unlimited Claims").

## (b) Survival Periods.

- (i) Claims arising from Entity Breaches shall have a survival period of 18 months following the Closing Date.
- (ii) Special Entity Claims and claims arising from breaches of representations and warranties in Sections 8.1 (Organization; Authorization), 8.2 (Sellers), 8.8 (Tax Matters), and 8.10 (No Employees) and Claims arising from breaches of Special Covenants and ADA Litigation Matters shall have a survival period equal to the applicable statute of limitation under federal and state laws relating thereto.
  - (iii) If a Claim falls within more than one survival period, the longest survival period shall apply.
  - (iv) All Claims arising from Property Breaches shall have a survival period of nine months after Closing.
- (v) A Claim shall only be deemed duly initiated by written notice of the Claim to the Indemnifying Party or filing of a complaint for enforcement of same in a court of competent jurisdiction. Once a Claim is duly initiated within the applicable survival period, the rights and liabilities of the Indemnifying Party and Indemnified Party shall survive until a final non-appealable determination by a court of competent jurisdiction. No party shall have the right to initiate a Claim after the expiration of any applicable survival period with respect to such Claim.
- (c) Amount Limitation for Property Claims. Notwithstanding anything to the contrary provided for in this Agreement, neither Sellers, nor any Indemnifying Party related to Sellers, shall have any liability, responsibility or obligation of any kind or nature whatsoever after the Closing for any Claims arising from or relating to Property Breaches: (1) to the extent the Losses which are the subject of such Claim were noted in a Buyer's Breach Notice delivered on or prior to the Closing Date; (2) unless the aggregate amount of all Losses arising from or related to Property Breaches (excluding Property Claims that relate to or constitute NOI Losses) exceeds \$250,000 (which shall be a threshold but not a deductible and once such threshold is exceeded, Sellers shall be responsible for the entire amount of such \$250,000 threshold); (3) in the case of NOI Losses, except to the extent that the aggregate NOI Losses (whether identified before or after Closing) exceed \$5,000,000; and (4) to the extent the aggregate of (A) all Price Adjustments for Property- Related Adjustment Losses pursuant to Section 15.2 hereof (less any

amounts in the Disputed Amount Escrow that are delivered to Seller) and (B) all Claims arising from or related to Property Breaches exceeds \$18,442,800. If any fact, circumstance or condition forming a basis for any Claim (or Loss) arising from or related to Property Breaches shall overlap with any fact, circumstance, condition, agreement or event forming the basis of any other Claim (or Loss), there shall be no duplication in the calculation of the amount of the Claims (or Losses).

- (d) Amount Limitation For Entity Claims. Notwithstanding anything to the contrary provided for in this Agreement, neither Sellers, nor any Indemnifying Party related to Sellers shall have any liability, responsibility or obligation of any kind or nature whatsoever after the Closing for any Claims arising from or related to Entity Breaches (1) to the extent the Losses which are the subject of such Claim were noted in a Buyer's Breach Notice received on or prior to the Closing Date; (2) unless the aggregate amount of all such Losses exceeds \$250,000 (which shall be a threshold and not a deductible, and once such threshold is exceeded, Sellers shall be responsible for the entire amount of such \$250,000 threshold); or (3) to the extent that the aggregate of (A) all Losses pursuant to Section 15.2 arising from or related to Entity Breaches exceeds \$200,000,000. If any fact, circumstance, condition, agreement or event forming a basis for any Claim (or Loss) arising from or related to Entity Breaches shall overlap with any fact, circumstance, condition, agreement or event forming the basis of any other Claim (or Loss), there shall be no duplication in the calculation of the amount of the Claims (or Losses). The foregoing limitations shall not apply to Broker Claims (all of which are unlimited).
- (e) <u>Limitation with Respect to Special Covenants</u>, <u>Broker Claims</u>, <u>Special Entity Claims</u>, <u>Third Party Environmental Claims and ADA Litigation Matters</u>. Notwithstanding anything to the contrary provided for in this Agreement, neither Sellers, nor any Indemnifying Party related to Sellers shall have any liability, responsibility or obligation of any kind or nature whatsoever for the breach of any Special Covenant or for any Broker Claims, Special Entity Claims, Third Party Environmental Claims or ADA Litigation Matters to the extent that the aggregate liability for all Claims (of all types) exceeds the Purchase Price. If any fact, circumstance, condition, agreement or event forming a basis for any Claim (or Loss) arising from or related to the breach of any Special Covenant or for any Broker Claims, Special Entity Claims, Third Party Environmental Claims or ADA Litigation Matters shall overlap with any fact, circumstance, condition, agreement or event forming the basis of any other Claim (or Loss), there shall be no duplication in the calculation of the amount of the Claims (or Losses).
- **15.8.** Release of Buyer's Other Environmental Remedies. Buyer agrees that its rights and remedies relating to the use, presence, discharge or release of Hazardous Substances or Waste on, under, in, above, about, within, without or emanating from any of the Properties (or any portion thereof) shall be expressly limited to its rights under Sections 2.5 (Escrows at Closing), 5.6 (Environmental Remediation and Insurance), 8.12 (Environmental Matters), 15.3(h) (Breaches, Remedies and Indemnification), 16 (Termination; Default) and 20 (Miscellaneous), and such rights and remedies constitute Buyer's sole and exclusive means of recourse with respect thereto and except as provided herein Buyer waives and releases any and all Claims it may have against Sellers, now or in the future arising under, in connection with or relating to any Environmental Law (whether by statute, regulation, or common law) in connection with any Property.

**15.9.** No Waiver. Except to the extent expressly waived in writing or deemed waived by reason of a Price Adjustment pursuant to Section 15.2(d), the Closing of the Transactions shall not constitute a waiver by any party of its rights to indemnification hereunder.

# 16. TERMINATION; DEFAULT

**16.1.** <u>Termination By Mutual Agreement</u>. This Agreement may be terminated without further liability of any party at any time prior to the Closing by mutual written agreement of Buyer and Sellers in which case Sellers shall return the Deposit to Buyer, plus all accrued but unpaid interest.

# 16.2. Termination before Deposit.

- (a) <u>Termination by Buyer</u>. If there has been a failure of satisfaction of a condition to the obligations of Buyer contained in Section 11.2 (Conditions to Obligations of Buyer in Connection With Deposit) (other than through the failure of Buyer to comply fully with its obligations under this Agreement) which has not been waived by the Deposit Date, then Buyer may, by written notice to Sellers at any time prior to the payment of the Deposit, terminate this Agreement or elect to seek specific performance in accordance with Section 16.5.
- (b) <u>Termination by Sellers</u>. If there has been a failure of satisfaction of a condition to the obligations of the Sellers contained in Section 11.1 (Conditions to Obligations of Sellers at Deposit Date) (other than through the failure of Sellers to comply fully with their obligations under this Agreement) which has not been waived by the Deposit Date, then Sellers may, by written notice to Buyer at any time prior to the payment of the Deposit, terminate this Agreement.
- (c) <u>Termination for Excess Losses</u>. This Agreement may be terminated by Buyer pursuant to Section 15.2(d)(ii) or (iii), subject to reinstatement by Sellers in accordance with the provisions of Section 15.2.
- **16.3.** <u>Termination If Closing Fails to Occur</u>. Either Buyer or Sellers may terminate if the Closing has not occurred (other than through the failure of any party seeking to terminate this Agreement to comply fully with its obligations under this Agreement) on or before October 31, 2005, or such later date as the parties may agree.

# 16.4. <u>Termination Upon Default by Buyer</u>.

(a) In the event that Buyer and Investors breach their duty to tender the Deposit on the Deposit Date pursuant to the terms and conditions of this Agreement, then Sellers shall have the right to terminate this Agreement on notice to Buyer and Buyer shall pay to the Companies liquidated damages in the sum of \$75,000,000 (expressly subject, in the case of the Investors, to the obligation of Investors pursuant to Section 20.1 (Guarantee By Investors) hereof), as their sole exclusive remedy (whether at law or in equity) and as liquidated damages (actual damages being difficult if not impossible to ascertain) and not as a penalty and this Agreement shall be absolutely, automatically and completely null, void and of no further force or effect, and no parties shall have any further rights or obligations hereunder except for those items that, by the terms of this Agreement, expressly survive such termination. The liquidated

damages in the sum of \$75,000,000 shall be the full, agreed liquidated damages and all other claims to damages or other remedies being hereby expressly waived by Selling Parties. THE PARTIES HERETO EXPRESSLY AGREE AND ACKNOWLEDGE THAT SELLERS' ACTUAL DAMAGES IN THE EVENT OF A DEFAULT BY BUYER WOULD BE EXTREMELY DIFFICULT OR IMPRACTICABLE TO ASCERTAIN AND THAT THE SUM OF \$75,000,000 REPRESENTS THE PARTIES' REASONABLE ESTIMATE OF SUCH DAMAGES.

- (b) In the event that, after payment of the Deposit, Buyer breaches its duty to tender the Purchase Price pursuant to the terms and conditions of this Agreement, then Sellers shall have the right to terminate this Agreement on notice to Buyer, Buyer shall pay to the Companies liquidated damages in the sum of \$300,000,000 (expressly subject, in the case of the Investors, to the obligation of Investors pursuant to Section 20.1 (Guarantee By Investors) hereof), as their sole exclusive remedy (whether at law or in equity) and as liquidated damages (actual damages being difficult if not impossible to ascertain) and not as a penalty and this Agreement shall be absolutely, automatically and completely null, void and of no further force or effect, and no parties shall have any further rights or obligations hereunder except for those items that, by the terms of this Agreement, expressly survive such termination. The liquidated damages in the sum of \$300,000,000 shall be the full, agreed liquidated damages after the date of payment of the Deposit and all other claims to damages or other remedies being hereby expressly waived by Selling Parties. THE PARTIES HERETO EXPRESSLY AGREE AND ACKNOWLEDGE THAT SELLERS' ACTUAL DAMAGES IN THE EVENT OF A DEFAULT BY BUYER AFTER PAYMENT OF THE DEPOSIT WOULD BE EXTREMELY DIFFICULT OR IMPRACTICABLE TO ASCERTAIN AND THAT THE SUM OF \$300,000,000 REPRESENTS THE PARTIES' REASONABLE ESTIMATE OF SUCH DAMAGES.
- (c) Payments made by Buyer under this Section 16.4 shall be made to the Companies pursuant to the instructions delivered in writing by Sellers for such purpose.

# 16.5. Termination Upon Default by Sellers.

- (a) In the event the payment of the Deposit does not occur on the Deposit Date or the Closing does not occur on the Closing Date as provided herein by reason of a breach by Selling Parties of their representations, warranties, covenants and agreements contained in this Agreement, it being understood that payment of the Deposit and occurrence of the Closing are not expressly conditioned on an absence of such a breach, Buyer may elect, by notice to Sellers as its sole and exclusive rights and remedies of Buyer, either to (i) terminate this Agreement or (ii) exercise a right of specific performance.
- (b) If Buyer shall elect to terminate the Agreement pursuant to Section 16.5(a), then Buyer shall receive a return of its Deposit, plus accrued but unpaid interest pursuant to Section 2.3(a) (Payment of Deposit). In addition, Buyer shall receive from the Companies reimbursement for all of Buyer's and Investors' reasonable, out of pocket Transaction-related expenses, including accounting and legal fees in connection with the preparation of this Agreement and other Transaction-related expenses if: (i) Selling Parties have intentionally and materially breached any of their covenants; (ii) such breach is directly related to the basis for

such termination; and (iii) Buyer provided Sellers with written notice of such breach and Sellers failed to cure the breach within five (5) Business Days following receipt of written notice of such breach. Except as expressly provided in this Agreement, upon such payments by the Companies, this Agreement shall be absolutely, automatically and completely null, void and of no further force or effect, and no parties shall have any further rights or obligations hereunder, except for those items that, by the terms of this Agreement, expressly survive such termination.

- (c) If Buyer shall elect to specifically enforce the obligation of Sellers pursuant to this Section 16.5, Buyer shall commence an action for specific performance of Sellers' obligations under this Agreement within sixty (60) days after Buyer's tender of notice in the forum specified in Section 20.23 (Venue). THE PARTIES IN THIS AGREEMENT ACKNOWLEDGE AND AGREE THAT THE REMEDY OF SPECIFIC PERFORMANCE IS NECESSARY AND APPROPRIATE AS THE SUBJECT MATTER OF THIS AGREEMENT IS PECULIAR, UNIQUE AND HAS NO READILY ASCERTAINABLE MARKET VALUE SUCH THAT MONETARY DAMAGES WOULD BE ADEQUATE TO MAKE BUYER WHOLE.
- (d) Buyer shall be deemed to have elected to terminate this Agreement, and to have waived the right to seek specific performance, if Buyer fails to deliver to Sellers written notice of its intent to file a cause of action for specific performance against Sellers on or prior to the tenth Business Day following the Deposit Date, or the Closing Date, as applicable, on which the payment of the Deposit or the Closing did not occur for the reasons set forth in Section 16.5(a), or having given Sellers such notice, Buyer fails to commence an action seeking specific performance in a court of competent jurisdiction within sixty (60) days.
- (e) Notwithstanding the foregoing, Buyer shall have the right to waive any default of Sellers, in whole or in part, in its sole and absolute discretion, and proceed to cause the Deposit to be paid on the Deposit Date or the Purchase Price at the Closing without any credit or adjustment to the Purchase Price or further liability of Sellers with respect to such default other than as expressly provided for in this Agreement.
- **16.6.** Exclusive Remedy Prior to Closing. The parties agree that Sections 16.4 or 16.5, as applicable, shall be the parties' sole and exclusive remedy (at law or in equity) if the Deposit is not paid on the Deposit Date or the Closing does not occur on the Closing Date in accordance with the terms and conditions of this Agreement.
- **16.7.** Effect of Termination. If this Agreement is duly terminated pursuant to Section 16 of this Agreement, (i) Sellers shall pay to Buyer the Deposit plus all accrued interest, and, if Sellers have duly terminated this Agreement pursuant to Section 16.4, less the amount of any liquidated damage to which the Sellers are entitled pursuant to Section 16.4; and (ii) upon the payment of the Deposit to Buyer, the Membership Interests and other documents held pursuant to the Escrow Agreement shall be released and the Blocked Account Agreement and the Escrow Agreement shall terminate. If this Agreement is terminated, all obligations of the Parties hereunder shall terminate, except for the obligations set forth in this Section 16 and in Sections 17 (Confidentiality) and 20 (Miscellaneous).

## 17. CONFIDENTIALITY; ANNOUNCEMENTS

- (a) Buyer expressly acknowledges and agrees that, unless and until the Closing occurs, except as otherwise contemplated by this Agreement, the Confidential Information shall be held in the strictest confidence by Buyer and shall not be disclosed by Buyer to any third persons without the prior written consent of Sellers, except to Buyer's legal counsel, accountants, consultants, officers, partners, directors and other agents and any prospective lenders, financial partners and their agents, consultants and representatives (collectively, the "Authorized Representatives"), and then only to the extent that any such Authorized Representatives, in Buyer's sole and reasonable judgment, needs to know such information for the purpose of Buyer's performance under this Agreement. Buyer agrees that it shall instruct each of its Authorized Representatives to maintain the confidentiality of the Confidential Information in accordance with the provisions of this Section 17. Upon the termination of this Agreement for any reason whatsoever, Buyer shall promptly return to Sellers, and shall instruct its Authorized Representatives to return to Sellers, all copies and originals of all Confidential Information provided to Buyer by Sellers, on or before ten (10) Business Days after such time as this Agreement is terminated. Nothing contained in this Section 17 shall preclude or limit either party from disclosing or accessing any Confidential Information (i) in connection with the party's enforcement of its rights following a disagreement hereunder or (ii) in response to lawful process or subpoena or other valid or enforceable order of a court of competent jurisdiction or any filings with Governmental Authorities required by reason of the Transactions or (iii) as may otherwise be required by applicable law, provided, to the extent practical, the disclosing party shall provide the other party with reasonable prior written notice of such disclosure obligation.
- (b) Sellers, Investors, and the Buyer agree (i) to consult with and cooperate with each other on the content and timing of all press releases and other public announcements relating to the Transactions contemplated by this Agreement and (ii) that the press release to be issued upon execution of this Agreement will be in a form agreed to by the parties hereto prior to the execution of this Agreement.
- (c) Notwithstanding the foregoing, the parties acknowledge and agree that Investors are permitted to disclose the Agreement, the terms of the Transactions, and general information about the Properties and their financial performance and if Investors deem appropriate, a copy of, and the fact of the execution and delivery of, this Agreement, in connection with their disclosure obligations under the securities laws and in connection with MCW's efforts to sell units in a public offering.
- (d) Sellers expressly covenant and agree that the Confidential Information and all confidential information of Buyer and Investors be held in strictest confidence by Sellers and the Company Group after the Closing, and may not be disclosed to third parties after the Closing without the prior written consent of Buyer.

# 18. ASSIGNMENT

Buyer may not assign this Agreement, nor any of the rights, interests or obligations hereunder, directly or indirectly (including any direct or indirect change in equity structure, ownership or control of any Person which is the Buyer), in whole or in part, by operation of law

or otherwise, without Sellers' prior written consent, to any Person, and any such assignment made without Sellers' consent shall be void *ab initio*. Notwithstanding the foregoing, Buyer shall have the right to assign all or any portion of its right, title and interest in this Agreement to an Affiliate in which Investors collectively have a 20% or more equity interest or which is under common control of Investors to take title to the Membership Interests, provided, that in the event of any such assignment, Buyer named in this Agreement shall remain liable to Sellers with respect to all liabilities and obligations of Buyer to Sellers arising under this Agreement. Buyer shall also have the right to designate wholly-owned entities as the entity(ies) to acquire particular Membership Interests, provided, that in the event of any such designation, Buyer named in this Agreement shall remain liable to Sellers with respect to all liabilities and obligations of Buyer to Sellers arising under this Agreement. Neither Seller may assign this Agreement, nor any of its rights, interests or obligations hereunder without the prior written consent of Buyer.

# 19. RESOLUTION OF DISPUTES

Any dispute, controversy or claim arising out of or relating to Price Adjustments (Section 2.2), Losses, Breach Events, Future Remedial Actions, Deferred Maintenance Conditions, Prorations (Section 3.1), (the amount of same and if such condition, fact or circumstance constitutes a Breach Event, Deferred Maintenance Condition, or if a Future Remedial Action is required, collectively, a "**Disputed Amount**") shall be solely and exclusively resolved by the following procedure referred to as "**Baseball Arbitration**": Notwithstanding anything herein to the contrary, Entity Claims and Claims for breaches of representations and warranties other than those referred to in Section 15.2(a)(vi) (Property Breach) ARE NOT SUBJECT TO BASEBALL ARBITRATION.

- (a) The parties shall, for a period of not less than three (3) Business Days and for such longer period of time as they mutually agree, and by such means as they may deem appropriate in their sole discretion, attempt to mutually agree upon a value for any such Disputed Amount.
- (b) If the dispute is not resolved in the manner described in Section 19(a), any party may initiate Baseball Arbitration by serving the other party with a "Notice of Intent to Pursue Arbitration." At a mutually agreed upon date and time, but no more than five Business Days after the date the Notice of Intent to Pursue Arbitration was served, each party shall simultaneously present to the other party their final determination of the monetary value of the Disputed Amount (the "Final Offers").
- (c) If the higher of the Final Offers submitted by the parties is not more than one hundred five percent (105%) of the lower of such amounts, the value in question shall be deemed to be the average of the two (2) submitted Final Offers and the procedures set forth below shall not be required. If, however, the higher value is more than one hundred five percent (105%) of the lower value, then the procedure set forth below shall be followed.
- (d) If the Disputed Amount is not resolved in the manner described in Section 19(a) or (c), either Buyer or Sellers (the "Initiating Party") shall have the right to appoint one (1) Qualified Arbitrator and provide the other party (the "Responding Party") with notice thereof. Within five (5) Business Days following the Initiating Party's giving of notice to the

Responding Party designating such party's Qualified Arbitrator, the Responding Party shall either (i) agree that such Qualified Arbitrator shall determine the Disputed Amount or (ii) appoint a second (2nd) Qualified Arbitrator and give notice to the Initiating Party of the Qualified Arbitrator so appointed. If the Responding Party fails to appoint such second (2nd) Qualified Arbitrator within such five (5) Business Day period, the Qualified Arbitrator appointed by the Initiating Party shall proceed to make the determination of the Disputed Amount as herein set forth, and the determination of the Disputed Amount made by the Qualified Arbitrator appointed by the Initiating Party shall be conclusive and binding on the parties. If two (2) Qualified Arbitrators are appointed, then such appointed Qualified Arbitrators shall thereafter mutually appoint a third (3rd) Qualified Arbitrator and give notice thereof to the parties. If the two (2) appointed Qualified Arbitrators fail to appoint a third (3rd) Qualified Arbitrator within ten (10) days following the giving of notice by the Responding Party notifying the Initiating Party of the appointment of the second (2nd) Qualified Arbitrator, either party may petition a court of competent jurisdiction to appoint a third (3rd) Qualified Arbitrator.

(e) On the seventh (7th) Business Day following (i) the Initiating Party's giving of notice to the Responding Party designating the Initiating Party's Qualified Arbitrator, if the Responding Party either agrees that the Initiating Party's Qualified Arbitrator shall determine the Disputed Amount or fails to appoint a second (2nd) Qualified Arbitrator within the five (5) Business Day period specified in Section 19(d), or (ii) the appointment of the third (3rd) Qualified Arbitrator, if the Responding Party appoints a second (2nd) Qualified Arbitrator within the five (5) Business Day period specified in Section 19(b), Sellers and Buyer shall each provide, their respective the Final Offers set forth in Section 19(b) to the Qualified Arbitrator(s) and the other party together with written statements and supporting materials in support of their respective Final Offer and in opposition to the competing Final Offer. In the event that either Sellers or Buyer shall fail to submit its Final Offer to the Qualified Arbitrator(s) within the seven (7) Business Day period specified above, then the Disputed Amount in question shall be deemed to be the amount submitted by the non-defaulting party, and such determination of the Disputed Amount shall be conclusive and binding on the parties.

(f) The Qualified Arbitrator, or the three (3) Qualified Arbitrators jointly, as the case may be, shall promptly fix a date and place for the determination of the Disputed Amount, which date shall not be later than forty-five (45) days from the last day provided for the parties' delivery of the materials pursuant to Section 19(e) or the conclusion of the evidential hearing provided for below. In setting such time for determination, the Qualified Arbitrator(s) shall determine whether it is necessary and appropriate for the parties to submit additional evidence or argument and, if so, set a date, time, and location for such a proceeding which shall occur in Atlanta, Fulton County, Georgia and sufficiently in advance of the time for determination so as to permit a decision within ninety (90) days the parties' delivery of the materials pursuant to Section 19(e). Upon the request of any party, the Qualified Arbitrator(s) shall permit discovery in accordance with the Commercial Arbitration Rules of the American Arbitration Association. The Qualified Arbitrator or, if there are three (3) Qualified Arbitrators, the Qualified Arbitrators by majority vote, shall select one or the other of the amounts submitted by the parties, but no other amount, as the amount closest, in the opinion of such Qualified Arbitrator(s), to the actual amount of the Disputed Amount, and such amount shall constitute the Disputed Amount in question and such determination of the Disputed Amount shall be conclusive and binding on the parties. The Qualified Arbitrator(s) shall have no authority to

conclude that the Disputed Amount is not either of the amounts submitted by the parties. The final decision of the Qualified Arbitrator(s) is final and binding upon the parties and their successors in interest and judgment thereon shall be entered by any court of competent jurisdiction. Any award including payment of any delinquent amounts shall include interest on such amounts at 6.1% per annum simple interest, accruing daily.

(g) Prior to the final decision of the Qualified Arbitrator(s) the Buyer and Sellers shall pay for the services and costs of the Qualified Arbitrator(s) equally. Following the decision of the Qualified Arbitrator(s), the party whose Final Offer was selected by the Qualified Arbitrator(s) shall be entitled to reimbursement of all reasonable costs incurred in the arbitration, including costs and fees advanced for the Qualified Arbitrator(s) and reasonable attorneys' fees and costs and any expert witness fees and costs expended by that party, with such fees and expenses to be determined by the Qualified Arbitrator(s). Any fees and expenses awarded by the Arbitrator shall be paid within five (5) Business Days by the obligated party.

## 20. MISCELLANEOUS

- **20.1.** Guarantee By Investors. The Investors, jointly and severally, guarantee to Selling Parties the Buyer's performance of its obligations under this Agreement, subject to the terms, conditions and limitations contained herein, provided that the total aggregate liability of Investors for breaches of this Agreement (including breaches by Buyer) in respect of the guaranteed obligations shall not exceed \$75,000,000 in the aggregate prior to payment of the Deposit and the total amount payable in respect of the guaranteed obligations shall not exceed \$300 million of the Deposit in the aggregate after payment of the Deposit.
- 20.2. <u>Joint and Several Liability of the Sellers</u>. Each Seller shall be jointly and severally liable and responsible for all obligations of the Sellers under this Agreement. Notwithstanding anything to the contrary in this Agreement or any other Transaction Document, except for each Seller's obligation to deliver its Membership Interests in the Companies and for the representations and warranties in Section 8.2 (Sellers) (which in each case shall be a separate obligation of the applicable Seller), First Washington's exclusive liabilities and obligations with respect to or under this Agreement and any and all Transaction Documents, and Buyer's, Buyer's Affiliates' and the Company Group's sole and exclusive remedy with respect to First Washington, shall be limited to, individually and in the aggregate, the Agreed Percentage of the total amount of any Losses with respect to which any Buyer, Buyer's Affiliate or entity in the Company Group is entitled to receive indemnification under this Agreement. For purposes of this Section 20.2, "Agreed Percentage" shall mean First Washington's weighted average Combined Percentage Interest (as defined in the Organizational Documents of the Companies), as applicable, with respect to the Companies as of the Closing Date. The Sellers shall agree to the Agreed Percentage in accordance with the existing agreements between the Sellers, and shall notify Buyer on the Closing Date with respect to the finally determined Agreed Percentage. (To avoid confusion, Buyer may elect to recover from California Public Employees' Retirement System the total amount of Sellers' liabilities to Buyer.)
- **20.3.** <u>Survival</u>. All covenants, obligations, representations and warranties contained in this Agreement, the certificates delivered pursuant to Sections 12.1 (Deliveries by the Selling

Parties at Time of Deposit) and 12.2 (Deliveries by Buyer at Time of Deposit) and the other Transaction Documents will survive the Closing, except as set forth in Section 16 (Termination; Default).

**20.4.** Notices. Any and all notices, requests, demands or other communications hereunder shall be deemed to have been duly given if in writing and if transmitted by hand delivery with receipt therefore, by facsimile delivery or electronic mail (with a confirmation copy simultaneously transmitted by one of the other methods provided herein), by nationally recognized overnight courier, or by registered or certified mail, return receipt requested, first class postage prepaid, addressed as follows (or to such new address as the addressee of such a communication may have notified the sender thereof):

To Buyer, the Investors or (after the Closing) the

Companies: Macquarie CountryWide-Regency II, LLC

c/o Regency Centers Corporation 121 Forsyth Street, Suite 200 Jacksonville, FL 32202 Attention: Martin E. Stein, Jr. Telephone No.: (904) 598-7000 Fax No.: (904) 356-8214

E-mail: HStein@regencycenters.com

With copies (which shall not constitute notice) to:

Regency Centers Corporation 121 Forsyth Street, Suite 200 Jacksonville, FL 32202

Attention: Bruce M. Johnson, Managing Director and Chief Financial Officer

Telephone No.: (904) 598-7000 Fax No.: (904) 354-3448

E-mail: BJohnson@regencycenters.com

and with copies (which shall not constitute

notice) to:

Macquarie CountryWide Trust No. 1 Martin Place, Level 13 Sydney, NSW 2000 Australia

Attention: Kylie K. Rampa, Chief Executive Officer

Telephone No.: +61-2-8232-8502 Fax No.: +61-2-8232-6510

E-mail: Kylie.rampa@macquarie.com

and with copies (which shall not constitute notice) to: Mayer, Brown, Rowe & Maw LLP

190 S. LaSalle Street Chicago, Illinois 60603

Attention: Matthew A. Posthuma, Esq. Telephone No.: (312) 782-0600 Fax No.: (312) 701-7711

E-mail: mposthuma@mayerbrownrowe.com

and with copies (which shall not constitute notice) to: Foley & Lardner LLP One Independent Drive, Suite 1300 Jacksonville, FL 32202 Attention: Linda Y. Kelso, Esquire

Telephone No.: (904) 359-2000 Fax No.: (904) 359-8700 E-mail: lkelso@foley.com

To Sellers', the applicable Selling Parties, FWR or (before the Closing) the Companies:

California Public Employees' Retirement System

400 P Street

Sacramento, California 95814

Attention: Senior Investment Officer, Real Estate Investments

Fax No.: (916) 326-3965

and

First Washington Realty, Inc. 4350 East-West Highway, Suite 400 Bethesda, Maryland 20814

Attention: Stuart D. Halpert, Chairman Telephone No.: (301) 907-7800 Fax No.: (301) 907-4911

E-mail: shalpert@firstwash.com

With copies (which Shall not constitute notice) to:

Hogan & Hartson L.L.P. 555 13<sup>th</sup> Street, NW Washington, D.C. 20004 Attention: Bruce W. Gilchrist Telephone No.: (202) 637-5600 Fax No.: (202) 637-5910 E-mail: bwgilchrist@hhlaw.com

and with copies (which shall not constitute notice) to: Pillsbury Winthrop LLP 50 Fremont Street San Francisco, CA 94105 Attention: Frederick D. Minnes Telephone No.: (415) 983-1000 Fax No.: (415) 983-1200

E-mail: fminnes@pillsburywinthrop.com

and with copies (which shall not constitute notice) to: Latham & Watkins 885 Third Avenue, Suite 1000 New York, NY 10022-4834 Attention: R. Ronald Hopkinson Telephone No.: (212) 906-1200 Fax No.: (212) 751-4864

E-mail: Ron.Hopkinson@lw.com

Any notice or other communication given hereunder shall be deemed given, on (a) if received on a Business Day on or before the normal business hours of the recipient, the date of receipt, or (b) if received on a day other than a Business Day or on a Business Day after the normal business hours of the recipient, the first Business Day following the date of receipt. Any counsel designated respectively by Sellers, Investors or Buyer above, or such counsel designated by notice to the other parties, is hereby authorized to give notices hereunder on behalf of its respective client.

- **20.5.** Governing Law. This Agreement shall be governed by and construed in accordance with the laws of the State of Florida, without regard to the conflict of laws principles thereof provided, however, that (i) for any matter or issue involving compliance with Environmental Laws, the law of the applicable state where the Property is located shall control, and (ii) the transfer of the Membership Interests and any related matters shall be governed by the laws of the State of Delaware.
- **20.6. Headings**. The captions and headings and table of contents herein are for convenience and reference only and in no way define or limit the scope or content of this Agreement or in any way affect its provisions.
- **20.7.** <u>Business Days</u>. If any date herein set forth for the performance of any obligations of any party or for the delivery of any instrument or notice as herein provided should be on a day which is not a Business Day, then the date for compliance with such obligations or delivery shall be deemed extended until the immediately following Business Day. As used herein, the term "Business Day" shall mean a day other than Saturday, Sunday or legal holiday. As used herein, the term "legal holiday" means any holiday for which financial institutions or post offices are generally closed in Sydney, New South Wales, Australia or the State of Florida.
- **20.8.** <u>Counterpart Copies</u>; <u>Facsimile</u>. This Agreement may be executed in two or more counterpart copies, all of which counterparts shall have the same force and effect as if all parties hereto had executed a single copy of this Agreement. This Agreement may be executed by facsimile signature.
- **20.9.** <u>Binding Effect</u>. Subject to the provisions of Section 18 (Assignment), this Agreement shall be binding upon, and inure to the benefit of, the parties hereto and their respective successors and permitted assigns. Except as expressly provided herein, this Agreement is solely for the benefit of the parties hereto, and no provision of this Agreement should be deemed to confer upon other parties any remedy, claim, liability, reimbursement, cause of action or other right

- **20.10.** Interpretation. This Agreement shall not be construed more strictly against one party than against the other merely by virtue of the fact that it may have been prepared by counsel for one of the parties, it being recognized that both Sellers and Buyer have contributed substantially and materially to the preparation of this Agreement. When a reference is made in this Agreement to a Section, Exhibit or Schedule such reference will be to a Section, Exhibit or Schedule of or to this Agreement unless otherwise indicated. All words or terms used in this Agreement, regardless of the number or gender in which they are used, shall be deemed to include any other number and any other gender as the context may require. The plural of any term defined in the singular, and the singular of any term defined in the plural, shall have a meaning correlative to such defined term. Whenever the words "include", "includes" or "including" are used in this Agreement, they will be deemed to be followed by the words "without limitation." All references to amounts in dollars or "\$" shall mean United States dollars.
- **20.11.** Entire Agreement; Amendments; Waiver. This Agreement and the Exhibits and Schedules attached hereto contain the final and entire agreement between the parties hereto with respect to the sale and purchase of the Membership Interests and are intended to be an integration of all prior negotiations and understandings, both written and oral, between the parties with respect to the subject matter of this Agreement. Buyer, Sellers, Investors and their agents shall not be bound by any terms, conditions, statements, warranties or representations, oral or written, not contained herein. No change or modification to this Agreement shall be valid unless the same is in writing and signed by the parties hereto. Each party reserves the right to waive any of the terms or conditions of this Agreement which are for their respective benefit and to consummate the Transactions in accordance with the terms and conditions of this Agreement which have not been so waived, provided that no waiver by any party of another party's breach of any term, covenant or condition contained in this Agreement shall be deemed to be a waiver of any subsequent breach of the same or any other term, covenant or condition of this Agreement. Any such waiver must be in writing signed by the party for whose benefit the provision is being waived. No failure of either party to exercise any power given such party hereunder or to insist upon strict compliance by either party with its obligations hereunder, and no custom or practice of the parties at variance with the terms hereof, shall constitute a waiver of either party's right to demand exact compliance with the terms hereof.
- **20.12.** <u>Severability</u>. If any one or more of the provisions hereof shall for any reason be held to be invalid, illegal or unenforceable in any respect, such invalidity, illegality or unenforceability shall not affect any other provision hereof, and this Agreement shall be construed as if such invalid, illegal or unenforceable provision had never been contained herein.
- **20.13.** Exhibits and Schedules. The Exhibits and Schedules attached hereto are incorporated herein by reference. Sellers have made available to Buyer at Sellers' office at 4350 East-West Highway, Suite 400, Bethesda, Maryland, or by delivery to Buyer's designees, true and correct copies of all information set forth on the Schedules.
- **20.14.** <u>Time</u>. The parties hereto acknowledge and agree that, except as otherwise provided in this Agreement, **TIME IS OF THE ESSENCE** for the performance of all actions (including the giving of notices, the delivery of documents and the funding of money) required or permitted to be taken under this Agreement. Whenever action must be taken (including the

giving of notice, the delivery of documents or the funding of money) under this Agreement, prior to the expiration of, by no later than or on a particular date, unless otherwise expressly provided in this Agreement, such action must be completed by 5:00 p.m. (Eastern Time) on such date, provided that such action must be completed by 3:00 p.m. (Eastern Time) with respect to the payment of the Deposit and the Purchase Price and other payments by Buyer on the Deposit Date and the Closing Date.

- **20.15. Prevailing Party.** If any action is brought by either party against the other to enforce any of the provisions hereof (whether before or after Closing, and including any claims or actions involving amounts held in escrow), or otherwise in connection with or arising out of this Agreement or any of the documents and instruments delivered in connection herewith, the non-prevailing party in any final judgment agrees to pay the other party's reasonable expenses, including reasonable attorneys' fees and expenses, in or out of Baseball Arbitration or litigation and, if in litigation, trial, appellate, bankruptcy or other proceedings, expended or incurred in connection therewith, as determined by a court of competent jurisdiction.
- **20.16. No Recording.** Neither this Agreement nor any memorandum or short form hereof shall be recorded or filed in any public land or other public records of any jurisdiction, by either party and any attempt to do so may be treated by the other party as a breach of this Agreement.
- 20.17. WAIVER OF TRIAL BY JURY. THE RESPECTIVE PARTIES HERETO SHALL AND HEREBY DO WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM BROUGHT BY EITHER OF THE PARTIES HERETO AGAINST THE OTHER ON ANY MATTERS WHATSOEVER ARISING OUT OF OR IN ANY WAY CONNECTED WITH THIS AGREEMENT, OR FOR THE ENFORCEMENT OF ANY REMEDY UNDER ANY STATUTE, EMERGENCY OR OTHERWISE.
- **20.18.** Further Assurances. If at any time prior to or after the Closing any further action is necessary to carry out the purposes of this Agreement or any other documents or agreements reasonably requested by either Sellers, Investors or Buyer in order to consummate the Transactions, the parties to this Agreement shall take all such necessary action; provided, however, that such further assurances do not increase such party's obligations or liabilities hereunder or decrease such party's rights hereunder.
- **20.19.** No Offer. This Agreement shall not be deemed an offer or binding upon Sellers, Investors or Buyer until this Agreement is fully executed and delivered by Sellers, Investors and Buyer.
- **20.20. OFAC Policy.** Neither Buyer nor any of its Affiliates have engaged in any dealings or transactions, directly or indirectly, (a) in contravention of any U.S., international or other money laundering regulations or conventions, including the United States Bank Secrecy Act, the United States Money Laundering Control Act of 1986, the United States International Money Laundering Abatement and Anti-Terrorist Financing Act of 2001, Trading with the Enemy Act (50 U.S.C. §1 et seq., as amended), or any foreign asset control regulations of the United States Treasury Department (31 CFR, Subtitle B, Chapter V, as amended) or any

enabling legislation or executive order relating thereto, or (b) in contravention of Executive Order No. 13224 dated September 24, 2001 issued by the President of the United States (Executive Order Blocking Property and Prohibiting Transactions with Persons Who Commit, Threaten to Commit, or Support Terrorism), as may be amended or supplemented from time to time ("Anti-Terrorism Order") or on behalf of terrorists or terrorist organizations.

**20.21.** <u>Limitation of Liability.</u> Notwithstanding anything to the contrary contained in this Agreement, it is understood and agreed that none of the employees, directors, officers, members, partners, managers, principals, consultants, shareholders, advisors, attorneys, or agents of Seller, shall have any personal liability or obligation whatsoever for obligations under this Agreement or under any documents delivered at Closing, and the individual assets of such parties shall not be subject to any claims of any Person relating to such obligations. Notwithstanding anything to the contrary contained in this Agreement, it is understood and agreed that none of the employees, directors, officers, members, partners, managers, principals, consultants, shareholders, advisors, attorneys or agents of Buyer or Investors shall have any personal liability or obligation whatsoever for any obligations under this Agreement or under any documents delivered at Closing, and the individual assets of such parties shall not be subject to any claims of any Person relating to such obligations. However, the foregoing shall not in any way limit the parties' obligations and liabilities under this Agreement.

### 20.22. Macquarie CountryWide Management Limited.

- (a) Macquarie CountryWide Management Limited ("MCML") enters into this Agreement only in its capacity as responsible entity of MCW and in no other capacity.
- (b) A liability arising under or in connection with this Agreement can be enforced against MCML, only to the extent to which it can be satisfied out of the assets of MCW out of which MCML is actually indemnified for the liability.
- (c) This limitation of MCML's liability applies notwithstanding any other provision of this Agreement and extends to all liabilities and obligations of MCML in any way connected with any representation, warranty, conduct, omission, agreement or transaction related to this Agreement.
- (d) Any party to this Agreement may not sue MCML in any capacity other than as responsible entity of MCW, including seeking the appointment to MCML of a receiver (except in relation to MCW's assets), liquidator, administrator or any similar Person.
- (e) The provisions of this Section 20.22 shall not apply to any obligation or liability of MCML to the extent that it is not satisfied because there is a reduction in the extent of MCML's indemnification out of the assets of MCW, as a result of MCML's fraud, negligence or breach of trust.
- (f) MCML is not obliged to do or refrain from doing anything under this Agreement (including incurring any liability) unless MCML's liability as responsible entity is limited in the same manner as set out in this Section 20.22.

20.23. Venue. Any legal action or proceeding with respect to this Agreement shall be solely and exclusively brought in a Federal or state court of competent jurisdiction sitting in Jacksonville, Duval County, Florida (including the appellate courts thereof) (each, a "Court") and by execution and delivery of this Agreement, each party to this Agreement hereby accepts, generally and unconditionally, the jurisdiction of the Courts. Each party to this Agreement hereby expressly and irrevocably submits the person of such party to this Agreement to the personal jurisdiction of the Courts in any suit, action or proceeding arising, directly or indirectly, out of or relating to this Agreement. To the extent permitted under applicable Law, this consent to personal jurisdiction shall be self-operative and no further instrument or action, other than service of process in one of the manners specified in this Agreement or as otherwise permitted by Law, shall be necessary in order to confer jurisdiction upon the person of such party to this Agreement in any such Court. To the fullest extent permitted under applicable Law, each party to this Agreement irrevocably waives and agrees not to assert, by way of motion, as a defense or otherwise, any objection which it may now or hereafter have to the laying of the venue of any such suit, action or proceeding brought in a Court and any claim that any such suit, action or proceeding has been brought in an inconvenient forum, any claim that it is not personally subject to the jurisdiction of any such Court or that this Agreement or the subject matter hereof may not be enforced in or by such Court.

**20.24.** Control of Company. Notwithstanding anything to the contrary contained herein, (a) any reference to a Company or other entity in the Company Group taking or not taking an action at or prior to the Closing shall mean that Sellers shall cause such Company or other entity in the Company Group to take or not take such action, at Sellers' sole cost and expense, and (b) after the Closing Sellers shall have no obligations to cause or prevent the Company Group from taking or failing to any action.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement under seal on the date or dates set forth below.

Date: February 14, 2005

MACQUARIE COUNTRYWIDE-REGENCY II, LLC, a Delaware limited liability company

By: Regency Centers, L.P., a Member

By: Regency Centers Corporation, its General Partner

By: /s/ Martin E. Stein, Jr.

Its: Chairman and Chief Executive Officer

MACQUARIE COUNTRYWIDE TRUST

By: Macquarie CountryWide Management Limited, as responsible entity

By: /s/ Mark Baillie

Its: Director

REGENCY CENTERS CORPORATION, a Florida corporation

By: /s/ Martin E. Stein, Jr.

Its: Chairman and Chief Executive Officer

USRP TEXAS GP, LLC, a Delaware limited liability company

By First Washington Realty, Inc.,

a Maryland corporation, its manager

By: /s/ Stuart D. Halpert

Its: Chairman

U.S. RETAIL PARTNERS, LLC, a Delaware limited liability company

By First Washington Realty, Inc., a Maryland corporation, its manager

By: /s/ Stuart D. Halpert

Its: Chairman

EASTERN SHOPPING CENTER HOLDINGS, LLC, a Delaware limited liability company

By First Washington Realty, Inc., a Maryland corporation, its manager

By: /s/ Stuart D. Halpert

Its: Chairman

FIRST WASHINGTON INVESTMENT I, LLC, a Delaware limited liability company

By: /s/ William J. Wolfe

Its: Member

CALIFORNIA PUBLIC EMPLOYEES' RETIREMENT SYSTEM, a unit of the State and Consumer Services Agency of the State of California

By: /s/ Michael B. McCook

Its: Senior Investment Officer – Real Estate

# The following EXHIBITS and SCHEDULES are intentionally omitted. Copies will be provided to the Commission upon request.

## List of Exhibits

Exhibit A Definitions and Meanings (attached)

Exhibit B Properties

Exhibit C Form of Tenant Estoppel Exhibit D Form of Seller Estoppel

Exhibit E Form of Sellers' Ground Lease Estoppel Certificate

Exhibit F Terms of Management Agreement and Accounting Services Agreement

## List of Schedules

Schedule 2.2(a)(v) Construction Contracts Schedule 2.3(a) Wire Transfer Instructions

Schedule 2.10 Excluded Assets
Schedule 2.14 Option Parcels
Schedule 2.15 MPB&T Parcel

Schedule 3.1(m) Rent Guaranty and Other Escrows
Schedule 5.1(a)-1 Existing Policies and Existing Surveys
Schedule 5.1(a)-2 Exceptions Not Provided to Buyer

Schedule 5.1(a)-3 Extended Coverage

Schedule 5.1(b)(ii) List of Existing Rights of First Refusal or Options to Purchase Under Leases and Approved New Leases

Schedule 5.1(b)(v) Other Exceptions

Schedule 5.6(a)(i) Ongoing Remediation Projects

Schedule 5.6(a)(ii) Ongoing Remediation Escrow Allocated Shares Schedule 5.6(f) Form of Buyer's Environmental Insurance Policy

Schedule 7.3(a)(iii) Permitted Sales

Schedule 8.1(e) Jurisdictions Where Company Group is Licensed or Qualified to Do Business

Schedule 8.1(f) Capitalization of the Companies

Schedule 8.1(g) Subsidiaries

Schedule 8.1(h) Company Organizational Documents Schedule 8.3 Properties and Title to Properties

Schedule 8.5 No Violation
Schedule 8.6 Financial Statements
Schedule 8.8(a) Taxes; Tax Returns
Schedule 8.8(d) Partnership Status

Schedule 8.9 Litigation

Schedule 8.11 Ongoing Condemnation Proceedings

Schedule 8.14 Rent Rolls; Leases

Schedule 8.15 Accounts Receivable Aging

Schedule 8.16 Service Contracts
Schedule 8.17 Material Contracts
Schedule 8.18 Existing Loans
Schedule 8.19 Ground Leases
Schedule 8.20 No Other Contracts

Schedule 8.22 Insurance

Schedule 8.24 Taxpayer Identification Numbers

Schedule 8.27 Shadow Anchors
Schedule 11.2(c) Underwriting Agreement
Schedule 13.2(a) Extended Lockout Properties

# EXHIBIT A DEFINITIONS AND MEANINGS

In addition to any other terms whose definitions are fixed and defined by this Agreement, each of the following defined terms, when used in this Agreement with an initial capital letter, shall have the meaning ascribed thereto by this **Exhibit A**:

- "Accounts Receivable Aging" is defined in Section 11.2(d).
- "ADA Litigation Matters" is defined in Section 15.3(i).
- "Additional Rent" is defined in Section 3.1(c)(iii) hereof.
- "Adjustment Time" is defined in Section 3.1 hereof.
- "Affiliate" of a person means a person that directly or indirectly, through one of its intermediaries, controls or is under common control with, the other person.
  - "Aggregate Loss Amount" is defined in Section 15.2(d)(ii) hereof.
- "Agreement" means this Agreement of Purchase and Sale, together with any and all amendments, modifications, or supplements hereof and any and all exhibits, schedules, and addenda attached hereto.
  - "Agreed Losses" is defined in Section 15.2(c)(ii) hereof.
  - "Agreed Percentage" is defined in Section 20.2 hereof.
- "Agreed Value" as to each Property shall mean the amount set forth on Exhibit B, and as to each Option Parcel and the MPB&T Parcel, the amount set forth on Schedule 2.14.
  - "Alleged Diminution of Value" is defined in Section 5.2 hereof.
  - "Alleged Losses" is defined in Section 15.2(c)(i) hereof.
- "Ancillary Instruments" means all other documents, agreements and instruments to be executed and/or delivered by the Selling Parties or Buyer at or prior to the Closing as specifically required pursuant to this Agreement.
  - "Apex Contracts" is defined in Section 5.6(a).
  - "Applicable Rent Period" shall have the meaning set forth in Section 3.1(c)(iii) hereof.
  - "Approved New Lease" is defined in Section 7.3(d) hereof.
  - "Authorized Representatives" is defined in Section 17(a) hereof.
  - "Baseball Arbitration" is defined in Section 19 hereof.

- "Blocked Account Agreement" is defined in Section 2.3(b) hereof.
- "Breach Event" is defined in Section 15.2(a)(ii) hereof.
- "Broker" is defined in Section 9.3 hereof.
- "Business Day" is defined in Section 20.7 hereof.
- "Buyer" is defined in the preamble to this Agreement.
- "Buyer's Affiliates" is defined in Section 15.3 hereof.
- "Buyer's Breach Notice" is defined in Section 15.2(c)(i) hereof.
- "Buyer's Broker Rep" is defined in Section 9.3 hereof.
- "Buyer's Consent" is defined in Section 7.3(g) hereof.
- "Buyer's Environmental Insurance Policy" is defined in Section 5.6(f).
- "CAM" is defined in Section 3.1(c)(iii) hereof.
- "Claim" is defined in Section 15.3 hereof.
- "Closing" is defined in Section 2.4 hereof.
- "Closing Date" means the time and date, established under Section 2.4 hereof, when the purchase and sale contemplated by this Agreement is to be consummated, as such date may be extended by mutual agreement of the parties or pursuant to the express provisions hereof.
  - "Closing Statement" is defined in Section 3.1(n)(ii) hereof.
  - "Code" is defined in Section 2.8 hereof.
  - "Company" is defined in the preamble to this Agreement.
  - "Company Group" is defined in Recital A hereof.
  - "Competing Transaction" is defined in Section 7.2(h) hereof.
- "Confidential Information" shall mean Company Group's confidential information about the Properties, business operations, financial results and similar confidential information, heretofore or hereafter furnished or made available by or through any Sellers Party or Company Group to Buyer or any Affiliate of Buyer, or any summary of the foregoing. Confidential Information shall not include information that is otherwise known by or readily available to the public.
  - "Conforming Ground Lease Estoppel" is defined in Section 11.2(e).

- "Consent Fees" is defined in Section 2.9 hereof.
- "Court" is defined in Section 20.23 hereof.
- "Deeds" means all instruments pursuant to which title to the Properties or any other portion thereof has been conveyed to or any of the Company Group.
- "Deferred Maintenance Losses" is defined in Section 15.2(a)(iii).
- "Deferred Maintenance Conditions" shall mean those facts, circumstances and conditions which, consistent with prudent industry management practices and standards for similar properties, require this to complete repairs needed to be made within the ensuing six (6) months subsequent to the date of this Agreement to the roofs, parking lots and structural elements of the Improvements which (i) if not made within such period of time will directly result in an unsafe condition at the applicable Property or will cause damage to other portions of the Improvements, and (ii) should be capitalized under generally accepted accounting principles.
  - "Delinquent Rent" is defined in Section 3.1(d) hereof.
  - "Deposit" is defined in Section 2.3(a) hereof.
  - "Deposit Date" shall mean March 25, 2005.
  - "Disputed Amount" is defined in Section 19 hereof.
  - "Disputed Amount Escrow" is defined in Section 2.5 hereof.
  - "Disputed Losses" is defined in Section 15.2(c)(ii) hereof.
  - "Dropped Asset" is defined in Section 2.13(a) hereof.
  - "Dropped Property Notice" is defined in Section 2.13(a) hereof.
  - "Dropping Party" is defined in Section 2.13(a) hereof.
  - "Eastern" is defined in the preamble of this Agreement.
  - "Edgewater" is defined in Section 2.11(a).
  - "Edgewater Buy-Sell Amount" is defined in Section 2.11(b).
- "Encumbrances" means any charge, claim, equitable interest, lien, option, pledge, security interest, right of first refusal, or restriction of any kind, including any restriction on use, voting, transfer, receipt of income, or exercise of any other attribute of ownership.
  - "Entity Breach" is defined in Section 15.2(a)(iv).
  - "Entity Claim" is defined in Section 15.3(d) hereof.

"Environmental Laws" means the following, as the same may be amended from time to time: the Comprehensive Environmental Response, Compensation and Liability Act of 1980 (commonly known as "CERCLA"), as amended, the Superfund Amendment and Reauthorization Act (commonly known as "SARA"), the Resource Conservation and Recovery Act (commonly known as "RCRA"), and any other federal, state, city or county Laws applicable to the Properties, and identified by its terms as relating to or regulating the protection of the environment or pertaining to Hazardous Substances or Waste, together with their implementing regulations.

- "ERISA" means the Employee Retirement Income Security Act of 1974, as amended, and the rules and regulations promulgated thereunder.
- "Escrow Agent" is defined in Section 2.3(c) hereof.
- "Escrow Agreement" is defined in Section 2.3(c) hereof.
- "Estoppel Threshold" is defined in Section 11.2(d) hereof.
- "Excess Losses" is defined in Section 15.2(a)(v) hereof.
- "Existing Environmental Reports" is defined in Section 8.12 hereof.
- "Existing Loans" is defined in Section 8.18 hereof.
- "Existing Policy" is defined in Section 5.1(a) hereof.
- "Existing Survey" is defined in Section 5.1(a) hereof.
- "Extended Coverage" is defined in Section 5.1(a) hereof.
- "**Final Closing Adjustment**" is defined in Section 3.1(n)(iii) hereof.
- "Final Offers" is defined in Section 19(b) hereof.
- "First Washington" is defined in the preamble of this Agreement.
- "Final Statement" is defined in the preamble of this Agreement.
- "Fixed Rent" is defined in Section 3.1(c)(ii) hereof.
- "FNF Entities" is defined in Section 5.1 hereof.
- "Fixed Financial Obligations" is defined in Section 15.2(a)(i).

"Future Remedial Action" means any investigative, monitoring or remedial action (including any consulting fees, design fees, attorneys fees and engineering fees related thereto) required under any Environmental Law relating to the presence of Hazardous Substances or Waste at, on, about, under or within or emanating from any of the Properties (or any portion thereof) and any reasonably related investigative or remedial action. This includes but is not

limited to any investigation, monitoring or remedial action (including any consulting fees, design fees, attorneys fees and engineering fees related thereto) undertaken pursuant to any state voluntary cleanup program or brownfield program.

- "Future Remediation Escrow" is defined in Section 2.5(c) hereof.
- "Future Remediation Escrow Agreement" is defined in Section 2.5(c) hereof.
- "Future Remediation Estimate" is defined in Section 5.6(b)(iii).
- "Future Remediation Projects" is defined in Section 5.6(b)(i) hereof.
- "Future Remediation Sites" is defined in Section 5.6(b)(i) hereof.
- "GAAP" means the generally accepted United States accounting principles, applied on a basis consistent with the basis on which the 2003 financial statements referred to in Section 8.6(a).
- "Governmental Authority" means, with respect to any Property, any federal, state, county, municipal, or other governmental authority or quasi-governmental authority, agency, board or office having jurisdiction of such Property.
  - "Governmental Restrictions" is defined in Section 5.1(b)(vi) hereof.
  - "Ground Lease Estoppels" is defined in Section 11.2(e) hereof.
- "Ground Leases" means all ground leases or other agreements pursuant to which any entity in the Company Group or any of the Selling Parties lease any real property.
  - "Guaranties" means all guaranty agreements related to any of the Leases which guarantee the obligations of the Tenant under the applicable Lease.
- "Hazardous Substances or Waste" means petroleum (including gasoline, crude oil or any crude oil fraction), and chemicals or hazardous substances of any nature, including radioactive materials, PCBs, asbestos, mold, perchloroethylene and its degradation products, pesticides, herbicides, pesticide or herbicide containers, untreated sewerage, industrial process sludge, and any item which is identified as a hazardous substance, hazardous material, or hazardous waste in any Environmental Law.
- "Improvements" shall mean all buildings, components of buildings, roads, streets, curbs, gutters, storm drainage, water systems, sewer systems, electrical systems, gas systems and other utilities, equipment and landscaping, whether or not located on any of the Properties, which are required in order to serve any of the Properties.
  - "Indemnified Party" is defined in Section 15.5(a) hereof.
  - "Indemnifying Party" is defined in Section 15.5(a) hereof.
  - "Initial Remediation Estimate" is defined in Section 2.5(c) hereof.

- "Initiating Party" is defined in Section 19(d) hereof.
- "Inspection Period" is defined in Section 4.1 hereof.
- "Interest Rate" is defined in Section 2.3(a) hereof.
- "Investor" is defined in the preamble to this Agreement.
- "Joint Instructions" is defined in Section 14.1(o) hereof.
- "Knowledge of the Selling Parties," "to the Knowledge of Sellers," "to Sellers' Knowledge," and other phrases of similar import mean the current, actual, conscious (and not constructive, imputed or implied) knowledge, without any duty to investigate, review files or inquire, of the following designees of the Sellers: Stuart D. Halpert, William J. Wolfe, James G. Blumenthal, Jeffrey D. Distenfeld, James G. Pounds and Henry M. Renaud Anything herein to the contrary notwithstanding, no such designee shall have any personal liability or obligation whatsoever with respect to any of the matters set forth in this Agreement and any Ancillary Documents or any of the representations made by the Sellers being or becoming untrue, inaccurate or incomplete in any respect.
- "Law" means any law, statute, rule, regulation, ordinance and other pronouncements having the effect of law of the United States of America, any foreign country or any domestic or foreign state, county, city or any other political subdivision or any governmental or regulatory body.
  - "Leases" means all leases, tenancy and occupancy agreements with respect to any of the Properties or any portion thereof.
  - "Leasing Parameters" is defined in Section 7.3 hereof.
- "Lender Retention Documents" means all documents necessary for the Retained Loans to remain in place after Closing, including an estoppel letter executed by each Lender's party to the Retained Loans.
  - "Lender Consents" is defined in Section 2.9 hereof.
  - "Lenders" means the holders of the Existing Loans.
- "Lien" means any adverse claim, mortgage, pledge, statutory lien or other lien arising by operation of law, security interest, trust arrangement, security deed, financing lease, collateral assignment or other encumbrance, conditional sale or title retention agreement, or any other interest in property designed to secure the repayment of any indebtedness or the performance of any obligation, or any option or other right to purchase or otherwise acquire, whether arising by agreement or under any Law or otherwise.
  - "Loan Fees" is defined in Section 2.9 hereof.
  - "Losses" is defined in Section 15.2(a)(i) hereof.

- "Major Tenant" is defined in Section 11.2(d) hereof.
- "Major Tenants" is defined in Section 11.2(d) hereof.
- "Mandatory Removal Exception" is defined in Section 5.3.
- "Material Contracts" is defined in Section 8.17 hereof.
- "MCW" is defined in the preamble to this Agreement.
- "Membership Interests" are defined in Recital A to this Agreement.
- "MPB&T Parcel" is defined in Section 2.15 hereof.
- "NOI Losses" is defined in Section 15.2(a)(i) hereof.
- "Non-Conforming Ground Lease Estoppel" is defined in Section 11.2(e) hereof.
- "Non-Conforming Property" is defined in Section 11.2(d) hereof.
- "Non-Conforming Tenant Estoppel" is defined in Section 11.2(d) hereof.
- "Non-Retained Loans" means the Existing Loans which will be paid by the Company Group at Closing.
- "Notice of Intent to Pursue Arbitration" is defined in Section 19(b) hereof.
- "Ongoing Remediation Escrow" is defined in Section 2.5(b) hereof.
- "Ongoing Remediation Escrow Agreement" is defined in Section 2.5(b) hereof.
- "Ongoing Remediation Amount" is defined in Section 2.2(a)(viii) hereof.
- "Ongoing Remediation Projects" is defined in Section 5.6(a)(i) hereof.
- "Ongoing Remediation Sites" is defined in Section 5.6(a)(i) hereof.
- "Option Holders" is defined in Section 2.14 hereof.
- "Option Parcels" is defined in Section 2.14 hereof.
- "Order" means any writ, judgment, decree, injunction or similar order of any court or governmental or regulatory body, in each case whether preliminary or final.
- "Organizational Documents" shall mean as to any Person, its (i) certificate or articles of incorporation or similar corporate charter or other instruments of organization; (ii) articles of association, by-laws or other similar instruments; and (iii) shareholder agreements, limited liability company agreements or operating agreements and other similar governing corporate documents.

- "Other Exceptions" are defined in Section 5.1(c) hereof.
- "Outside Partners" is defined in Section 8.1(h) hereof.
- "Owner's Title Policy" is defined in Section 5.1(a) hereof.
- "Percentage Rent" is defined in Section 3.1(c)(iv) hereof.
- "Permitted Exceptions" is defined in Section 5.1(b) hereof.
- "Person" shall mean an individual, corporation, partnership, limited liability company, joint venture, association, trust, unincorporated organization or other entity.
- "Phase I" means a Phase I environmental site assessment prepared in general accordance with ASTM 1527-E-00, modified as necessary due to time constraints.
  - "Phase II Investigation" is defined in Section 5.6(b) hereof.
  - "Phase II Reports" is defined in Section 5.6(b)(i) hereof.
  - "Physical Testing" is defined in Section 4.1(a) hereof.
  - "Position Items" is defined in Section 3.1 hereof.
  - "Post Escrow Future Remediation Costs" is defined in Section 5.6(b)(iv) hereof.
  - "Prepayment Fees" is defined in Section 2.9 hereof.
  - "Price Adjustment" is defined in Section 2.2(a) hereof.
- "**Property**" and "**Properties**" are defined in Recital A hereof and described on Exhibit B; however, the definition of Properties shall be modified to the extent a Property is removed from.
  - "Property Breach" is defined in Section 15.2(a) hereof.
  - "Property-Related Adjustment Losses" is defined in Section 15.2(a)(vii).
  - "Proration Items" is defined in Section 3.1 hereof.
  - "PSE&G" is defined in Section 2.11(c) hereof.
  - "PSE&G Facilities" is defined in Section 2.11(c) hereof.
- "Purchase Price" means the amount which Buyer shall pay to Sellers to consummate the purchase and sale of the Membership Interests as provided in Section 2.1 hereof.

- "Qualified Arbitrator" means a real estate professional who is knowledgeable about the subject matter giving rise to a particular dispute, controversy or claim and has no past or present personal or professional relationship with Sellers or Buyer or their respective affiliates.
  - "Recent Balance Sheet" is defined in Section 8.6 hereof.
  - "Regency" is defined in the preamble to this Agreement.
- "Remediation" means any investigative, monitoring or remedial action required under any Environmental Law relating to the presence of Hazardous Substances or Waste at, on, about, under or within or emanating from any of the Properties (or any portion thereof). This includes but is not limited to any investigation, monitoring or remedial action in furtherance of obtaining a "no further action" or equivalent determination pursuant to any state voluntary cleanup program or brownfield program. "Remediation" does not include work to abate asbestos or any work that is not necessary to obtain a "no further action" letter or equivalent determination from the relevant Governmental Authority.
- "Remediation Costs" means all reasonable costs and expenses (including reasonable costs of investigation, monitoring, cleanup consulting fees, design fees, attorney fees and engineering fees) paid to third parties which are incurred with respect to Remediation and which are documented by third party invoices.
  - "Rent" is defined in Section 3.1(d) hereof.
  - "Rent Rolls" is defined in Section 8.14 hereof.
  - "Responding Party" is defined in Section 19(d) hereof.
  - "Retained Loans" means the Existing Loans which may be retained by an entity in the Company Group after Closing.
  - "Section 2.13 Rights" is defined in Section 15.2(d)(ii) hereof.
  - "Seller" or "Sellers" means the Sellers as described in the preamble to this Agreement.
  - "Seller Estoppel" is defined in Section 11.2(d) hereof.
  - "Seller Indemnified Party" is defined in Section 15.4.
  - "Sellers' Broker Rep" is defined in Section 8.21 hereof.
  - "Sellers' Environmental Insurance" is defined in Section 5.6(b)(ii) hereof.
  - "Sellers' GL Estoppel" is defined in Section 11.2(e) hereof.
  - "Sellers' Post-Closing Occurrence Claims" is defined in Section 15.4(c) hereof.
  - "Sellers' Response Notice" is defined in Section 15.2(c)(ii) hereof.

- "Selling Parties" is defined in the preamble to this Agreement.
- "Service Contracts" are defined in Section 8.16 hereof.
- "Shadow Anchor Agreements" are defined in Section 8.16.
- "Special Covenants" are defined in Section 7.3 hereof.
- "Special Entity Claims" are defined in Section 15.3(g) hereof.
- "Subsidiary" is defined in Section 8.1(g) hereof.
- "Taxes" is defined in Section 8.8(a) hereof.
- "Tax Return" is defined in Section 8.8(a) hereof.
- "**Tenant Estoppels**" is defined in Section 11.2(d) hereof.
- "Tenants" means the tenants under the Leases.
- "Third Party Environmental Claims" is defined in Section 15.3(h) hereof.
- "Title Affidavits" is defined in Section 14.1(e) hereof.
- "Title Commitments" are defined in Section 5.1(a) hereof.
- "Title Company" means Land America Title Insurance Company.
- "Transaction Documents" means (i) this Agreement, including the Schedules and Exhibits; and (ii) the Ancillary Instruments.
- "Transactions" means the transactions contemplated by the Transaction Documents.
- "Unfunded Leasing Expenses" is defined in Section 3.1(j) hereof.
- "USRP" is defined in the preamble of this Agreement.
- "U.S. Retail" is defined in the preamble of this Agreement.

EXHIBIT B PROPERTIES

B-1

EXHIBIT C FORM OF TENANT ESTOPPEL

# EXHIBIT D FORM OF SELLER ESTOPPEL

# EXHIBIT E FORM OF SELLERS' GROUND LEASE ESTOPPEL CERTIFICATE

# EXHIBIT F TERMS OF MANAGEMENT AGREEMENT AND ACCOUNTING SERVICES AGREEMENT

# Certification of Chief Executive Officer Pursuant to Section 302 of the Sarbanes-Oxley Act and Rule 13a-14(a) or 15d-14(a) under the Securities Exchange Act of 1934

#### I, Martin E. Stein, Jr., certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Regency Centers Corporation ("registrant");
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2005
/s/ Martin E. Stein, Jr.

Martin E. Stein, Jr.
Chief Executive Officer

# Certification of Chief Financial Officer Pursuant to Section 302 of the Sarbanes-Oxley Act and Rule 13a-14(a) or 15d-14(a) under the Securities Exchange Act of 1934

#### I, **Bruce M. Johnson**, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Regency Centers Corporation ("registrant");
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e)) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2005
/s/ Bruce M. Johnson

Bruce M. Johnson
Chief Financial Officer

# Certification of Chief Operating Officer Pursuant to Section 302 of the Sarbanes-Oxley Act and Rule 13a-14(a) or 15d-14(a) under the Securities Exchange Act of 1934

#### I, Mary Lou Fiala, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Regency Centers Corporation ("registrant");
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officers and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: May 9, 2005
/s/ Mary Lou Fiala
Mary Lou Fiala
Chief Operating Officer

# Written Statement of the Chief Executive Officer Pursuant to 18 U.S.C. §1350

Solely for the purposes of complying with 18 U.S.C. §1350, I, the undersigned Chairman and Chief Executive Officer of **Regency Centers Corporation** (the "Company"), hereby certify, based on my knowledge, that the Quarterly Report on Form 10-Q of the Company for the quarter ended **March 31, 2005** (the "Report") fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934 and that information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 9, 2005		
/s/ Martin E. Stein, Jr.		
Martin E. Stein, Jr.		
Chief Executive Officer		

# Written Statement of the Chief Financial Officer Pursuant to 18 U.S.C. §1350

Solely for the purposes of complying with 18 U.S.C. §1350, I, the undersigned Managing Director and Chief Financial Officer of **Regency Centers Corporation** (the "Company"), hereby certify, based on my knowledge, that the Quarterly Report on Form 10-Q of the Company for the quarter ended **March 31, 2005** (the "Report") fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934 and that information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 9, 2005	
/s/ Bruce M. Johnson	
Bruce M. Johnson Chief Financial Officer	

# Written Statement of the Chief Operating Officer Pursuant to 18 U.S.C. §1350

Solely for the purposes of complying with 18 U.S.C. §1350, I, the undersigned President and Chief Operating Officer of **Regency Centers Corporation** (the "Company"), hereby certify, based on my knowledge, that the Quarterly Report on Form 10-Q of the Company for the quarter ended **March 31, 2005** (the "Report") fully complies with the requirements of Section 13(a) of the Securities Exchange Act of 1934 and that information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: May 9, 2005	
/s/ Mary Lou Fiala	
Mary Lou Fiala Chief Operating Officer	